UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (Date of earliest event reported): November 1, 2024

NEKTAR THERAPEUTICS (Exact Name of Registrant as Specified in Charter)

Delaware	0-24006	94-3134940		
(State or Other Jurisdiction of Incorporation)	(Commission File Number)	(IRS Employer Identification No.)		
(A	455 Mission Bay Boulevard South San Francisco, California 94158 Address of Principal Executive Offices and Zip Co	ode)		
Registra	ant's telephone number, including area code: (415)	482-5300		
Check the appropriate box below if the Form 8-I following provisions:	K filing is intended to simultaneously satisfy the	filing obligation of the registrant under any of the		
☐ Written communications pursuant to Rule 425	under the Securities Act (17 CFR 230.425)			
☐ Soliciting material pursuant to Rule 14a-12 und	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
☐ Pre-commencement communications pursuant	to Rule 14d-2(b) under the Exchange Act (17 CFR	240.14d-2(b))		
☐ Pre-commencement communications pursuant	to Rule 13e-4(c) under the Exchange Act (17 CFR 2	240.13e-4(c))		
Securities registered pursuant to Section 12(b) of	f the Act:			
Title of each class	Trading symbol(s)	Name of each exchange on which registered		
Common Stock, \$0.0001 par value	NKTR	Nasdaq Capital Market		
Indicate by check mark whether the registrant is a chapter) or Rule 12b-2 of the Securities Exchange A		405 of the Securities Act of 1933 (§230.405 of this		
Emerging growth company \square				
If an emerging growth company, indicate by check or revised financial accounting standards provided pr		tended transition period for complying with any new		

Item 1.01. Entry into a Material Definitive Agreement.

On November 1, 2024, Nektar Therapeutics, a Delaware corporation ("Nektar"), entered into an Asset Purchase Agreement (the "APA") with an affiliate of Ampersand Management LLC d/b/a Ampersand Capital Partners, a Delaware limited liability company (said affiliate, "Purchaser"). Pursuant to the APA, on the terms and subject to the conditions set forth therein, Nektar has agreed to sell to Purchaser, and Purchaser has agreed to purchase from Nektar, Nektar's manufacturing facility located in Huntsville, Alabama (the "Facility") and certain other manufacturing assets related thereto, as well as assume certain related liabilities (collectively, the "Transactions").

The total base purchase price payable by Purchaser to Nektar at the closing of the Transactions is \$90 million (subject to customary adjustments based on closing net working capital, closing inventory and the amount of any indebtedness assumed by Purchaser), consisting of \$70 million in cash and \$20 million in an equity position in a parent entity of Purchaser (which equity position will be accounting for at fair value under U.S. GAAP). Ampersand has also committed to invest additional growth equity capital into Purchaser.

The closing of the Transactions is expected to occur in the fourth quarter of 2024, subject to the satisfaction or waiver of certain customary closing conditions set forth in the APA.

The APA contains customary representations and warranties of Nektar and Purchaser. Nektar has agreed to provide certain specific and customary indemnities to Purchaser for certain excluded liabilities not covered by a representation and warranties policy to be obtained by Purchaser in connection with the Transactions. In addition, the APA contains certain restrictive covenants, which will restrict, as of the closing of the Transactions, Nektar's ability to manufacture and sell PEG reagents as presently conducted by Nektar at the Facility.

In addition, the APA provides that, at the closing of the Transactions, each of Nektar and Purchaser will enter into certain ancillary and customary agreements to effect the Transactions, including a mutual transition services agreement, and one or more manufacturing and supply agreements.

The foregoing description of the APA and the Transactions does not purport to be complete and is qualified in its entirety by reference to the full text of the APA. The APA (with certain confidential terms redacted, as indicated thereon) is filed as Exhibit 1.1 to this Current Report on Form 8-K and is incorporated herein by reference. The APA has been filed to provide information regarding its terms. The APA is not intended to modify or supplement any factual disclosures about Nektar or the Facility in any public reports filed or to be filed with the U.S. Securities and Exchange Commission (the "SEC") by Nektar. In particular, the assertions embodied in the representations, warranties, and covenants contained in the APA will be made only for purposes of the APA and as of specified dates, will be solely for the benefit of the parties to the APA, and will be subject to limitations agreed upon by the parties to the APA. Moreover, certain representations and warranties in the APA are intended for the purposes of allocating risk between the parties to the APA instead of establishing matters of fact. Accordingly, the representations and warranties in the APA may not constitute the actual state of facts about Nektar, Nektar's business, the Facility or Purchaser. The representations and warranties set forth in the APA may also be subject to a contractual standard of materiality different from that generally applicable under federal securities laws. Investors should not rely on the representations, warranties, or covenants or any descriptions thereof as characterizations of the actual state of facts or the actual condition of Nektar, Nektar's business, the Facility or any of Nektar's subsidiaries or affiliates. Moreover, information concerning the subject matter of the representations and warranties may change after the date of this report, which subsequent information may or may not be fully reflected in Nektar's public disclosures.

Item 7.01 Regulation FD Disclosure.

A copy of Nektar's press release announcing the execution of the APA and the Transactions is attached to this report as Exhibit 99.1. The information contained in Exhibit 99.1 is being furnished pursuant to Item 7.01 of this Form 8-K (including the exhibits hereto) and shall not be considered "filed" under the Securities Exchange Act of 1934, as amended, nor shall it be incorporated by reference into any of the registrant's filings under the Securities Act of 1933, as amended, or under the Securities Exchange Act of 1934, as amended, unless the registrant expressly states in such filing that such information is to be considered "filed" or incorporated by reference therein.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits.

Exhibit No.	Description
1.1	Asset Purchase Agreement, dated as of November 1, 2024, between Nektar and Purchaser.
99.1	Press Release.
104	Cover Page Interactive Data File (embedded within the Inline XBRL document).

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: November 4, 2024

NEKTAR THERAPEUTICS

By: /s/ Mark A. Wilson

Mark A. Wilson

Chief Legal Officer and Secretary

Execution Version

CONFIDENTIAL

ASSET PURCHASE AGREEMENT

by and between

NEKTAR THERAPEUTICS

and

[* * *]

Dated as of November 1, 2024

TABLE OF CONTENTS

Clause	_		Page
1.	Defini	TIONS AND TERMS	1
	1.1	Definitions	1
	1.2	Other Definitional Provisions	16
2.	Purchase and Sale		17
	2.1		1.7
	2.1	Purchase and Sale of Assets	17 18
	2.2	Consents Excluded Assets	18
	2.4	Assumption of Certain Obligations	21
	2.5	Excluded Liabilities	21
	2.6	Purchase Price	22
	2.7	Purchase Price Adjustment	23
	2.8	Allocation of the Purchase Price	26
	2.9	Transfer Taxes	27
	2.10	Other Closing Costs	28
		C	
3.	Closii	NG	28
	3.1	Closing	28
	3.2	Closing Deliveries	28
4.	Condi	TIONS TO CLOSING	29
	4.1	Conditions to the Obligations of Purchaser and Seller	29
	4.2	Conditions to the Obligations of Purchaser	30
	4.3	Conditions to the Obligations of Seller	30
5.	Denne	SENTATIONS AND WARRANTIES OF SELLER	31
J.	KEPRE	SENTATIONS AND WARRANTIES OF SELLER	31
	5.1	Organization	31
	5.2	Authority; Binding Effect	31
	5.3	Non-Contravention	31
	5.4	Financial Statements; Undisclosed Liabilities	32
	5.5	Inventory	32
	5.6	Taxes	33
	5.7	Absence of Changes	33
	5.8	No Litigation	34
	5.9	Compliance with Laws	34
	5.10	Governmental Authorizations	34
	5.11	Specified Assumed Contracts	34
	5.12	Customers and Suppliers	36
	5.13	Transferred Know-How and Transferred Patents	37
	5.14	Title	38
	5.15	Assets Used in the Facility Operations	39
	5.16	Environmental Matters	39

i

	5.17	Benefit Plans	39
	5.18	Employee and Labor Matters	40
	5.19	Real Estate.	42
	5.20	Data Privacy and Security.	44
	5.21	U.S. Foreign Corrupt Practices Act	44
	5.22	Trade Control Laws	45
	5.23	Brokers	45
6.	REPRE	SENTATIONS AND WARRANTIES OF PURCHASER	45
	6.1	Organization	45
	6.2	Authority; Binding Effect	45
	6.3	Non-Contravention	46
	6.4	No Litigation	46
	6.5	Financial Capability	46
	6.6	Guarantee	47
	6.7	Solvency	47
	6.8	Brokers	47
	6.9	Board Approval	47
7.	COVEN	IANTS	48
	7.1	Information and Documents; Notifications	48
	7.2	Conduct of Facility Operations	48
	7.3	Consents	50
	7.4	Employees and Employee Benefits	51
	7.5	Bulk Transfer Laws	54
	7.6	Litigation Support; Election to Control Non-Indemnified Claims	54
	7.7	Trade Notification	55
	7.8	Further Assurances	55
	7.9	Payments To or From Third Parties	56
	7.10	Excluded Records	56
	7.11	Insurance	56
	7.12	Tax Matters	56
	7.13	Assistance with Financial Reporting	57
	7.14	Shared Contracts	57
	7.15	Use of Trademarks	58
	7.16	Representation and Warranty Insurance	58
	7.17	Exclusive Dealings	58
	7.18	Title Insurance; Due Diligence	59
	7.19	Non-Competition and Non-Solicitation of Business Contacts	59
0	C	•	(0
8.	SURVIV	VAL AND INDEMNIFICATION	60
	8.1	Survival	60
	8.2	Indemnification by Seller	61
	8.3	Indemnification by Purchaser	61
	8.4	Claim Procedures	61
	8.5	Certain Limitations	63
	8.6	Losses Net of Insurance, Etc.	63
	8.7	Sole Remedy/Waiver	64

	8.8	Fraud	64
	8.9	No Special Damages	64
9.	Termin	ATION	65
	9.1	Termination	65
	9.2	Effect of Termination	65
10.	Miscel	LANEOUS	67
	10.1	Notices	67
	10.2	Amendment; Waiver	68
	10.3	Assignment	68
	10.4	Access to Records After Closing	68
	10.5	Nature of Information	69
	10.6	Entire Agreement	70
	10.7	Fulfillment of Obligations	70
	10.8	Parties in Interest	70
	10.9	Public Disclosure	70
		Expenses	70
	10.11	Interpretation; Disclosure Letter	71
		Governing Law; Jurisdiction; No Jury Trial	71
		Disclaimer of Warranties	72
	10.14	Specific Performance	72
	10.15	Non-Recourse	73
		Release	73
		Counterparts; Exchange by Electronic Transmission	73
	10.18	Severability	73
EXHIBI	TS		
Exhibit A		Equity Commitment Letter	
Exhibit 1		Limited Guarantee	
Exhibit (Rollover Agreement	
Exhibit 1		Accounting Principles	
Exhibit 1		Bill of Sale, Assignment and Assumption Agreement	
Exhibit 1		Inventory Valuation Principles	
Exhibit		License Agreement	
Exhibit 1		Manufacturing and Supply Agreement	
Exhibit 1		Mutual Transition Services Agreement	
Exhibit.		Patent Assignment Agreement	
Exhibit 1		Statutory Warranty Deed	
Exhibit 1		LLC Agreement	
Exhibit 1	M	Subscription Agreement	

SCHEDULES

Schedule 1.1(a)	Facility Contractors
Schedule 1.1(b)	Facility Employees
Schedule 1.1(c)	Permitted Liens
Schedule 1.1(d)	Permitted Encumbrances
Schedule 1.1(e)	Shared Services
Schedule 2.1(a)	Facility
Schedule 2.1(b)	Excluded Tangible Personal Property
Schedule 2.1(c)	Excluded Inventory
Schedule 2.1(f)	Transferred Patents
Schedule 2.1(g)	Transferred Business Systems
Schedule 2.1(1)	Purchased Accounts Receivable
Schedule 2.2(c)	Approvals
Schedule 2.3(d)	Excluded Contracts
Schedule 2.3(e)	Intercompany Contracts
Schedule 2.3(f)	Shared Contracts
Schedule 2.3(1)	Excluded Intellectual Property
Schedule 2.4(b)	Assumed Accounts Payable
Schedule 2.4(d)	Assumed Indebtedness
Schedule 7.2	Interim Facility Operations
	· -

ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement (this "Agreement") is made and entered into as of the 1st day of November, 2024, by and between Nektar Therapeutics, a Delaware corporation ("Seller"), and [* * *], a Delaware corporation ("Purchaser"). Seller and Purchaser are each individually referred to herein as a "Party" and collectively, as the "Parties".

RECITALS

WHEREAS, Seller desires to sell to Purchaser, and Purchaser desires to purchase from Seller, the Purchased Assets, and Purchaser desires to assume the Assumed Liabilities, all on the terms and subject to the conditions set forth herein;

WHEREAS, contemporaneously with the execution of this Agreement, as a condition and inducement to the willingness of Seller to enter into this Agreement, Purchaser has received an equity commitment letter, in the form of Exhibit A (the "Equity Commitment Letter"), from Ampersand 2023 Limited Partnership, a Delaware limited partnership (the "Sponsor") to provide equity financing in an aggregate amount set forth therein, on the terms and subject to the conditions set forth therein (the "Equity Financing");

WHEREAS, contemporaneously with the execution of this Agreement, as a condition and inducement to the willingness of Seller to enter into this Agreement, Seller has received a guarantee, in the form of Exhibit B (the "Limited Guarantee"), from the Sponsor pursuant to which the Sponsor has agreed to guarantee Purchaser's payment obligations in connection with this Agreement subject to and in accordance with the terms set forth in this Agreement and the Limited Guarantee;

WHEREAS, contemporaneously with the execution of this Agreement, Seller has entered into a contribution and exchange agreement in the form attached hereto as Exhibit C (the "Rollover Agreement") with [* * *], a Delaware limited liability company ("Parent"), [* * *], a Delaware corporation ("Midco") and Purchaser, pursuant to which, among other things, as of immediately prior to the Closing, Seller will contribute and assign to Midco a portion of Purchased Assets with an aggregate value equal to the Rollover Amount (the "Rollover Assets"), in exchange for the issuance of the Common Stock, par value \$0.001, in an aggregate value equal to the Rollover Amount (the "Exchange Shares"), on the terms and subject to the conditions in the Rollover Agreement, and immediately thereafter, Seller will contribute and assign to Parent the Exchange Shares, in exchange for the issuance of Common Units of Parent with an aggregate value equal to the Rollover Amount on the terms and subject to the conditions in the Rollover Agreement; and

NOW, THEREFORE, in consideration of the foregoing and the mutual covenants and agreements contained herein, the Parties hereby agree as follows:

1. DEFINITIONS AND TERMS

1.1 **Definitions**. As used in this Agreement, the following terms shall have the meanings set forth or as referenced below:

"Accounting Principles" shall mean the policies, practices and procedures set forth on Exhibit D.

- "Accounts Payable" shall mean all accounts payable of Seller, regardless of when asserted, billed or imposed.
- "Accounts Receivable" shall mean all accounts receivable owed by any third party to Seller or any of its Affiliates, regardless of when asserted, billed or imposed.
- "Acquisition Proposal" means any inquiry, proposal or offer from any Person (other than the Purchaser or its Affiliates) concerning the direct or indirect sale, lease, exchange or other disposition of all or substantially all of the Facility or the Purchased Assets other than the Transactions.
- "Action" shall mean any action, suit, claim, complaint, litigation, investigation, audit, proceeding, charge, mediation, arbitration or other similar dispute.
- "Adjustment Amount" shall mean an amount, which may be expressed as a positive or negative number, equal to (a) (i) the Closing Net Working Capital Amount minus (ii) the Target Net Working Capital Amount plus (b) (i) the Closing Inventory Amount minus (ii) the Target Inventory Amount, minus (c) the Closing Indebtedness Amount.
- "Affiliate" shall mean, with respect to any Person, any other Person controlling, controlled by or under common control with such Person. For purposes of this definition and this Agreement, the term "control" (and correlative terms) shall mean the power, whether by contract, equity ownership or otherwise, to direct the policies or management of a Person.
 - "Agreement" shall have the meaning set forth in the preamble.
 - "Allocation" shall have the meaning set forth in Section 2.8(a).
 - "Amgen Easement" shall mean [* * *].
- "Ancillary Agreements" shall mean, collectively, the Bill of Sale, the Manufacturing and Supply Agreement, the Mutual Transition Services Agreement, the Patent Assignment Agreement, the License Agreement, the Rollover Agreement, the Equity Commitment Letter, and any other agreements and instruments executed and delivered in connection with the Transactions.
 - "Assumed Accounts Payable" shall have the meaning set forth in Section 2.4(b).
 - "Assumed Contracts" shall have the meaning set forth in Section 2.1(d).
 - "Assumed Indebtedness" shall have the meaning set forth in Section 2.4(d).
 - "Assumed Liabilities" shall have the meaning set forth in Section 2.4.
 - "Base Purchase Price" shall mean an amount equal to \$90,000,000.
- "Benefit Plan" shall mean each (a) "employee benefit plan," as defined in Section 3(3) of ERISA, whether or not subject to ERISA, and (b) equity or equity-based compensation, severance, change of control, retention, bonus, incentive, commission, deferred compensation, profit sharing, employment agreement, executive compensation plans, programs, agreements or arrangements, retirement, welfare, cafeteria, fringe, or vacation or paid-time off or other compensation or any other benefit plan, program, policy or agreement not described in (a) above, in each case that Seller or any of its Affiliates sponsors, contributes to, maintains, or provides benefits through, or has any obligation to contribute to or provide benefits through, if such plan or arrangements provides benefits to or otherwise covers any Facility Employee or Facility Contractor.

- "Bill of Sale" shall mean the Bill of Sale, Assignment and Assumption Agreement in the form of Exhibit E.
- "Business Area" shall have the meaning set forth in Section 7.19(a).
- "Business Day" shall mean any day other than: (a) a Saturday, Sunday or federal holiday; or (b) a day on which commercial banks in New York, New York, San Francisco, California or Huntsville, Alabama are authorized or required to be closed pursuant to applicable Law.
 - "Business Systems" shall have the meaning set forth in Section 5.13(a).
- "Cash and Investments" shall mean cash, checks, money orders, marketable securities, short-term instruments, deposits with third parties and other cash equivalents, funds in time and demand deposits or similar accounts, and any evidence of indebtedness issued or guaranteed by any Governmental Authority.
 - "Claim Release" shall have the meaning set forth in Section 10.16.
 - "Closing" shall mean the consummation of the Transactions pursuant to the terms of this Agreement.
 - "Closing Cash Payment" shall have the meaning set forth in Section 2.6.
 - "Closing Date" shall have the meaning set forth in Section 3.1(a).
 - "Closing Indebtedness Amount" shall have the meaning set forth in Section 2.7(b).
 - "Closing Inventory Amount" shall have the meaning set forth in Section 2.7(b).
 - "Closing Net Working Capital Amount" shall have the meaning set forth in Section 2.7(b).
 - "Closing Statement" shall have the meaning set forth in Section 2.7(b).
 - "Code" shall mean the Internal Revenue Code of 1986.
 - "Competitive Business" means [* * *].

"Confidential Information" means, with respect to a Person, such Person's trade secrets, processes, patent applications, product development, price, customer and supplier lists, pricing and marketing plans, policies and strategies, details of contracts, operations, methods, supplier information and relationships, product development techniques, business acquisition plans, new personnel acquisition plans and all other confidential or proprietary documents and information relating to the business and affairs of such Person, including any notes, analyses, compilations, studies, forecasts, interpretations or other documents that are derived from, contain, reflect or are based upon any such information and the organizational documents and other organizational records and information of such Person.

"Confidentiality Agreement" shall mean that certain confidentiality agreement dated on or about June 17, 2024, by and between Seller and Purchaser.

"Contract" shall mean any agreement, undertaking, lease, license, contract, note, mortgage, indenture, arrangement or other written obligation (other than any Benefit Plan).

"Copyrights" shall mean copyrights and other rights associated with works of authorship, whether or not copyrightable, and all registrations, applications, extensions and renewals in connection therewith.

"<u>Data Room</u>" shall mean the virtual data room hosted on behalf of Seller by Intralinks in the online workspace captioned "Project Nova" in connection with the Transactions.

"<u>Disclosure Letter</u>" shall mean the Disclosure Letter with numbered sections corresponding to sections of this Agreement delivered to Purchaser on behalf of Seller on the date hereof.

"Disputed Items" shall have the meaning set forth in Section 2.7(e).

"Encumbrances" shall mean, with respect to any property or asset, Liens, defects or irregularities in title, easements, rights-of-way, covenants, restrictions and other similar claim, restriction or limitation of any kind, character or description, whether of record or note, or any contract to give any of the foregoing, in respect of such property or asset.

"Environmental Law" shall mean any Law relating to (a) the protection of the environment or natural resources (including any Law imposing strict liability), (b) occupational health and safety (solely in respect of exposure to Hazardous Materials) or (c) the use, management, manufacture, storage, recycling, treatment, generation, transportation, remediation, cleanup, processing, handling, labeling, recycling, Release or disposal of, or exposure to, Hazardous Materials, including the federal Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. Sec. 9601 et seq. and any similar state Law and the Alabama Uniform Environmental Covenants Act, Ala. Code Secs. 35-19-1 et seq.

"Environmental Liabilities" shall mean all Liabilities and Losses arising under Environmental Laws, including Liabilities and Losses to the extent arising out of or resulting from (a) any non-compliance with, or violation of, any Environmental Law, (b) failure to obtain or comply with any required Environmental Permit, (c) any Release or threatened Release (including any remediation, cleanup, investigation, corrective action, reporting or monitoring) of Hazardous Materials, (d) any exposure to Hazardous Materials or (e) any disposal, transportation or arrangement for transportation of any Hazardous Material sent to any third party property.

"Environmental Permit" shall mean a permit, license, certificate, approval or authorization issued by a Governmental Authority pursuant to an Environmental Law.

"Equity Commitment Letter" shall have the meaning set forth in the Recitals.

"Equity Financing" shall have the meaning set forth in the Recitals.

- "Estimated Indebtedness Amount" shall have the meaning set forth in Section 2.7(a).
- "ERISA" shall mean the Employee Retirement Income Security Act of 1974.
- "ERISA Affiliate" means any entity, trade or business that is, or at any applicable time was, a member of a group described in Section 414(b), (c), (m) or (o) of the Code or Section 4001(b)(1) of ERISA that includes Seller.
- "Estimated Adjustment Amount" shall mean an amount, which may be expressed as a positive or negative number, equal to (a) (i) the Estimated Net Working Capital Amount minus (ii) the Target Net Working Capital Amount, plus (b) (i) the Estimated Inventory Amount minus (ii) the Target Inventory Amount, minus (c) the Estimated Indebtedness Amount.
 - "Estimated Closing Statement" shall have the meaning set forth in Section 2.7(a).
 - "Estimated Inventory Amount" shall have the meaning set forth in Section 2.7(a).
 - "Estimated Net Working Capital Amount" shall have the meaning set forth in Section 2.7(a).
- "Estimated Purchase Price" shall mean an amount equal to (a) the Base Purchase Price <u>plus</u> (b) the Estimated Adjustment Amount, <u>minus</u> (c) the Estimated Seller Transfer Taxes.
 - "Exchange Shares" shall have the meaning set forth in the Recitals.
 - "Excluded Assets" shall have the meaning set forth in Section 2.3.
 - "Excluded Business Systems" shall have the meaning set forth in Section 2.3(m).
 - "Excluded Contracts" shall have the meaning set forth in Section 2.3(d).
 - "Excluded Inventory" shall mean the PEG Products and Raw Materials set forth on Schedule 2.1(c).
 - "Excluded Liabilities" shall have the meaning set forth in Section 2.5.
 - "Excluded Records" shall have the meaning set forth in Section 2.3(h).
 - "Facility" shall have the meaning set forth in Section 2.1(a).
 - "Facility Books, Documents and Records" shall have the meaning set forth in Section 2.1(i).

"Facility Contractors" shall mean the independent contractors, staffing agencies or other service providers engaged by or on behalf of Seller or its Affiliates to provide services exclusively in connection with the Facility Operations pursuant to the agreements, contracts or other arrangements listed on Schedule 1.1(a), which shall be determined by Seller in good-faith consultation with Purchaser and updated, as needed with the consent of Purchaser, from time to time prior to the Closing Date (it being understood that Purchaser's consent shall not be required to the extent that any updates to Schedule 1.1(a) are a result of any engagement, or termination, of any Facility Contractor that is expressly permitted by Section 7.2(a) or for which Purchaser has otherwise given its prior written consent). Seller shall provide a final Schedule 1.1(a) at least five Business Days prior to the Closing Date.

"Facility Employees" shall mean the employees of Seller or its Affiliates (including temporary employees hired through a staffing agency) listed on Schedule 1.1(b), which shall be determined by Seller in good-faith consultation with Purchaser and updated, as needed with the consent of Purchaser, from time to time prior to the Closing Date (it being understood that Purchaser's consent shall not be required to the extent that any updates to Schedule 1.1(b) are a result of any engagement, or termination, of any Facility Employee that is expressly permitted by Section 7.2(a) or for which Purchaser has otherwise given its prior written consent). Seller shall provide a final Schedule 1.1(b) to Purchaser at least five Business Days prior to the date that Purchaser is required to make offers of employment to any Facility Employees pursuant to Section 7.4(a).

"Facility Operations" shall mean [* * *].

"Facility Personal Property" shall have the meaning set forth in Section 2.1(a).

"Facility Value" shall have the meaning set forth in Section 2.8(b)(i).

"Final Adjustment Amount" shall have the meaning set forth in Section 2.7(g).

"FCPA" shall have the meaning set forth in Section 5.21.

"FIRPTA" shall have the meaning set forth in Section 5.19(f).

"Fraud" shall mean fraud within the meaning of Delaware common law by or on behalf of a Party regarding the representations and warranties expressly made by such Party herein or any certificate or schedule delivered pursuant to this Agreement by such Party.

"GAAP" shall mean accounting principles and practices generally accepted in the United States of America.

"Good Manufacturing Practices" shall mean the requirements for the quality system under which drug products and their ingredients are manufactured, including the requirements set forth in the GMP rules of the World Health Organisation, the current Good Manufacturing Regulations set forth in the United States Code of Federal Regulations (Title 21, Parts 210-211, Parts 600 et seq. and Parts 808, 812 and 820), the European Union Guide to Good Manufacturing Practice including Annexes in the Production of Pharmaceutical Products, and other guidelines and regulations of any applicable Governmental Authority.

"Governmental Authority" shall mean any supranational, national, federal, state or local judicial, legislative, executive or regulatory authority.

"Governmental Authorizations" shall mean any licenses, permits, certificates or other authorization or approval under any applicable Law issued by a Governmental Authority.

"Governmental Order" shall mean any order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority.

"Hazardous Material" shall mean any substance, material, or waste that is listed, classified or regulated as hazardous or toxic, or as a pollutant or contaminant pursuant to Environmental Law, or that is otherwise regulated by or for which liability or standards of care are imposed under Environmental Law, and includes oils, petroleum, petroleum products or byproducts, radioactive materials, asbestos and asbestos-containing materials, polychlorinated biphenyls, regulated medical or infectious waste, heavy metals, chlorinated solvents, urea formaldehyde foam insulation, mold that would reasonably be expected to have a material adverse impact on human health or property and per- or polyfluoroalkyl substances.

"Indebtedness" means, as of any date and time, without duplication, (a) all indebtedness for borrowed money, together with accrued and unpaid interest thereon, as of such date and time, including obligations evidenced by notes, bonds, debentures or similar instruments, (b) all capital lease obligations as of such date and time (which are required to be classified as capital leases under GAAP), (c) all reimbursement obligations as of such date and time under letters of credit to the extent such letters of credit have been drawn, (d) all unpaid property or similar Taxes attributable to the Purchased Assets for any Pre-Closing Tax Period (determined with respect to a Straddle Period in accordance with Section 7.12(d)) that are payable by Purchaser or its Affiliates after the Closing, (e) liabilities and obligations under any futures contracts, interest rate and currency swaps, caps, collars and similar agreements or hedging devices (each valued at the termination value thereof), (f) all liabilities and obligations as of such date and time with respect to accrued but unpaid vacation or paid time off, any accrued but unpaid annual or other bonus or commission obligations, phantom stock or equity or similar arrangements (if any), defined benefit pension, multiemployer pension, post-retirement health and welfare benefit, any unpaid severance liabilities, retention bonuses and deferred compensation liabilities (including in each case any associated withholding Taxes and/or the employer portion of any payroll, social security, unemployment, employment or similar Taxes required to be paid in connection therewith), (g) any payroll Taxes that have been deferred under the CARES Act or IRS Notice 2020-65 (or any similar provision of federal, state, local or non-U.S. Law), (h) all obligations for the deferred or unpaid purchase price of assets, property, goods or services (other than trade payables or accruals incurred in the ordinary course of business that are not past due), (i) all liabilities for redemption premium, prepayment penalty, change of control premium, break fee, termination fee or similar payment in connection with clauses (a) through (h) of this definition, and (j) all obligations in the nature of guarantees of the obligations described in clauses (a) through (h) of this definition; provided, however, that "Indebtedness" shall not include (i) letters of credit (to the extent undrawn) and reimbursement obligations in respect of letters of credit (to the extent undrawn), (ii) any liabilities related to inter-company debt that are both not secured by any of the Purchased Assets and are Excluded Liabilities or (iii) any amounts included in the calculations of the Estimated Net Working Capital Amount or the Closing Net Working Capital Amount.

"Indemnified Party" shall have the meaning set forth in Section 8.4(a).

"Indemnifying Party" shall have the meaning set forth in Section 8.4(a).

"Independent Accountant" shall have the meaning set forth in Section 2.7(e).

"Intellectual Property" shall mean all intellectual property rights and related priority rights protected, created or arising under the Laws of the United States or any other jurisdiction or under any international convention, including all Patents, Trademarks, Copyrights, Trade Secrets, all other intellectual property, industrial, or similar rights, and all rights to seek, recover, and retain damages, costs, profits, injunctive relief and other remedies for any past or future infringement, violation, misappropriation or other impairment of any of the items set forth in above and the right to receive all proceeds and damages therefrom.

"Intercompany Contract" shall have the meaning set forth in Section 2.3(e).

"Intracompany Payables" shall mean all account, note or loan payables recorded on the books of Seller or any of its Affiliates for goods and services used in connection with the Facility Operations, or advances (cash or otherwise) or any other extension of credit in respect of the Facility Operations, whether current or non-current.

"Inventory" shall have the meaning set forth in Section 2.1(c).

"Inventory Valuation Principles" shall mean the policies, practices and procedures set forth on Exhibit F.

"IRS" shall mean the United States Internal Revenue Service.

"IT Assets" means systems, servers, computers, hardware, firmware, middleware, networks, data communications lines, routers, hubs, switches and all other information technology equipment, and all software and associated documentation related thereto.

"Know-How" shall mean any proprietary scientific or technical information, results, and data of any type whatsoever, including know-how, data, databases, technology, safety information, practices, designs, processes, methods, techniques, processes, formulations, formulae, inventions, improvements, developments, materials, algorithms, knowledge, test data, analytical and quality control data, stability data, studies and procedures, manufacturing process and development information, marketing reports, regulatory submissions, expertise, product specifications and packaging specifications, in each case whether or not confidential, proprietary, patentable and in written, electronic or any other form, except that Know-How shall exclude Patents.

"Knowledge of Seller" shall mean, as to a particular matter, the current actual knowledge of each of [* * *], in each case, after reasonable inquiry.

"Laws" shall include any federal, state, foreign or local law, common law, statute, ordinance, rule, regulation, code or Governmental Order.

"Liabilities" shall mean any and all debts, liabilities and obligations, whether accrued or fixed, known or unknown, absolute or contingent, matured or unmatured or determined or determinable.

"License Agreement" shall mean the License Agreement in the form of Exhibit G.

"Liens" shall mean any lien, security interest, mortgage, charge, deed of trust, pledge, hypothecation, encumbrance, security interest, easement, assignment, title retention agreement or deposit, license, sublicence, preference, priority, restriction, negative pledge, option, warrant, commitment, demand, collateral, conditional sale, encumbrance, security arrangement or agreement, any restriction on transfer or disposition, or any filing or agreement to file a financing statement as debtor under the Uniform Commercial Code or any similar statute, or other lien or encumbrance of any kind.

"Limited Guarantee" shall have the meaning set forth in the Recitals.

"Losses" shall mean any claims, actions, causes of action, judgments, awards, liabilities, losses, costs (including reasonable out-of-pocket attorney's fees and other documented out-of-pocket costs incurred in investigating, preparing and defending the foregoing) or damages.

"Manufacturing and Supply Agreement" shall mean the Master Manufacturing and Supply Agreement in the form of Exhibit H.

"Material Adverse Effect" shall mean any change event, circumstance, development, occurrence or effect (each, an "Effect") that, individually or taken together with any other Effect, has a material adverse effect on the Facility Operations, taken as a whole; provided, however, that none of the following shall be deemed, either alone or in combination, to constitute, and no Effect arising from or attributable or relating to any of the following shall be taken into account in determining whether there has been, a Material Adverse Effect: (a) the negotiation, execution, announcement or performance of this Agreement or any of the Transactions, or the consummation of the Transactions, including the impact thereof on the relationships of Seller with suppliers, vendors, consultants, employees or independent contractors or other third parties with whom Seller has any relationship; (b) general economic, political, regulatory or other conditions or securities, credit, financial or capital markets conditions (including interest rates, exchange rates, the value of any currency or any fluctuations in any of the foregoing and uncertainty in the banking industry); (c) the taking (or not taking) of any action expressly required by this Agreement to be taken (or not taken); (d) any action taken or not taken by Seller with the express written consent of Purchaser or at the express written request of Purchaser; (e) any Effect to the extent proximately caused by any breach by Purchaser of this Agreement or the Confidentiality Agreement; (f) any development or change in GAAP or other applicable accounting standards (or interpretation thereof) after the date hereof or any adoption, promulgation, modification or reinterpretation of any Law, or the enforcement thereof, that occurs subsequent to the date hereof; (g) any acts of war or terrorism, armed hostilities or cyber-attacks or the worsening, continuation or escalation thereof; (h) any acts of God, earthquakes, wildfires, tornados, hurricanes, floods or other natural disasters or the worsening, continuation or escalation thereof; (i) any loss of Facility Employees or Facility Contractors; or (j) any item disclosed in the Disclosure Letter, except in the case of the foregoing clauses (b), (f), (g) and (h), to the extent any such Effect has a material and disproportionate adverse effect on the Facility Operations relative to the operations of other Persons engaged in the same or a similar industry as the Facility Operations.

"Material Customers" shall have the meaning set forth in Section 5.12(a).

"Material Permits" shall have the meaning set forth in Section 5.10.

"Material Suppliers" shall have the meaning set forth in Section 5.12(b).

"Multiemployer Plan" means an employee pension or welfare benefit plan to which more than one unaffiliated employer contributes and which is maintained pursuant to one or more collective bargaining agreements.

"Mutual Transition Services Agreement" shall mean the Mutual Transition Services Agreement in the form of Exhibit I.

"Net Working Capital" shall mean an amount equal to (a) the current assets to the extent arising or held in connection with the Facility Operations (but only to the extent included in the Purchased Assets), minus (b) the reserves and current Liabilities to the extent arising or held in connection with the Facility Operations (but only to the extent included in the Assumed Liabilities), in each case, as of the Reference Time and calculated in accordance with the Accounting Principles and as set forth as line items on Exhibit D; provided, however, that the computation of Net Working Capital shall not include any Taxes. For purposes of calculating Net Working Capital, (i) current assets shall exclude all Cash and Investments and Inventory and (ii) current liabilities shall exclude all Indebtedness.

"Non-Indemnified Claim" shall have the meaning set forth in Section 7.6(b).

"Nonassigned Asset" shall have the meaning set forth in Section 2.2(b).

"Objection Notice" shall have the meaning set forth in Section 2.7(d).

"OFAC" shall have the meaning set forth in Section 5.19(m).

"Patent Assignment Agreement" shall mean the Patent Assignment Agreement in the form of Exhibit J.

"Patents" shall mean patents, provisional patent applications and patent applications, and continuations, continuations-in-part, divisions, reissues, renewals, extensions and reexaminations thereto, and all similar rights in any part of the world.

```
"PEG" shall mean [* * *].

"PEG Product" shall mean [* * *].

"PEG Reagent" shall mean [* * *].
```

"Permitted Encumbrances" shall mean (a) statutory Liens arising out of operation of Law with respect to a Liability incurred in the ordinary course of business and which is not delinquent and for which adequate reserves have been established and maintained in accordance with GAAP; (b) Liens for Taxes not yet due or subject to penalties for nonpayment or are being contested in good faith by appropriate proceedings and for which adequate reserves have been established and maintained in accordance with GAAP; (c) any Lien set forth on Schedule 1.1(c); (d) any Encumbrance created in connection with or pursuant to any Assumed Liabilities and which is disclosed on Schedule 1.1(d); (e) mechanics', materialmens', carriers', workmens', warehousemens', repairmens', landlords' or other like Liens and security obligations that are incurred in the ordinary course of business and are not delinquent or are being contested in good faith and for which adequate reserves have been established and maintained in accordance with GAAP; (f) rights relating to any utility wires, sewers, poles, pipes, conduits and appurtenances thereto, on, under or across the Facility that, individually and in the aggregate, do not and would not reasonably be expected to materially detract from the value or impair the use of the property subject thereto or make such property unmarketable; (g) minor variations between tax lot liens and lines of record title that, individually and in the aggregate, do not and would not reasonably be expected to materially detract from the value or impair the use of the property subject thereto or make such property unmarketable, (h) all applicable Laws, including building, environmental and zoning laws, ordinances and regulations, now or hereafter in existence; (i) any defects or imperfections of title, easements, encroachments, conditions, covenants and other restrictions or Encumbrances, including easements, rights of way, options, reservations or other similar matters or restrictions or exclusions that would be disclosed on current title reports or accurate surveys or during physical inspections, and any Encumbrances of public record that, individually and in the aggregate, do not and would not reasonably be expected to materially detract from the value or impair the use of the property subject thereto or make such property unmarketable; (j) Liens granted to any lender at the Closing in connection with any financing by Purchaser of the Transactions or created by this Agreement or any Ancillary Agreement, whether prior to, at or after the Closing; (k) Liens with respect to real property created by or resulting from the acts or omissions of Purchaser or any of its Affiliates; (1) nonexclusive licenses of Intellectual Property to vendors, manufacturers, suppliers, distributors, or other Persons performing manufacturing, supply, marketing or other services on behalf of Seller or any of its Subsidiaries, in each case, (i) that are granted in the ordinary course of business or (ii) where the grant of rights under any Intellectual Property are incidental, and not material to, any performance under an agreement with such third party, and (m) any other Encumbrance arising out of the acts or omissions of Purchaser, its Affiliates or their respective representatives.

"Person" shall mean an individual, a limited liability company, a joint venture, a corporation, a partnership, an association, a trust, a division or operating group of any of the foregoing or other entity or organization.

"Personal Information" shall mean "personal information" or any similar term (e.g., "personal data" or "personally identifiable information") provided by Privacy Laws or information that directly or indirectly identifies or would reasonably be linked with a particular individual, device or household.

"Post-Closing Tax Period" shall mean (a) any taxable period beginning after the Closing Date and (b) the portion of any Straddle Period beginning immediately after the Closing Date and ending on the last day of such Straddle Period.

"Pre-Closing Tax Period" shall mean (a) any taxable period ending at or before the close of the Closing Date and (b) the portion of any Straddle Period beginning on the first day of such Straddle Period and ending at the close of the Closing Date.

"Prepaid Expenses" shall have the meaning set forth in Section 2.1(k).

"Privacy Laws" shall mean any applicable Laws that govern the receipt, collection, compilation, use, storage, processing, sharing, safeguarding, security, disposal, destruction, disclosure or transfer of Personal Information, including, for the avoidance of doubt, to the extent applicable, all U.S. state and federal Laws concerning the privacy and/or security of Personal Information, all privacy Laws regulating biometric information, the California Invasion of Privacy Act and all other Laws concerning the interception and/or recording of communications, the CAN-SPAM Act, the Telephone Consumer Protection Act and all other Laws regarding marketing communications, the Federal Trade Commission Act and all analogous Laws in foreign jurisdictions in which the Facility Operations are conducted or in which Seller is otherwise subject to the legal jurisdiction of such Laws in respect of the Facility Operations.

"Privacy Requirements" shall have the meaning set forth in Section 5.20(a).

"Proposed Allocation" shall have the meaning set forth in Section 2.8(b)(ii).

"Purchase Price" shall have the meaning set forth in Section 2.6.

"Purchased Accounts Receivable" shall have the meaning set forth in Section 2.1(1).

- "Purchased Assets" shall have the meaning set forth in Section 2.1.
- "Purchaser" shall have the meaning set forth in the preamble.
- "Purchaser Benefit Plans" shall have the meaning set forth in Section 7.4(e).
- "Purchaser DC Plans" shall have the meaning set forth in Section 7.4(c).
- "Purchaser FSA Plans" shall have the meaning set forth in Section 7.4(h).
- "<u>Purchaser Fundamental Representations</u>" shall mean the representations and warranties of Purchaser set forth in <u>Section 6.1</u> (Organization), <u>Section 6.2</u> (Authority; Binding Effect), <u>Section 6.3</u> (Non-Contravention), <u>Section 6.4</u> (No Litigation), <u>Section 6.5</u> (Financial Capability) and <u>Section 6.8</u> (Brokers).
- "<u>Purchaser Material Adverse Effect</u>" shall mean any Effect that, individually or taken together with any other Effect, prevents, materially delays or materially impairs, or would reasonably be expected to prevent, materially delay, or materially impair, the ability of Purchaser to consummate the Transactions or timely perform its obligations hereunder or under any of the Ancillary Agreements.
 - "Purchaser Welfare Plans" shall have the meaning set forth in Section 7.4(f).
 - "Raw Materials" shall mean [* * *].
 - "Recorded Environmental Covenant" means [* * *].
- "Recorded Environmental Covenant Notice" means the notice required by Section 3.B of the Recorded Environmental Covenant of the transfer of the Facility contemplated by this Agreement.
 - "Reference Time" shall mean 11:59 p.m. Pacific Time on the date before the Closing Date.
- "Release" shall mean any release, spill, emission, leaking, pumping, pouring, emptying, escape, injection, deposit, disposal, discharge, dispersal, dumping, leaching or migration of Hazardous Materials into or through the indoor or outdoor environment, including the movement of Hazardous Materials through or in the air (including indoor air), soil, soil vapor, sediment, surface water, groundwater, or surface or subsurface strata.
 - "Released Claims" shall have the meaning set forth in Section 10.16.
 - "Released Parties" shall have the meaning set forth in Section 10.16.
 - "Restriction Period" shall have the meaning set forth in Section 7.19.
 - "Retained Names and Marks" shall have the meaning set forth in Section 7.15(a).

- "Review Period" shall have the meaning set forth in Section 2.7(d).
- "Rollover Agreement" shall have the meaning set forth in the Recitals.
- "Rollover Amount" shall mean an amount equal to \$20,000,000.
- "Rollover Assets" shall have the meaning set forth in the Recitals.
- "RWI Policy" shall mean the representations and warranties insurance policy obtained by and issued to Purchaser (or one of its Affiliates) by the insurer thereunder in connection with this Agreement.
 - "SDN List" shall have the meaning set forth in Section 5.19(m).
 - "Securities Act" shall mean the Securities Act of 1933.
 - "Seller" shall have the meaning set forth in the preamble.
 - "Seller DC Plans" shall have the meaning set forth in the Section 7.4(c).
 - "Seller FSA Plans" shall have the meaning set forth in Section 7.4(h).
- "Seller Fundamental Representations" shall mean the representations and warranties of Seller set forth in Section 5.1 (Organization), Section 5.2 (Authority; Binding Effect), clause (i) and (iv) of Section 5.3(a) (Non-Contravention), Section 5.14 (Title) and Section 5.23 (Brokers).
- "Seller IT Assets" shall mean any and all IT Assets owned, leased by or licensed to Seller and used or held for use by Seller in connection with the Facility Operations as currently conducted.
- "Seller Transaction Expenses" shall mean (a) all third party out-of-pocket fees, costs and expenses (including all fees and expenses of counsel, accountants, investment bankers, experts and consultants to Seller or any of its Affiliates), incurred by Seller and its Affiliates in connection with or related to the authorization, preparation, negotiation, execution and performance of this Agreement and the Ancillary Agreements, and (b) any transaction, change in control, retention, severance or similar bonuses or payments payable by Seller or any of its Affiliates to any Facility Employees as a result of, or in connection with, the consummation of the Transactions, including any such payments pursuant to arrangements established by Seller or any of its Affiliates on or prior to Closing (including the employer portion of any payroll, social security, unemployment, employment or similar Taxes required to be paid in connection therewith, calculated, to the extent the Closing occurs prior to December 31, 2024, by assuming that any applicable social security or other wage base has already been exceeded), but excluding any arrangements established by Seller at the express written direction of Purchaser.
 - "Settlement Agreement" shall mean [* * *].
 - "Shared Contracts" shall have the meaning set forth in Section 2.3(f).
- "Shared Services" shall mean the shared services and systems provided by Seller or any of its Affiliates, or on their behalf, in connection with the Facility Operations and which are listed on Schedule 1.1(e).

"Software" shall mean (a) computer software programs, including all software implementations of algorithms, modes and methodologies, (b) program files, data files, computer-related data, field and data definitions and relationships, data definition specifications, data models, program and system logic, interfaces, program modules, routines, subroutines, algorithms, program architecture, design concepts, system designs, program structure, sequence and organization, screen displays and report layouts, (c) descriptions, flow charts and other work product used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons, and (d) all documentation including user manuals and other training documentation related to any of the foregoing, and any improvements, updates, upgrades or derivative works of any of the foregoing.

- "Specifically Excluded Records" shall have the meaning set forth in Section 2.3(h).
- "Specified Obligations" shall have the meaning set forth in Section 10.16.
- "Sponsor" shall have the meaning set forth in the Recitals.
- "Statutory Warranty Deed" shall have the meaning set forth in Section 3.2(a)(iii).
- "Straddle Period" shall mean any taxable period beginning on or before and ending after the Closing Date.
- "Subscription Agreement" shall have the meaning set forth in Section 3.2(b)(iii).

"Subsidiary" shall mean, with respect to any Person, any other Person of which at least a majority of the securities or ownership interests having by their terms ordinary voting power to elect a majority of the board of directors or other Persons performing similar functions is directly owned or controlled by such Person or by one or more of its Subsidiaries.

- "Target Inventory Amount" shall mean an amount equal to \$[* * *].
- "Target Net Working Capital Amount" shall mean an amount equal to \$[* * *].

"<u>Tax Return</u>" shall mean any return, report, declaration, information return, statement or other document filed or required to be filed with any Taxing Authority in connection with the determination, assessment or collection of any Tax or the administration of any Laws relating to any Tax.

"Taxes" shall mean all taxes, charges, duties, customs, fees, levies or other assessments, including income, excise, property, sales or use, rental, value added, profits, license, business license, withholding (with respect to compensation or otherwise), payroll, employment, net worth, capital gains, transfer, stamp, social security, environmental, occupation and franchise taxes, whether disputed or not, imposed by any Taxing Authority, and including any interest, penalties and additions attributable thereto.

"<u>Taxing Authority</u>" shall mean any Governmental Authority exercising any authority to impose, regulate or administer the imposition of Taxes (including any third party firms authorized to act as their designee).

"Termination Date" shall have the meaning set forth in Section 9.1(d).

- "Third Party Claim" shall have the meaning set forth in Section 8.4(b).
- "Third Party Claim Notice" shall have the meaning set forth in Section 8.4(b)(i).
- "Title Company" shall mean First American Title Insurance Company.
- "<u>Title Policy</u>" shall mean a standard American Land Title Association owner's title insurance policy for the Facility issued by the Title Company subject to Permitted Encumbrances, using the current policy jacket customarily provided by the Title Company in the State of Alabama.
 - "Trade Approvals" shall have the meaning set forth in Section 5.22(a).
 - "Trade Laws" shall have the meaning set forth in Section 5.22(a).
- "<u>Trade Secrets</u>" shall mean trade secrets, Know-How, and confidential and proprietary information, in each case, that derive independent economic value, actual or potential, from not being generally known or readily ascertainable by others.
- "Trademarks" shall mean registered or unregistered trademarks, service marks, trade dress, trade names, logos, corporate names, internet domain names, social media accounts, and all applications, registrations and renewals in connection therewith, together with all goodwill associated with the foregoing.
 - "Transactions" shall mean the transactions contemplated by this Agreement and the Ancillary Agreements.
- "Transfer Taxes" shall mean any federal, state, county, local, foreign and other sales, use, transfer, value added, conveyance, documentary transfer, stamp duty, recording or other similar tax, fee or charge imposed in connection with the Transactions or the recording of any sale, transfer or assignment of property (or any interest therein) effected pursuant to this Agreement.
 - "Transferred Business Systems" shall have the meaning set forth in Section 2.1(g).
 - "Transferred Employee" shall have the meaning set forth in Section 7.4(a).
 - "Transferred Governmental Authorizations" shall have the meaning set forth in Section 2.1(e).
- "Transferred Know-How" shall mean the Know-How exclusively (a) relating to the operation of the equipment located on or at the Facility or (b) used or held for use in the manufacture of PEG Products. For clarity, Transferred Know-How excludes Know-How that: (i) relates generally to the operation of other equipment not located on or at the Facility, (ii) relates to the manufacture of compounds or products (other than PEG Products), (iii) relates to warehousing for Seller or any of its Affiliates, (iv) relates to the conduct of quality testing services for Seller or any of its Affiliates or (v) is licensed to Purchaser pursuant to the License Agreement for the Manufacture (as defined in the License Agreement) of PEG Products for the Licensed Field (as defined in the License Agreement).
 - "Transferred Patents" shall mean the Patents set forth on Schedule 2.1(f).
 - "WARN Act" shall have the meaning set forth in Section 5.18(f).

1.2 Other Definitional Provisions.

- (a) The table of contents and headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. Where a reference in this Agreement is made to an Exhibit, Article, Section or Schedule, such reference shall be to an Exhibit, Article, Section or Schedule of this Agreement unless otherwise indicated.
- (b) If a term is defined as one part of speech (such as a noun), it shall have a corresponding meaning when used as another part of speech (such as a verb). The rule known as the *ejusdem generis* rule shall not apply, and, accordingly, general words introduced by the word "other" shall not be given a restrictive meaning by reason of the fact that they are preceded by words indicating a particular class of acts, matters or things. The words "hereof," "herein" and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement. The word "or" shall not be exclusive. Any capitalized term used in any Schedule or Exhibit but not otherwise defined therein shall have the meaning given to them as set forth in this Agreement. All accounting terms used herein and not expressly defined herein shall have the meanings given to them under GAAP. References to "written" or "in writing" include documents in electronic form or transmission by email.
- (c) The terms defined in the singular shall have a comparable meaning when used in the plural, and vice versa, and words denoting any gender include all genders.
 - (d) The words "dollars" and "\$" shall mean United States dollars.
 - (e) The words "including" shall mean "including, without limitation."
- (f) Except as otherwise specifically provided herein, all references in this Agreement to any Law include the rules and regulations promulgated thereunder, in each case as amended, re-enacted, consolidated or replaced from time to time and in the case of any such amendment, re-enactment, consolidation or replacement, reference herein to a particular provision shall be read as referring to such amended, re-enacted, consolidated or replaced provision and shall also include, unless the context otherwise requires, all applicable guidelines, bulletins or policies made in connection therewith; provided, that for purposes of any representations and warranties set forth in this Agreement that are made as of a specific date, references to any Law shall be deemed to refer to such Law as amended as of such date. All references in this Agreement to any Contract shall include such Contract, as amended or modified from time to time.
- (g) Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. Whenever any action must be taken hereunder on or by a day that is not a Business Day, then such action may be validly taken on or by the next day that is a Business Day. When calculating the period of time before which, within which or following which an act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.

- (h) The phrase "made available" shall mean that a copy of the information or documents has been provided to Purchaser in the Data Room at least one Business Day prior to the date hereof or has been provided directly to Purchaser or its representatives (including by email) at least one Business Day prior to the date hereof.
- (i) Each representation or warranty of Seller in this Agreement is made only with respect to the Purchased Assets or the Facility Operations and, notwithstanding anything in this Agreement to the contrary, Seller is not making any representation with respect to any other business or activities conducted by Seller or any of its Affiliates.

2. PURCHASE AND SALE

- **2.1 Purchase and Sale of Assets**. Upon the terms and subject to the conditions set forth herein, at the Closing, Seller shall sell, convey, assign and transfer to Purchaser, and Purchaser shall purchase, acquire and accept from Seller, free and clear of all Liens, other than Permitted Encumbrances, all the right, title and interest of Seller in, to or under the assets set forth below, in each case, other than the Rollover Assets and the Excluded Assets (collectively, the "Purchased Assets"):
 - (a) the real property located in Huntsville, Alabama with a legal description set forth on <u>Schedule 2.1(a)</u>, together with Seller's right, title and interest in, to and under all buildings, plants, facilities, structures, improvements and fixtures thereon, all easements and rights of way pertaining thereto or accruing to the benefit thereof and all other appurtenances pertaining thereto (collectively, the "<u>Facility</u>");
 - (b) all tangible personal property (including machinery, equipment, appliances, furniture, vehicles, computer hardware, tools, accessories, change parts, spare parts and assets under construction) and interests therein located on or at the Facility as of the Closing (the "<u>Facility Personal Property</u>"), in each case, other than any tangible personal property set forth on <u>Schedule 2.1(b)</u>;
 - (c) all PEG Products and Raw Materials located on or at the Facility as of the Closing, in each case, other than the Excluded Inventory (collectively, "Inventory,");
 - (d) other than any Excluded Contracts, subject to <u>Section 2.2</u>, all Contracts exclusively used in the conduct of the Facility Operations, including, for the avoidance of doubt, the Specified Assumed Contracts (the "<u>Assumed Contracts</u>");
 - (e) all Governmental Authorizations exclusively used in the conduct of the Facility Operations, to the extent transferable without violating the terms thereof or applicable Law (collectively, the "<u>Transferred Governmental Authorizations</u>");
 - (f) the Transferred Patents and Transferred Know-How, including all claims, causes of action, remedies and rights to sue for and collect damages for past, present and future infringement, misappropriation, or other violation or impairment thereof and all royalties, fees, income, payments, and other proceeds hereafter due or payable with respect thereof;
 - (g) the Seller IT Assets set forth on Schedule 2.1(g), in each case, to the extent transferable (the "Transferred Business Systems");

- (h) all warranties from third parties applicable to the Facility, the Facility Personal Property or any Inventory, to the extent transferrable in accordance with their respective terms;
- (i) to the extent transferable under applicable Law, all books, records, files and papers containing drawings, formulae, test results and technical and quality records exclusively related to the Purchased Assets or the Facility Operations (other than the Specifically Excluded Records), in each case, whether in hard copy or electronic format, including production reports, service and warranty records, equipment logs, operating guides and manuals, studies, audit reports, quality system documents of the Facility, evaluations, assessment studies or tests, correspondence and other similar documents (but excluding (i) records or files not reasonably separable from documents or databases that do not relate exclusively to the Purchased Assets or the Facility Operations or the transfer of which is not permitted under applicable Laws without the necessary consent and (ii) records and files related to Shared Services) (collectively, the "Facility Books, Documents and Records"); provided, however, that Seller and its Affiliates shall have the right to inspect all Facility Books, Documents and Records in accordance with Section 10.4;
- (j) all vendor lists to the extent exclusively related to the Facility Operations (including credit information and historical pricing information to the extent relating to the vendors listed therein), in each case, as of the Closing;
- (k) all credits, prepaid expenses, deferred charges, advance payments, refunds, security deposits, prepaid items and duties that have been prepaid by Seller pursuant to the Assumed Contracts, in each case, as of the Reference Time (collectively, the "Prepaid Expenses");
 - (1) the Accounts Receivable of Seller set forth on Schedule 2.1(1) (the "Purchased Accounts Receivable"); and
- (m) all rights to claims, refunds, rights of recovery or set off and causes of action against third parties to the extent related to the Purchased Assets described in clauses (a) through (l) of this <u>Section 2.1</u> or the Assumed Liabilities, including all rights against manufacturers or suppliers of any of the Facility Personal Property or Inventory.

2.2 Consents.

(a) Notwithstanding anything in this Agreement to the contrary, this Agreement shall not constitute an agreement to assign or transfer any Assumed Contract, Transferred Governmental Authorization or other Purchased Asset that is not assignable or transferable without the approval, consent or waiver of any Person, other than Seller, Purchaser or any of their respective Affiliates, to the extent that such consent shall not have been given prior to the Closing; provided, however, that Seller shall use, both prior to and for a period of 12 months following the Closing, commercially reasonable efforts to obtain, and Purchaser shall use its commercially reasonable efforts to assist and cooperate with Seller in connection therewith, any such required approval, consent or waiver necessary for the assignment and transfer thereof; provided, further, that (i) neither Seller nor any of its Affiliates shall be required to pay money to any third party, commence any Action or offer or grant any accommodation (financial or otherwise) to any third party and (ii) to the extent the foregoing shall require any action by Seller or any of its Affiliates that would, or would continue to, affect the Facility Operations after the Closing, such action shall require the prior written consent of Purchaser (which consent shall not be unreasonably withheld, conditioned or delayed). Upon obtaining any requisite third party approvals, consents or waivers thereto, such Assumed Contracts, Transferred Governmental Authorizations and other Purchased Assets shall be transferred and assigned to Purchaser hereunder.

- (b) With respect to any Assumed Contract, Transferred Governmental Authorization or other Purchased Asset that is not transferred or assigned to Purchaser at the Closing by reason of Section 2.2(a) (each, a "Nonassigned Asset"), after the Closing and until such time as any requisite approval, consent or waiver is obtained, the Parties shall use commercially reasonable efforts to cooperate with each other in any reasonable arrangements compliant with applicable Law designed to provide Purchaser the benefits of use of any such Nonassigned Asset and the burdens, liabilities, costs, expenses and obligations with respect thereto. Notwithstanding anything in this Agreement to the contrary, to the extent that any Assumed Liability relates to any Nonassigned Asset, such Assumed Liability shall be deemed to be an Excluded Liability unless such Nonassigned Asset is transferred and assigned to Purchaser, or unless (but solely to the extent that) Purchaser obtains the benefit of such Nonassigned Asset under this Section 2.2(b).
- (c) Purchaser acknowledges that the approvals, consents or waivers in connection with the Transactions set forth on <u>Section 5.3(a)</u> of the Disclosure Letter may be required from parties to the Assumed Contracts, Transferred Governmental Authorizations and other rights and that such approvals, consents or waivers have not been and may not be obtained; <u>provided</u>, nothing in the foregoing sentence shall affect or limit the Parties' respective obligations pursuant to <u>Section 7.3</u> hereof.
- **2.3 Excluded Assets**. Purchaser acknowledges and agrees that it is not acquiring, pursuant to this Agreement, any of the Ancillary Agreements or any of the Transactions, any rights, title or interest in, to or under any asset that is not expressly included as a Purchased Asset (all such assets that are not expressly included as Purchased Assets, collectively, the "Excluded Assets"). Specifically, notwithstanding anything in Section 2.1 to the contrary, neither Seller nor any of its Affiliates is contributing, and Purchaser is not acquiring, the following Excluded Assets:
 - (a) all Cash and Investments of Seller;
 - (b) all Accounts Receivable of Seller other than the Purchased Accounts Receivable;
 - (c) the Excluded Inventory;
 - (d) all Contracts of Seller set forth on Schedule 2.3(d) (the "Excluded Contracts");
 - (e) all Contracts between or among Seller and any of its Affiliates or between or among Affiliates of Seller, whether arising before, on or after the Closing Date (including any contracts, agreements or commitments related to Shared Services) (each, an "Intercompany Contract" and collectively, the "Intercompany Contracts");

- (f) other than any Assumed Contract, all Contracts under which a Person other than Seller or any of its Affiliates provides assets, services, rights or benefits with respect to the Purchased Assets or the Facility Operations and one or more other operations or businesses of Seller or its Affiliates set forth on Schedule 2.3(f) (each, a "Shared Contract" and collectively, the "Shared Contracts");
- (g) all of Seller's and its Affiliates' rights under any policies of insurance purchased by or on behalf of Seller or its Affiliates, or any benefits, proceeds or premium refunds payable or paid thereunder or with respect thereto, including all rights to insurance recoveries thereunder and to assert claims with respect to any such insurance recoveries, whether arising before, on or after the Closing Date;
- (h) all books, records, files and papers prepared and maintained by Seller or any of its Affiliates other than the Facility Books, Documents and Records, in each case, whether in hard copy or electronic format (the "Excluded Records"), and further excluding (clauses (i) through (vii), the "Specifically Excluded Records"): (i) all documentation, samples, and product and packaging specifications related to reagents, compounds and products, including batch records or any proprietary information related to such reagents, compounds and products, (ii) all personnel, payroll and employment records, (iii) any books, records, files and papers relating to Taxes (including Tax Returns) of Seller or any of its Affiliates, (iv) standard operating procedures, including any description of the steps used to manufacture reagents, compounds and products, (v) financial information, (vi) books, records, files and papers prepared in connection with the Transactions, including bids, letters of intent and expressions of interest received from other Persons and analyses relating to the Purchased Assets or the Facility Operations and (vii) any attorney work product, attorney-client communications and other items protected by attorney-client privilege;
- (i) (i) other than the Transferred Governmental Authorizations, all Governmental Authorizations of Seller or any of its Affiliates and (ii) all Transferred Governmental Authorizations that are not transferable without the consent of the applicable Governmental Authority and with respect to which such required consent is not obtained;
- (j) all (i) Tax attributes of Seller and (ii) rights to receive refunds or credits with respect to any and all Taxes that constitute Excluded Liabilities;
- (k) any rights, claims and credits of Seller or any of its Affiliates relating to any Excluded Asset or any Excluded Liability, including any such items arising under insurance policies and all guarantees, warranties, indemnities and similar rights in favor of Seller or any of its Affiliates relating to any Excluded Asset or any Excluded Liability;
- (l) all Intellectual Property not expressly included in the Purchased Assets, including the Intellectual Property set forth on <u>Schedule 2.3(l)</u>;
- (m) all IT Assets and Software not included in the Transferred Business Systems or not included in the Assumed Contracts (the "Excluded Business Systems"); and

- (n) any other assets, properties or rights not expressly included in the Purchased Assets.
- **2.4 Assumption of Certain Obligations**. Upon the terms and subject to the conditions set forth herein, Purchaser shall assume, effective as of the Closing, all of the following Liabilities of Seller (or any of its Affiliates), in each case other than the Excluded Liabilities (all such Liabilities so assumed, collectively, the "<u>Assumed Liabilities</u>"):
 - (a) other than as set forth in this <u>Section 2.4</u>, all Liabilities arising out of, or relating to, any Purchased Asset or the Facility Operations from and after the Closing;
 - (b) the Accounts Payable of Seller set forth on Schedule 2.4(b) (the "Assumed Accounts Payable");
 - (c) all Liabilities arising out of, or relating to, compliance with the Recorded Environmental Covenant after the Closing Date; provided, that, after the Closing Date, Seller shall deliver the Recorded Environmental Covenant Notice to the Alabama Department of Environmental Management in accordance with the terms thereof;
 - (d) all Indebtedness of Seller arising out of, or relating to, the Purchased Assets or the Facility Operations set forth on <u>Schedule 2.4(d)</u> (the "<u>Assumed Indebtedness</u>");
 - (e) except as set forth in Section 7.4, all Liabilities arising out of or related to the employment or termination of service of any Transferred Employee by the Purchaser, or the engagement or termination of any Facility Contractor by the Purchaser, in each case, after the Closing Date; provided that if Purchaser fails to comply with Section 7.4(a) with respect to a Facility Employee, or fails to assume the Contracts for any Facility Contractor in accordance herewith, then the Liabilities (if any) arising out of, or related to, the termination of employment of such Facility Employee or termination of such Facility Contractor by Seller or one of its Affiliates shall also be Assumed Liabilities, but not including wage payment, severance (to the extent any such Liabilities arise out of any Facility Employee or Facility Contractor who does not accept an offer of employment from Purchaser or its Affiliates) and PTO obligations; and
 - (f) all Liabilities for (i) Taxes arising from the Purchased Assets or Facility Operations for any Post-Closing Tax Period (as determined for any Straddle Period in accordance with Section 7.12(d)), (ii) Transfer Taxes in accordance with Section 2.9 and (iii) Taxes included in Indebtedness.
- **2.5 Excluded Liabilities**. Notwithstanding anything in <u>Section 2.4</u> to the contrary, Seller shall retain and be responsible for the following Liabilities relating to the Facility Operations (all such Liabilities, collectively, the "<u>Excluded Liabilities</u>"):
 - (a) all Liabilities to the extent arising out of, or relating to, any Purchased Asset or the Facility Operations prior to the Closing (including, for the avoidance of doubt, any Liabilities to the extent arising out of, or relating to, the Settlement Agreement or the claims or Actions described therein);

- (b) all Accounts Payable of Seller other than the Assumed Accounts Payable;
- (c) all Seller Transaction Expenses;
- (d) all Intracompany Payables;
- (e) any Indebtedness of Seller or any of its Affiliates that does not arise out of, or relate to, any Purchased Assets or the Facility Operations;
 - (f) all Liabilities to the extent arising out of, or relating to, the Excluded Assets;
 - (g) all Liabilities under the Benefit Plans (including any Contracts related thereto) and all assets held with respect to the Benefit Plans;
- (h) all Liabilities for (i) Taxes arising from the Purchased Assets or Facility Operations for any Pre-Closing Tax Period (as determined for any Straddle Period in accordance with Section 7.12(d)), except to the extent such Taxes were included in Indebtedness, (ii) Taxes relating to the Excluded Assets and Excluded Liabilities for any taxable period and (iii) Transfer Taxes in accordance with Section 2.9;
- (i) any labor or employment-related Liabilities related to or arising from any period of time up to and including the Closing Date (specifically including all Liabilities under the WARN Act and any state law equivalent), including any Liabilities arising out of or related to the termination of employment of such Facility Employee or the termination of such Facility Contractor by Seller or one of its Affiliates, and all Liabilities with respect to any Facility Employee or Facility Contractor who does not accept an offer of employment from Purchaser or its Affiliates; and
- (j) any other liabilities or obligations of Seller or any of its Affiliates that are not Assumed Liabilities, other than any Environmental Liabilities arising out of, or relating to, actions occurring or conditions existing at the Facility or with respect to the Purchased Assets or Facility Operations on or prior to the Closing Date (regardless of when discovered or asserted), including any continuing effects of such actions or conditions after the Closing Date; provided, that, after the Closing Date, Seller shall deliver the Recorded Environmental Covenant Notice to the Alabama Department of Environmental Management in accordance with the terms thereof.
- **2.6 Purchase Price**. In consideration of the sale and transfer of the Purchased Assets, Purchaser agrees to (a) pay to Seller an aggregate amount equal to (i) the Estimated Purchase Price minus (ii) the Rollover Amount (the "Closing Cash Payment"), plus (iii) the Final Adjustment Amount, as finally determined pursuant to Section 2.7; (b) issue, or cause to be issued, to Seller, the Exchange Shares (collectively, the "Purchase Price"); and (c) assume the Assumed Liabilities. The Closing Cash Payment shall be subject to the adjustment provisions of Section 2.7 and the Purchase Price shall be allocated as described in Section 2.8.

2.7 Purchase Price Adjustment.

- (a) No later than [* * *] Business Days prior to the Closing Date, Seller shall deliver to Purchaser a statement (the "Estimated Closing Statement") setting forth Seller's good faith estimates of (i) the Net Working Capital as of the Reference Time determined in accordance with the Accounting Principles (the "Estimated Net Working Capital Amount"), (ii) the value of the Inventory as of the Reference Time determined in accordance with the Inventory Valuation Principles (the "Estimated Inventory Amount"), (iii) the amount of Assumed Indebtedness as of the Closing (the "Estimated Indebtedness Amount") and (iv) the Estimated Adjustment Amount based on the foregoing clauses (i) through (iii). The Estimated Closing Statement shall be prepared in accordance with the Accounting Principles, the Inventory Valuation Principles and the definitions set forth in this Agreement. Following the delivery of the Estimated Closing Statement, Seller shall reasonably cooperate with Purchaser and its representatives in connection with Purchaser's review of the Estimated Closing Statement, including by providing information reasonably necessary or useful in connection with Purchaser's review of the Estimated Closing Statement as reasonably requested by Purchaser. Seller shall reasonably consider in good faith any revisions to the Estimated Closing Statement delivered by Purchaser to Seller no later than [* * *] Business Days prior to the Closing Date; provided, that if Purchaser and Seller fail to mutually agree upon revisions to the Estimated Closing Statement on or prior to the Business Day prior to the Closing Date, then neither Purchaser nor Seller shall have the right to delay the Closing because of such failure and the amounts set forth in the Estimated Closing Statement, as determined by Seller in good faith, without any adjustment (other than any adjustments agreed to in writing by Seller) shall be the amounts used in the determination of the Estimated Purchase Price.
- (b) No later than [* * *] days after the Closing Date, Purchaser shall deliver to Seller a statement (the "Closing Statement") setting forth Purchaser's determination of (i) the Net Working Capital as of the Reference Time determined in accordance with the Accounting Principles (the "Closing Net Working Capital Amount"), (ii) the value of the Inventory as of the Reference Time determined in accordance with the Inventory Valuation Principles (the "Closing Inventory Amount"), (iii) the amount of Assumed Indebtedness as of the Closing (the "Closing Indebtedness Amount") and (iv) the Adjustment Amount based on the foregoing clauses (i) and (iii). Purchaser and Seller agree that the purpose of the Closing Statement is to measure variations in the components taken into consideration in determining the estimates included in the Estimated Closing Statement compared to the actual values to be included in the Closing Cash Payment, and, without limiting the generality of the foregoing, such process is not intended to permit the introduction of accounting principles, practices, methodologies and procedures that conflict with the Accounting Principles, the Inventory Valuation Principles or the definitions set forth herein.
- (c) From the delivery of the Closing Statement until the determination of the Final Adjustment Amount, Purchaser shall, and shall cause its Affiliates to, upon reasonable prior notice and subject to the execution of customary work paper access letters if requested by accountants of Purchaser: (i) provide Seller and its representatives with reasonable access during normal business hours to the facilities, books and records and work papers of the Facility and Purchaser, as the case may be; and (ii) cooperate with and assist Seller and its representatives in connection with the review of such materials, including by making available its employees, accountants and other personnel to the extent reasonably requested.

- (d) If Seller has any objections to the Closing Statement or any of the amounts set forth therein, Seller shall, no later than [* * *] days following the delivery of the Closing Statement (the "Review Period"), deliver to Purchaser a written notice (an "Objection Notice") setting forth in reasonable detail the nature of each disputed line item calculation and the basis therefor. If Seller delivers an Objection Notice to Purchaser during the Review Period, Purchaser and Seller shall work in good faith resolve Seller's objections within the [* * *]-day period (or such longer period as may be mutually agreed by Purchaser and Seller) following the delivery of the Objection Notice. All such discussions related thereto shall (unless otherwise agreed to by Purchaser and Seller) be governed by Rule 408 of the Federal Rules of Evidence and any applicable similar state Law. If Seller fails to deliver an Objection Notice during the Review Period, the Closing Statement shall be deemed final, conclusive and binding upon Purchaser and Seller, absent fraud or manifest error.
- (e) If Purchaser and Seller are not able to resolve all of Seller's objections set forth in the Objection Notice during the [* * *]-day period (or such longer period as may be mutually agreed by Purchaser and Seller) following delivery of an Objection Notice, the resolution of all such unresolved items (the "Disputed Items") shall be submitted to KPMG LLP (or, if KPMG LLP is unwilling or unable to serve, such other independent accounting firm of nationally recognized standing as Purchaser and Seller shall mutually agree upon) (such firm, the "Independent Accountant"). If Purchaser and Seller are unable to mutually agree on the selection of the Independent Accountant, the Independent Accountant will be selected by lot, after excluding one firm designated by each of Purchaser and Seller. Purchaser and Seller shall execute a customary engagement letter with the Independent Accountant as reasonably requested by the Independent Accountant.
- (f) Purchaser and Seller shall each, promptly (but in any event within [***] Business Days, or such longer time as Purchaser and Seller may mutually agree) following the formal engagement of the Independent Accountant, provide the Independent Accountant (copying the other upon submission) with a single written submission setting forth its respective calculations of and assertions regarding the Disputed Items and upon receipt thereof, each of Purchaser and Seller shall be entitled (no later than [***] Business Days following receipt of the other Party's initial submission) to submit to the Independent Accountant a single written response to such other Party's initial submission setting forth such Party's objections or rebuttals to the calculations or assertions set forth in such initial submission (which responses the Independent Accountant shall promptly distribute to the other applicable Party). There shall be no *ex parte* communications between Purchaser (or its representatives) or Seller (or its representatives), on the one hand, and the Independent Accountant, on the other hand, relating to the Disputed Items, and unless requested by the Independent Accountant in writing, no Party may present any additional information or arguments to the Independent Accountant, either orally or in writing.

- (g) The Independent Accountant shall be instructed to render its determination with respect to the Disputed Items as soon as reasonably possible and the Parties shall use commercially reasonable efforts to cause the Independent Accountant to complete its work within [* * *] days following the formal engagement of the Independent Accountant. The Independent Accountant shall act as an expert and not as an arbitrator to determine solely the Disputed Items based solely on the submissions and responses of Purchaser, on the one hand, and Seller, on the other hand, delivered pursuant to and in accordance with Section 2.7(e), without independent investigation and in accordance with the Accounting Principles, the Inventory Valuation Principles and the definitions set forth herein. In resolving any Disputed Items, the Independent Accountant may not assign a value to any Disputed Item greater than the greatest value for such Disputed Item claimed by either Party or less than the smallest value for such item claimed by either Party. The Parties agree that the determination of the Independent Accountant with respect to any Disputed Items is not intended to permit the introduction of accounting methods, policies, principles, practices, procedures, classifications or estimation methodologies that conflict with the Accounting Principles, the Inventory Valuation Principles or the definitions set forth herein. The determination of the Independent Accountant shall be in writing setting forth its calculation of the Disputed Items, along with its analysis in reasonable detail and the basis and quantification for such final resolution and shall be binding and final for purposes of this Agreement, and enforceable in any court of competent jurisdiction, absent fraud or manifest error. The Independent Accountant shall allocate its costs and expenses between Purchaser and Seller based upon the percentage of the aggregate contested amount submitted to the Independent Accountant that is ultimately awarded to Purchaser, on the one hand, or Seller, on the other hand, such that Purchaser bears a percentage of such costs and expenses equal to the percentage of the contested amount awarded to Seller, and Seller bears a percentage of such costs and expenses equal to the percentage of the contested amount awarded to Purchaser. For illustrative purposes only, if the contested amount submitted to the Independent Accountant is \$1,000,000, and the Independent Accountant determines that Purchaser has a valid claim for \$400,000 of the \$1,000,000, Purchaser shall bear 60% of the fees and expenses of the Independent Accountant and Seller shall bear the remaining 40% of the fees and expenses of the Independent Accountant. As used herein, "Final Adjustment Amount" shall mean (i) if no Objection Notice is delivered by Seller within the Review Period, the Adjustment Amount as shown on the Closing Statement as prepared by Purchaser or (ii) if an Objection Notice is delivered by Seller within the Review Period, either (x) the Adjustment Amount as agreed to in writing by Purchaser and Seller or (y) the Adjustment Amount as determined by the Independent Accountant pursuant to this Section 2.7(e). The scope of the disputes to be resolved by the Independent Accountant shall be limited to the unresolved Disputed Items, and the Independent Accountant is not to make any other determination. Any determinations by the Independent Accountant, and any work or analyses performed by the Independent Accountant, in connection with its resolution of any dispute under this Section 2.7 shall not be admissible in evidence in any Action between the Parties, other than to the extent necessary to enforce payment obligations under this Section 2.7(e).
 - (h) Within [* * *] Business Days following the determination of the Final Adjustment Amount:
 - (i) if the Final Adjustment Amount is less than the Estimated Adjustment Amount, then Seller shall pay to Purchaser, by wire transfer of immediately available funds in accordance with written instructions given to Seller by Purchaser, the amount of such shortfall; and
 - (ii) if the Final Adjustment Amount is greater than the Estimated Adjustment Amount, then Purchaser shall pay to Seller, by wire transfer of immediately available funds in accordance with written instructions given by Seller to Purchaser, the amount of such

2.8 Allocation of the Purchase Price.

- (a) Purchaser and Seller agree that they shall agree upon the allocation of the Purchase Price and the Assumed Liabilities among the Purchased Assets (the "Allocation") following the Closing Date.
 - (b) Purchaser and Seller agree that the Allocation shall be made pursuant to the following procedure:
 - (i) As soon as reasonably practicable after the date of this Agreement (and in any event at least [***] Business Days prior to the Closing Date), Purchaser shall deliver to Seller (A) a proposed value of the Facility for purposes of any applicable real property Transfer Taxes with respect to the transfer of the Facility pursuant to this Agreement (such value, the "Facility Value" and such estimate, the "Estimated Facility Value") and (B) the amount of Transfer Taxes allocable to Seller pursuant to Section 2.9 as a result of the Estimated Facility Value (the "Estimated Seller Transfer Taxes"). Seller shall deliver written notice of any disagreement with respect to the Estimated Facility Value to Purchaser within [***] days after Seller's receipt of the Estimated Facility Value. If Seller does not deliver such a written notice within such period, the Estimated Facility Value (as prepared by Purchaser) shall be final and binding on the Parties for all purposes of this Agreement. If Seller timely delivers such a written notice, Purchaser and Seller shall work in good faith to resolve Seller's objections within the [***]-day period following the delivery of such written notice; provided, that if Purchaser and Seller shall to mutually agree upon the Facility Value on or prior to the Business Day prior to the Closing Date, then neither Purchaser nor Seller shall have the right to delay the Closing because of such failure and the Estimated Facility Value shall be the amount used in the determination of the Estimated Seller Transfer Taxes.
 - (ii) Within [* * *] days of the determination of the Final Adjustment Amount, Seller shall deliver to Purchaser (A) an allocation of the Purchase Price and Assumed Liabilities among the Purchased Assets (the "Proposed Allocation") that is prepared in accordance with Section 1060 of the Code and the Treasury Regulations promulgated thereunder (and any similar provisions of state, local or non-U.S. Law) and the Estimated Facility Value (as determined pursuant to Section 2.8(b)(i)), (B) to the extent that the Facility Value has not been finally determined pursuant to Section 2.8(b)(i), the proposed Facility Value and (C) the amount of Transfer Taxes allocable to Seller pursuant to Section 2.9 as a result of such proposed Facility Value (the "Closing Seller Transfer Taxes"). Purchaser shall deliver written notice of any disagreement with respect to the Proposed Allocation (which notice shall include a reasonably detailed description of the disputed items), the proposed Facility Value and the Closing Seller Transfer Taxes to Seller within [* * *] days after Purchaser's receipt of the Proposed Allocation. If Purchaser does not deliver such a written notice within such period, the Proposed Allocation (as prepared by Seller), the proposed Facility Value and the Closing Seller Transfer Taxes shall be final and binding on the Parties for all purposes of this Agreement. If Purchaser timely delivers such a written notice, Purchaser and Seller shall work in good faith to resolve Purchaser's objections within the [* * *]-day period (or such longer period as may be mutually agreed by Purchaser and Seller) following the delivery of such written notice. If Purchaser and Seller are not able to resolve all of Purchaser's objections set forth in such written notice during the [* * *]-day period (or such longer period as may be mutually agreed by Purchaser and Seller) following delivery thereof, the resolution of all such unresolved items shall be submitted to the Independent Accountant and the terms of Section 2.7(e), Section 2.7(f) and Section 2.7(g) shall apply mutatis mutandis. The Independent Accountant shall be instructed to render its determination with respect to the Disputed Items as soon as reasonably possible and the Parties shall use commercially reasonable efforts to cause the Independent Accountant to complete its work within [* * *] days following the formal engagement of the Independent Accountant; provided that Purchaser shall not dispute, and the Independent Accountant shall not resolve, any item in a manner that is inconsistent with this Section 2.8.

- (c) All fees and expenses relating to the engagement of the Independent Accountant for purposes set forth in Section 2.8(b) shall be borne in the manner described in Section 2.7(g).
- (d) In the event that any Taxing Authority disputes the Allocation, Purchaser or Seller, as the case may be, shall promptly notify the other Party of the nature of such dispute.
- (e) Purchaser and Seller agree to file and prepare any Tax Returns in accordance with the Allocation, and they further agree not to take any tax position that is inconsistent with the Allocation, unless otherwise required pursuant to a "determination" (within the meaning of Section 1313 of the Code or any similar provision of applicable Law).
- 2.9 Transfer Taxes. All Transfer Taxes incurred in connection with this Agreement and the Transactions shall be split equally by Seller and Purchaser. Purchaser and Seller shall cooperate in timely preparing and filing all Tax Returns as may be required to comply with the provisions of such Transfer Tax laws. If Seller or Purchaser determines (in its sole reasonable discretion) that it is required by applicable Laws to pay any Transfer Taxes, then such Party shall pay such Transfer Taxes, and the other Party shall, subject to receipt of reasonably satisfactory evidence of the paying Party's payment thereof, promptly reimburse the paying Party for 50% of Transfer Taxes, whether or not such Transfer Taxes were correctly or legally imposed by the applicable Governmental Authority. The Parties shall cooperate in filing such forms and documents as may be necessary to permit any such Transfer Tax to be assessed and paid on or prior to the Closing Date in accordance with any available pre-sale filing procedure, and to obtain any exemption or refund of any such Transfer Tax. Notwithstanding the foregoing and without duplication, (a) in the event that the Transfer Taxes allocable to Seller pursuant to Section 2.9 as a result of the Facility Value (as finally determined pursuant to Section 2.8(b)(i) or Section 2.8(b)(ii) or Section 2.8(b)(ii), Seller shall pay to Purchaser, by wire transfer of immediately available funds in accordance with written instructions given to Seller pursuant to Section 2.9 as a result of the Facility Value (as finally determined pursuant to Section 2.8(b)(ii) or Section 2.8(b)(ii), within [* * *] Business Days of the final determination of the Facility Value pursuant to Section 2.8(b)(ii) or Section 2.8(b)(iii), Purchaser shall pay to Seller, by wire transfer of immediately available funds in accordance with written instructions given to Seller by Purchaser, the amount of such difference.

2.10 Other Closing Costs. Purchaser shall be responsible for the payment of the following costs and expenses of the Transactions: (a) all fees, costs and expenses incurred in connection with Purchaser's due diligence (including any underwriting, physical inspections and third-party reports), including the cost of title searches and examinations, any surveys and zoning reports, (b) all title insurance costs, (c) all fees, costs and expenses to record any instrument (including the Statutory Warranty Deed), other than title clearing or required release of Liens, (d) one-half of any escrow charges and (e) all other closing costs customarily apportioned to a purchaser in connection with the acquisition of properties located in the jurisdiction in which the Facility is located. Seller shall be responsible for the payment of all Seller Transaction Expenses.

3. CLOSING

3.1 Closing.

(a) Subject to the satisfaction or waiver (where permissible) of the conditions set forth in Article 4, the Closing shall be consummated on the second Business Day following the date on which all of the conditions set forth in Article 4 (excluding those that by their nature are to be satisfied as part of the Closing) have been satisfied or waived (where permissible), remotely via the electronic exchange of documents and signature pages, or at such other time and place as shall be agreed upon by Purchaser and Seller; provided that, except as otherwise waived by Seller, in no event shall the Closing take place prior to [* * *]. The date on which the Closing actually occurs is referred to herein as the "Closing Date".

3.2 Closing Deliveries.

- (a) At the Closing, Seller shall deliver or cause to be delivered to Purchaser all of the following:
 - (i) a duly executed and completed IRS Form W-9 from Seller;
 - (ii) the certificate referred to in Section 4.2(f);
- (iii) an original, fully executed and notarized statutory warranty deed conveying the Facility, subject to the Permitted Encumbrances, in substantially the form of Exhibit K (the "Statutory Warranty Deed");
- (iv) such documents as are reasonably required to evidence Seller's existence and authority to convey the Facility to Purchaser or to consummate the transaction contemplated by this Agreement, including (A) a customary owner's affidavit and gap indemnity in such form as is reasonably necessary to cause the title company to issue an ALTA owner's policy of title insurance insuring fee simple title to the Facility in Purchaser without taking exception for the standard preprinted exceptions or broker liens and any "gap" exception, (B) a termination of any Encumbrances in recordable form or an affidavit from Seller in recordable form that will enable the Title Company to issue the Title Policy without exception for the Amgen Easement or with affirmative insurance in form and substance reasonably acceptable to Purchaser over the Amgen Easement and (C) those affidavits and other documents required by Alabama Code. Section 40-18-86;

- (v) counterparts of each of the Ancillary Agreements (other than the Equity Commitment Letter), each duly executed on behalf of Seller:
- (vi) an Alabama Department of Revenue Real Estate Sales Validation Form (RT-1) and any required transfer tax forms or required forms for withholding (as determined by Purchaser); and
- (vii) a counterpart to the Amended and Restated Limited Liability Company Agreement of Parent, in substantially the form of <u>Exhibit L</u> (the "<u>LLC Agreement</u>"), duly executed by Seller.
- (b) At the Closing, Purchaser shall deliver or cause to be delivered to Seller the Closing Cash Payment, by wire transfer in immediately available funds to an account specified in writing by Seller at least one Business Day prior to the Closing Date, and all of the following:
 - (i) the certificate referred to in Section 4.3(c);
 - (ii) a current sales and use tax resale certificate (or sales tax license) issued by the Alabama Department of Revenue or other applicable state Taxing Authority;
 - (iii) a Subscription Agreement in substantially the form of Exhibit M, pursuant to which the Sponsor will fund cash required to fulfill Buyer's obligations pursuant to this Section 3.2(b) and \$[* * *] in balance sheet cash to be used in connection with the Facility Operations after the Closing, in each case, in exchange for the issuance of Preferred Units of Parent (the "Subscription Agreement"), duly executed by the Sponsor, Parent and the other parties thereto; and
 - $(iv)\ counterparts\ of\ each\ of\ the\ Ancillary\ Agreements,\ each\ duly\ executed\ on\ behalf\ of\ Purchaser.$

4. CONDITIONS TO CLOSING

- **4.1 Conditions to the Obligations of Purchaser and Seller**. The obligation of each of Purchaser and Seller to consummate the Transactions shall be subject to the satisfaction or (to the extent permissible under applicable Law) waiver by the applicable Party, on or prior to the Closing Date of the following conditions precedent:
 - (a) There shall not (i) be in effect any Law or Governmental Order issued by any court of competent jurisdiction or other Governmental Authority with competent jurisdiction that makes illegal or enjoins or prevents the consummation of the Transactions or (ii) have been commenced and be continuing any Action by any court of competent jurisdiction or other Governmental Authority with competent jurisdiction that seeks to make illegal, enjoin or prevent the Transactions.

- **4.2** Conditions to the Obligations of Purchaser. The obligation of Purchaser to consummate the Transactions shall, at the option of Purchaser, be subject to the satisfaction or (to the extent permissible under applicable Law) waiver, on or prior to the Closing Date of the following additional conditions precedent:
 - (a) (i) The Seller Fundamental Representations shall be true and correct (disregarding all qualifications as to materiality and Material Adverse Effect) in all material respects as of the Closing as though made on and as of the Closing (or, in the case of any representation or warranty which specifically relates to an earlier date, such earlier date) and (ii) all other representations and warranties of Seller contained in Article 5 shall be true and correct (disregarding all qualifications as to materiality and Material Adverse Effect) as of the Closing as though made on and as of the Closing (or, in the case of any representation or warranty which specifically relates to an earlier date, such earlier date), except for failures of representations and warranties to be true and correct that do not, individually or in the aggregate, have a Material Adverse Effect;
 - (b) Seller shall have performed in all material respects its agreements and obligations contained in this Agreement required to be performed by it at or before the Closing;
 - (c) Since the date hereof, no Material Adverse Effect shall have occurred;
 - (d) The Rollover Agreement shall be in full force and effect and shall not have been rescinded, terminated, amended, waived or otherwise modified;
 - (e) Purchaser shall have received executed offer letters from [* * *]% of the Facility Employees that were offered employment by Purchaser in accordance with Section 7.4(a); and
 - (f) Seller shall have delivered to Purchaser a certificate, dated as of the Closing Date, signed on behalf of Seller by a duly authorized officer of Seller, confirming the satisfaction of the conditions set forth in <u>Section 4.2(a)</u>, <u>Section 4.2(b)</u> and <u>Section 4.2(c)</u>.
- **4.3** Conditions to the Obligations of Seller. The obligation of Seller to consummate the Transactions shall, at the option of Seller, be subject to the satisfaction or (to the extent permissible under applicable Law) waiver, on or prior to the Closing Date of the following additional conditions precedent:
 - (a) (i) The Purchaser Fundamental Representations shall be true and correct in all material respects as of the Closing as though made on and as of the Closing (or, in the case of any representation or warranty which specifically relates to an earlier date, such earlier date) and (ii) all other representations and warranties of Purchaser contained in Article 6 shall be true and correct (disregarding all qualifications as to materiality and Purchaser Material Adverse Effect) as of the Closing as though made on and as of the Closing (or, in the case of any representation or warranty which specifically relates to an earlier date, such earlier date), except for failures of representations and warranties to be true and correct that do not, individually or in the aggregate, have a Purchaser Material Adverse Effect;

- (b) Purchaser shall have performed in all material respects its agreements and obligations contained in this Agreement required to be performed by it at or before the Closing; and
- (c) Purchaser shall have delivered to Seller a certificate, dated as of the Closing Date, signed on behalf of Purchaser by a duly authorized officer of Purchaser, confirming the satisfaction of the conditions set forth in Section 4.3(a) and Section 4.3(b).

5. Representations and Warranties of Seller

Except as set forth in the Disclosure Letter, Seller represents and warrants to Purchaser as follows:

5.1 Organization. Seller is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware.

5.2 Authority; Binding Effect.

- (a) Seller has all requisite corporate power and authority to (i) own and operate its properties and assets, to carry on its business (including the Facility Operations) as it is being conducted as of immediately prior to the date hereof and (ii) execute and deliver this Agreement and each other Ancillary Agreement and to perform its obligations hereunder and thereunder. The execution and delivery by Seller of this Agreement and the Ancillary Agreements to which it is or will be a party, and the performance by Seller of its obligations hereunder and thereunder, have been duly authorized by all requisite corporate action.
- (b) This Agreement has been duly executed and delivered by Seller and constitutes a valid and binding obligation of Seller, and each other Ancillary Agreement will be, prior to the Closing, duly executed and delivered by Seller and will, after the Closing, constitute a valid and binding obligation of Seller, in each case enforceable against Seller in accordance with its terms, except as enforcement may be limited by bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium or similar laws affecting creditors' rights generally or by general principles of equity (regardless of whether enforcement is sought in a proceeding in equity or law) (the "Enforceability Exceptions").

5.3 Non-Contravention.

(a) The execution and delivery of this Agreement by Seller does not, the execution and delivery of the Ancillary Agreements at the Closing will not, and the consummation of the Transactions and compliance with the terms and conditions hereof and thereof will not, conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation or loss of a material benefit under, or result in the creation of any Liens (other than Permitted Encumbrances) upon any of the Purchased Assets under, any provision of (i) the certificate of incorporation or bylaws of Seller, (ii) any Assumed Contract, (iii) any Transferred Governmental Authorization, or (iv) any judgment, order or decree, or, subject to the matters referred to in Section 5.3(b), Laws applicable to Seller, the Facility Operations or the Purchased Assets, other than, in the case of clauses (ii) and (iii) above, any such items that would not, individually or in the aggregate, reasonably be expected to be material to the Facility Operations or the Purchased Assets.

(b) No material consent, approval, license, permit, order or authorization of, or registration, declaration or filing with, any Governmental Authority is required to be obtained or made by or with respect to Seller in connection with the execution, delivery and performance of this Agreement, the Ancillary Agreements or the consummation of the Transactions other than (i) compliance with and filings under applicable Law in connection with the Specified Governmental Authorizations and (ii) such consents, approvals, licenses, permits, orders, authorizations, registrations, declarations and filings as may be required under Laws applicable to the Facility Operations or the Purchased Assets the absence of which, or the failure to make or obtain which, would not, individually or in the aggregate, reasonably be expected to be material to the Facility Operations or the Purchased Assets.

5.4 Financial Statements; Undisclosed Liabilities. The financial statements included in or incorporated by reference into (including the related notes and schedules) Seller's Form 10-K filed on March 5, 2024, Form 10-K, filed on February 28, 2023, and Form 10-Q filed on August 9, 2024 (the "Seller Financial Statements") have been prepared in conformity with GAAP (except, in the case of any unaudited Seller Financial Statements, to the extent permitted by the U.S. Securities and Exchange Commission) applied on a consistent basis throughout the periods indicated, except as may be noted therein or in the notes thereto. The Seller Financial Statements fairly present in all material respects the financial condition and operating results of Seller, including the Facility Operations, as of the dates, and for the periods, indicated therein. Except as set forth in the Seller Financial Statements, Seller is not subject to any material (as determined based on Seller's consolidated operations) liabilities or obligations, contingent or otherwise with respect to the Facility of the Facility Operations, other than (a) liabilities incurred in the ordinary course of business subsequent to the date of the last report contained in the Seller Financial Statements; (b) obligations under Contracts incurred in the ordinary course of business; or (c) liabilities and obligations of a type or nature not required under GAAP to be reflected in the Seller Financial Statements, in each case, which are not, individually or in the aggregate, material to the Facility, the Facility Operations or the Purchased Assets. The materials attached to Section 5.4 of the Disclosure Letter are true, accurate and complete in all material respects with respect to the information presented therein and for the time periods described thereby and can be reconciled in all material respects to the books and records that were used in the preparation of the Seller Financial Statements for the years ended December 31, 2022 and December 31, 2023 and the three-month p

5.5 Inventory. All Inventory included in the Purchased Assets consists of a quality and quantity usable and salable in the ordinary course of business consistent with past practice, except for obsolete, damaged, defective or slow-moving items that have been written off or written down to fair market value or for which adequate reserves have been established. All such Inventory is owned by Seller free and clear of all Liens (other than any Permitted Encumbrances that do not restrict the transferability of such Inventory), and no Inventory is held on a consignment basis. The quantities of each item of Inventory included in the Purchased Assets (whether raw materials, work-in-process or finished goods) are not excessive but are consistent with the ordinary course of business consistent with past practice of the Facility Operations. All Inventory included in the Purchased Assets has been obtained, manufactured, processed and stored in accordance with any Contracts applicable thereto, including any related Contracts governing the manufacturing, processing and storage of such Inventory, and is of a quality sufficient for the manufacture of GMP compliant products. None of the Inventory included in the Purchased Assets is expired or lacks commercially reasonable remaining shelf-life.

- **5.6 Taxes**. In each case, with respect to the Facility Operations and the Purchased Assets:
- (a) Seller has filed or caused to be filed all Tax Returns required to have been filed (taking into account any extensions) on or before the date hereof;
 - (b) All Taxes (whether or not shown to be due on any Tax Returns) have been timely paid;
 - (c) There are no Liens for Taxes upon any of the Purchased Assets, other than any Taxes not yet due and payable;
- (d) Seller has withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, securityholder or other third party, and all IRS Forms W-2 and 1099 required with respect thereto have been properly completed and timely filed in all material respects;
- (e) Seller has not waived any statute of limitations in respect of Taxes nor has Seller agreed or become subject to any extension of time with respect to a Tax deficiency or assessment;
 - (f) No deficiency for any amount of Tax has been asserted or assessed in writing by any Governmental Authority against Seller;
- (g) There is no Action or any written or, to the Knowledge of Seller, oral, notice of inquiry of any of the foregoing pending against or with respect to Seller or with respect to the Purchased Assets regarding Taxes, and no Action has been threatened in writing or, to the Knowledge of Seller, orally, against or with respect to Seller or with respect to the Purchased Assets regarding Taxes;
- (h) Seller has properly collected and remitted all sales, value added and similar Taxes with respect to sales or leases made or services provided to their customers; and
 - (i) Seller has materially complied with all applicable escheat and unclaimed property Laws.
- **5.7 Absence of Changes**. Since January 1, 2024: (a) there has not been any Material Adverse Effect; and (b) without limiting the foregoing, (i) Seller and its applicable Affiliates have conducted the Facility Operations in the ordinary course of business and past practice and (ii) neither Seller nor any of its Affiliates has, with respect to the Facility, the Purchased Assets or the Assumed Liabilities, taken any action that would, if taken after the date hereof, be prohibited by Section 7.2(a) without the prior written consent of Purchaser.

5.8 No Litigation. No Action relating to the Purchased Assets or the Facility Operations, is pending or, to the Knowledge of Seller, threatened against Seller.

5.9 Compliance with Laws.

- (a) Seller is, and since January 1, 2021, has been, in compliance in all material respects with all Laws applicable to the ownership of the Purchased Assets and conduct of the Facility Operations. Since January 1, 2021, Seller has not received any written or, to the Knowledge of Seller, oral, notice from any Governmental Authority that alleges that the Facility Operations are not in such compliance.
- (b) The conduct of the Facility Operations by Seller is, and since January 1, 2021, has been, in compliance in all material respects with all applicable regulatory Laws, including current Good Manufacturing Practices.
- **5.10 Governmental Authorizations**. Seller possesses or has the benefit of all material Governmental Authorizations necessary to the Facility Operations or that are necessary for the lawful ownership of the Purchased Assets ("Material Permits"). Seller is, and since January 1, 2021, has been, in compliance in all material respects with all such Material Permits. Copies of each Transferred Governmental Authorization have been made available to Purchaser.

5.11 Specified Assumed Contracts.

- (a) <u>Section 5.11(a)</u> of the Disclosure Letter sets forth, as of the date hereof, each of the following Assumed Contracts (each such Assumed Contract, a "<u>Specified Assumed Contract</u>"):
 - (i) any Assumed Contract for the purchase by Seller from a third party of materials, supplies or equipment or the provision to Seller by a third party of services, in each case, which is not terminable by Seller without liability or cost and (a) which involved aggregate payments by Seller to such third party of more than [***] in the year ended December 31, 2023, or (b) which Seller reasonably anticipates will involve aggregate payments by Seller to such third party of more than [***] in the year ended December 31, 2024;
 - (ii) any personal property lease or similar Assumed Contract with any third party in respect of the Facility Operations which has a remaining commitment of one year or more and which Seller reasonably anticipates will involve aggregate payments in excess of \$[* * *] or which encumbers any material Purchased Asset (whether in respect of real property or personal property);
 - (iii) any Assumed Contract involving the sale of any material Purchased Assets, other than Contracts involving Inventory or obsolete or excess equipment sold or disposed of in the ordinary course of business; or

- (iv) any Assumed Contract for any capital expenditures related to the Facility which is not terminable by Seller without liability or cost and which Seller reasonably anticipates will involve aggregate payments in excess of \$[***].
 - (v) any Assumed Contract that relates to any Assumed Indebtedness;
 - (vi) any Assumed Contract involving the granting of any Lien on the Purchased Assets or the Facility;
 - (vii) any Assumed Contract for leased real property;
 - (viii) any Assumed Contract that is a guaranty or which otherwise supports the business Liabilities of a third party;
- (ix) any joint venture Assumed Contract, strategic alliance, revenue or profit sharing arrangement, partnership entity agreement or limited liability company agreement;
- (x) any Assumed Contract containing covenants limiting in any material respect the freedom of Seller or any of its Affiliates to compete with any Person in a line of business, to solicit business from or perform services for any Person, or to operate in any geographic area;
- (xi) any Assumed Contract containing a "most favored nation", exclusivity, minimum purchase, or similar pricing provision or preferential right, or requirement for Seller to purchase all or substantially all of its requirements of a particular product or service;
- (xii) any Assumed Contract relating to the ownership of or investment in any Person, including any partnership, joint venture, strategic alliance, funding, profit sharing or similar arrangements, and any documents related thereto;
 - (xiii) any Assumed Contract with any Governmental Authority;
 - (xiv) any Assumed Contract with any labor union or any collective bargaining agreement;
- (xv) any Assumed Contract involving any severance, change-of-control, bonus, commission, retention or similar type of agreement;
- (xvi) any Assumed Contract involving any confidentiality, noncompetition, nonsolicitation, invention assignment or other restrictive covenant agreement with any Facility Employee;
- (xvii) any Assumed Contract involving the settlement, release, compromise or waiver of any material rights, claims, obligations, duties or liabilities;

- (xviii) any settlement agreement or release agreement with any Facility Employee or Facility Contractor;
- (xix) any Assumed Contract for the employment or other engagement of any Facility Employee or Facility Contractor; and
- (xx) any Assumed Contract pursuant to which Seller licenses Intellectual Property to a third party.
- (b) Each Specified Assumed Contract is valid, binding and in full force and effect and is enforceable by Seller in accordance with its terms, except as enforcement may be limited by the Enforceability Exceptions. Seller is not, and to the Knowledge of Seller, no other counterparty to any Specified Assumed Contract (with or without the lapse of time or the giving of notice, or both) in breach or default in any material respect under any Specified Assumed Contract. Seller has not received any written claim or notice (or to the Knowledge of Seller, any other claim or notice), of a material breach of or material default under any such Specified Assumed Contract, and no event has occurred which, individually or together with other events, would reasonably be expected to result in a breach of or a default under any such Specified Assumed Contract (in each case, with or without notice, lapse of time or both). Seller has not received any written claim or notice (or to the Knowledge of Seller, any other claim or notice), from any party to any Specified Assumed Contract of such party's intention to terminate or modify such Specified Assumed Contract where such termination or modification would be material individually or in the aggregate to the Facility Operations.

5.12 Customers and Suppliers.

- (a) Section 5.12(a) of the Disclosure Letter sets forth a correct list of the top [* * *] customers, measured by revenue, of Seller solely with respect to the Facility Operations for the 12-month period ended December 31, 2023 and September 30, 2024 (collectively, the "Material Customers") (and sets forth opposite the name of such Material Customer the amount of consolidated revenue attributable to such Material Customer).
- (b) Section 5.12(b) of the Disclosure Letter sets forth a correct list of the top [* * *] suppliers, measured by dollar volume of purchases, of Seller solely with respect to the Facility Operations for the 12-month period ended December 31, 2023 and September 30, 2024 (collectively, the "Material Suppliers") (and sets forth opposite the name of such Material Supplier the amount of consolidated purchases attributable to such Material Supplier).
- (c) Seller has not received any indication or written, or to the Knowledge of Seller, oral notice that any Material Customer or Material Supplier has terminated its relationship, plans to terminate its relationship, will cease doing business with Seller or materially reduce or materially alter (in a manner adverse to Seller, including, with respect to any Material Supplier, by materially increasing pricing terms) its relationship with Seller, will or intends to materially decrease the rate or alter the terms of the business it conducts with (or the revenue it generates for) the Facility or is in breach of the terms of any Contract with Seller, and since December 31, 2022 there has been no material change to volume, pricing or other terms with respect to any such Material Customer or Material Supplier.

5.13 Transferred Know-How and Transferred Patents.

- (a) Each of the representations and warranties of Seller related to intellectual property contained in the License Agreement are true and correct.
- (b) Seller is the sole and exclusive owner or licensee of the Transferred Know-How and the Transferred Patents, free and clear of all Liens except for Permitted Encumbrances. The consummation of the Transactions will not conflict with, alter or impair Seller's ability to convey or transfer any rights to the Transferred Know-How. As of the date hereof, no claims are pending or, to the Knowledge of Seller, threatened in writing, against Seller by any person with respect to the ownership, validity, enforceability, effectiveness or use of any Transferred Know-How or any Transferred Patents. To the Knowledge of Seller, the Transferred Patents are valid, subsisting, and enforceable. There are no written claims pending or threatened by Seller against any Person regarding any actual or potential infringement, dilution, misappropriation, violation, or other unauthorized use of the Transferred Know-How or Transferred Patents and, to the Knowledge of Seller, no third party is infringing, misappropriating, or otherwise violating any of the Transferred Know-How or Transferred Patents. All registration, filing, maintenance and renewal fees due as of the date hereof in connection with the Intellectual Property rights contained within the Purchased Assets have been timely paid prior to the due date.
- (c) Except for any services provided in the Mutual Transition Services Agreement, the Transferred Know-How and Transferred Patents, together with the rights of the Purchaser under this Agreement and the License Agreement, are sufficient in all material respects to conduct the Facility Operations in substantially the same manner as conducted during the 12 months preceding the date hereof. This Section 5.13(c) does not include any representation or warranty with respect to the composition of matter, or the utility of, any PEG Products or any drug substances or drug products that are manufactured or conjugated using PEG Reagent.
- (d) Each Person who is or was an employee or contractor of Seller, and who made contributions to the creation or development of any Intellectual Property by or for Seller that is contained in the Purchased Assets, has, as of the Closing Date, executed and delivered to Seller an enforceable agreement, executed on (or consistent in all material respects with) Seller's standard form of invention assignment and nondisclosure agreement, containing an assignment of all such Intellectual Property rights to Seller, and confidentiality provisions protecting the Purchased Assets, except as would not, individually or in the aggregate, have a Material Adverse Effect. To the Knowledge of Seller, no employee or former employer of Seller or any of its Affiliates has any claim, right, or interest to or in any Intellectual Property by or for Seller that is contained in the Purchased Assets.
- (e) No funding, facilities, resources or personnel of any Governmental Authority or any research or educational institution were used to develop or create any Transferred Know-How or Transferred Patents.

- (f) No trade secret constituting Intellectual Property by or for Seller that is contained in the Purchased Assets has been authorized to be disclosed or has been actually disclosed by Seller or any of its Affiliates to any employee, consultant or independent contractor or any third party, in each case, other than pursuant to a written nondisclosure agreement or other written agreement, in each case, including restrictions on the disclosure and use of such trade secret. To the Knowledge of Seller, no employee, consultant or independent contractor or third party has breached or is in breach of any such nondisclosure agreement or other written agreement.
- (g) Seller has not granted any options, licenses or agreements relating to the Transferred Know-How or Transferred Patents, except nonexclusive licenses, express or implied, to vendors, suppliers or distributors, in the ordinary course of business and under which the grant of rights is solely for such third party to provide the applicable services. Solely with respect to the operation of the Facility, Seller as of the date hereof is not bound by or a party to any material options, licenses or agreements of any kind relating to the Intellectual Property of any other Person, except for agreements, express and implied, relating to Software licensed to Seller in the ordinary course of business.
- (h) To the Knowledge of Seller, the Seller IT Assets exclusively relating to the operation of the equipment located on or at the Facility (the "Business Systems") and (i) that is not otherwise used by Seller or any of its Affiliates (A) in the manufacture of products or provision of services other than with respect to PEG Products, (B) in connection with warehousing for Seller or any of its Affiliates or (C) in the conduct of the Shared Services or (ii) that are not Excluded Business Systems, are in good working condition, do not contain any malicious code or defect that would prevent such Seller IT Assets from functioning in the manner Seller applies them to the Purchased Assets as of the date hereof, and operate and perform as necessary to operate or utilize the Purchased Assets in the manner Seller operates or utilizes the Purchased Assets in the twelve months preceding the date hereof. Seller has taken commercially reasonable steps to implement and maintain an information security program designed to protect the Seller IT Assets from unauthorized access or interruption. With respect to the Facility Operations, since January 1, 2021, there have not been any data security incidents, personal data breaches, ransomware incidents or unauthorized access to, or interruption of, any Seller IT Assets, Personal Information, or Seller data in the custody of Seller or, to the Knowledge of Seller, any service provider acting on behalf of Seller. Neither the execution and delivery by Purchaser of this Agreement and the Ancillary Agreements, nor the performance by Purchaser of its obligations hereunder and thereunder, including any transfer of Personal Information to Seller, will violate in any material respects any applicable Privacy Laws.

5.14 Title.

(a) Seller has, or as of the Closing Date will have, good and valid title to the Purchased Assets, free and clear of all Liens except for Permitted Encumbrances, and no payoff letters with respect to the Purchased Assets will be required to be delivered on or prior to Closing to transfer or assign the Purchased Assets to Purchaser, free and clear of all Liens except for Permitted Encumbrances, and consummate the Transactions. The tangible Purchased Assets are in the exclusive possession and control of Seller and no Person other than Seller is entitled to possession of any portion of the tangible Purchased Assets. All material machinery, equipment and other tangible property included in the Purchased Assets, taken as a whole, is in all material respects in good working order and condition, ordinary wear and tear excepted.

- (b) No pending or, to the Knowledge of Seller, threatened condemnation or eminent domain proceedings exist with respect to the Facility. Since January 1, 2021, Seller has not received any written or, to the Knowledge of Seller, oral notice of any material violation of any applicable Law relating to the Facility (including building and zoning laws, ordinances and regulations).
- **5.15 Assets Used in the Facility Operations**. Except for Shared Contracts, the Shared Services and services to be provided to Purchaser or any of its Affiliates pursuant to any of the Ancillary Agreements, the Purchased Assets constitute all of the material assets used by Seller to conduct the Facility Operations as conducted as of the date hereof.
- **5.16 Environmental Matters**. Seller (solely to the extent related to the Purchased Assets) and the Purchased Assets are, and since January 1, 2021, have been, in compliance in all material respects with all applicable Environmental Laws. Seller (solely to the extent related to the Purchased Assets) and the Purchased Assets hold, and are in compliance in all material respects with, all material Environmental Permits required for the conduct of the Facility Operations as currently conducted. Seller has not received, since January 1, 2021 (or earlier if unresolved), any written notice, claim or information request from any Governmental Authority or any other Person that alleges that Seller (solely to the extent related to the Purchased Assets) or Facility Operations are in violation of, or have any liability under, any Environmental Law. There are no pending or, to the Knowledge of Seller, threatened Actions, orders or decrees by, with or before a Governmental Authority against Seller relating to a violation of Environmental Laws or Environmental Permits or to a Release of Hazardous Material (in each case, solely to the extent related to the Purchased Assets). No Hazardous Material has been Released: (a) as a result of Facility Operations; or (b) at, to, on, under or from the Facility, in the case of each of (a) and (b), in an amount, manner, condition or concentration that has resulted, or would reasonably be expected to result, in material liability under Environmental Laws.

5.17 Benefit Plans.

- (a) Section 5.17(a) of the Disclosure Letter lists each material Benefit Plan. Seller has provided or made available to Purchaser a true and correct copy of each such Benefit Plan (or a summary of the material terms of each such Benefit Plan).
- (b) Except where it would not be expected to result in any Liability to Purchaser or its Affiliates, (i) each Benefit Plan is, and has been, established, maintained and operated in material compliance with applicable Laws and with its terms, and (ii) no Action (other than claims for benefits in the ordinary course of business) is pending or, to the Knowledge of Seller, threatened with respect to any material Benefit Plan.
- (c) Each Benefit Plan that is intended to be "qualified" within the meaning of Section 401(a) of the Code has received a favorable determination letter from the IRS or is entitled rely on an opinion or advisory letter from the IRS, as to its qualification under the Code, and to the Knowledge of Seller, no event or omission has occurred that would cause any Benefit Plan to lose such qualification or require corrective action to the IRS or Employee Plans Compliance Resolution System to maintain such qualification.

- (d) Neither Seller nor any ERISA Affiliate has ever maintained, contributed to, or been required to contribute to (i) any employee benefit plan that is or was subject to Title IV of ERISA, Section 412 of the Code, Section 302 of ERISA or (ii) a Multiemployer Plan and neither the Seller nor any ERISA Affiliate has ever incurred any liability under Title IV of ERISA that has not been paid in full.
- (e) Except where it would not be expected to result in any Liability to Purchaser or its Affiliates, neither Seller nor any of its Affiliates provides or has any obligation to provide health or welfare benefits or any other non-pension benefits to any Facility Employees after their employment is terminated (other than as required by Part 6 of Subtitle B of Title I of ERISA or similar state law).
 - (f) No Benefit Plan is subject to the laws of any jurisdiction outside the United States.
- (g) Except where it would not be expected to result in any Liability to Purchaser or its Affiliates, each Benefit Plan that constitutes in any part a nonqualified deferred compensation plan within the meaning of Section 409A of the Code has been operated and maintained in all material respects in operational and documentary compliance with Section 409A of the Code and applicable guidance thereunder. No payment to be made under any Benefit Plan is, or to the Knowledge of Seller, will be, subject to the penalties of Section 409A(a)(1) of the Code that would result in any Liability to Purchaser or its Affiliates.
- (h) Neither the execution and delivery of this Agreement nor the consummation of the Transactions could (either alone or in conjunction with any other event) (i) result in, or cause the accelerated vesting payment, funding or delivery of, or increase the amount or value of, any payment or benefit to any Facility Employee or Facility Contractor; (ii) result in any "excess parachute payment" as defined in Section 280G of the Code; or (iii) result in a requirement to pay any tax "gross-up" or similar "make-whole" payments to any Facility Employee or Facility Contractor.
- (i) None of the Benefit Plans, nor any Liability of any kind thereunder or with respect thereto, will be required by operation of Law or otherwise (except as expressly provided herein or in any of the Ancillary Agreements) to be transferred to Purchaser and/or its Affiliates as a result of the Transactions.

5.18 Employee and Labor Matters.

(a) Section 5.18(a) of the Disclosure Letter contains a true and complete listing, as of the date hereof, of each Facility Employee, setting forth for each Facility Employee: his or her position or title; whether classified as exempt or nonexempt for wage and hour purposes; whether paid on a salary, hourly or commission basis and the Facility Employee's annual base salary, hourly or other rates of compensation; bonus potential; average scheduled hours per week; date of hire; business location; status (*i.e.*, active or inactive and if inactive, the type of leave and estimated duration); any visa or work permit status and the date of expiration, if applicable; and the total amount of bonus, retention, severance and other amounts to be paid to such Facility Employee at the Closing or otherwise in connection with the transactions contemplated hereby.

- (b) Section 5.18(b) of the Disclosure Letter also contains a complete and accurate list of all of the Facility Contractors, setting forth for each Facility Contractor: his, her or its role or nature of services provided; fee or compensation arrangements; fees and other compensation accrued and/or paid, whichever is greater, to such Facility Contractor thus far during calendar year 2024; date of engagement; the primary location (e.g., country, U.S. state) from which services are performed; and expected end date.
- (c) (i) There is no, and since January 1, 2021, there has not been, any labor strike, picketing of any nature, organizational campaigns, labor dispute, slowdown or any other concerted interference with normal operations, stoppage or lockout pending or threatened against or affecting the Facility Operations; (ii) as of the date hereof, neither Seller nor any of its Affiliates have any duty to bargain with any union or labor organization or other person purporting to act as exclusive bargaining representative ("Union") of any Facility Employee's or Facility Contractors with respect to the wages, hours or other terms and conditions of employment of any Facility Employee or Facility Contractor; (iii) there is no collective bargaining agreement or other Contract with any Union, or work rules or practices agreed to with any Union, binding on Seller, or being negotiated, with respect to the Facility Operations or any Facility Employee or Facility Contractor; and (iv) Seller has not engaged in any unfair labor practice with respect to Facility Employees and Facility Contractors.
- (d) Since January 1, 2021, Seller has been in compliance in all material respects with all applicable Laws respecting labor and employment matters with respect to Facility Employees and Facility Contractors, including fair employment practices, pay equity, restrictive covenants, the classification of Facility Contractor, workplace safety and health, work authorization and immigration, unemployment compensation, workers' compensation, affirmative action, terms and conditions of employment, employee leave and wages and hours, including payment of minimum wages and overtime. Seller is not delinquent in any payments to any Facility Employee or Facility Contractor for any wages, salaries, commissions, bonuses, fees or other direct compensation due with respect to any services performed for it or amounts required to be reimbursed to such Facility Employees or Facility Contractors.
- (e) Since January 1, 2021, Seller has not been involved in any Action, or internal employee investigation, in either case, with respect to employment or labor matters regarding Facility Employees or Facility Contractors (including allegations of employment discrimination, retaliation, noncompliance with wage and hour laws, the misclassification of Facility Contractors, violation of restrictive covenants, sexual harassment, other unlawful harassment or unfair labor practices).
- (f) Seller has not experienced a "plant closing," "business closing," or "mass layoff" or similar group employment loss as defined in the federal Worker Adjustment and Retraining Notification Act (the "WARN Act") or any similar state, local or foreign Law with respect to the Facility Operations. During the 90-day period preceding the date hereof, no Facility Employee or Facility Contractor has suffered an "employment loss" as defined in the WARN Act with respect to Seller.

- (g) In the past 12 months, (i) no Person who was a member of Facility management or a key employee at the Facility has been terminated for any reason; and (ii) no Facility Employee has expressed any plans to terminate his, her or its employment or service arrangement with Seller.
- (h) Seller has made available to Purchaser its form of non-competition, non-solicitation, confidentiality and/or invention assignment agreement with Facility Employees. Section 5.18(h) of the Disclosure Letter identifies each Facility Employee who is not subject to a non-competition, non-solicitation, confidentiality and/or invention assignment agreement with Seller on a similar form.
- (i) All Facility Employees are employed at-will and no Facility Employee is subject to any employment contract with Seller, whether oral or written.
- (j) Since January 1, 2021, no allegations or investigations of sexual harassment, other harassment or unlawful discrimination or retaliation have been made to or involved Seller with respect to any Facility Employee or Facility Contractor, and Seller has not otherwise become aware of any such allegations or investigations. To the Knowledge of Seller, there are no facts that would reasonably be expected to give rise to claim of sexual harassment, other unlawful harassment or unlawful discrimination or retaliation against or involving any Facility Employee or Facility Contractor.

5.19 Real Estate.

- (a) No bankruptcy, insolvency, rearrangement or similar action or proceedings, whether voluntary or involuntary, is pending or threatened in writing or, to the Knowledge of Seller, orally, against Seller with respect to the Facility Operations or the Facility.
- (b) There are no Actions for condemnation, eminent domain or taking with respect to the Facility that are pending, ongoing or threatened in writing or, to the Knowledge of Seller, orally.
- (c) Seller has good, valid and marketable title to the Facility, free and clear of all Liens other than Permitted Encumbrances and has the lawful right to sell the Facility. All improvements related to the Facility are located within boundary lines of the described parcels of land and the Facility has direct access to a public street. No improvements owned by third parties encroach onto the Facility property.
- (d) There are no existing or pending Contracts of sale, leases, options to purchase or rights of first refusal (or the like) granted or entered into by Seller affecting the Facility.
- (e) The Facility is now, and at Closing shall be, free and clear of all tenancies or rights of possession granted or entered into by Seller. The Facility is not subject to covenants, conditions and restrictions which would require approval from any third party for construction, landscaping or signage. The Facility has sufficient parking and loading to comply with applicable zoning requirements, and which meet the current needs of the business as conducted at the Facility.

- (f) Seller is not a foreign person, as that term is defined in Section 1445 of the Internal Revenue Code as amended by the Foreign Investment in Real Property Tax Act of 1980 ("FIRPTA"), and Seller shall provide Purchaser with an affidavit to that effect in compliance with FIRPTA at Closing.
- (g) To the Knowledge of Seller, Seller is not aware of or has notice of any work being done within the period of time which would entitle a third party to a mechanics or materialman's lien, or of any assessment, violation or other notice issued by any federal, state, municipal or public body or authority concerning the Facility.
- (h) The existing water, sewer, gas and electricity lines, storm sewer and other utility systems serving the Facility are adequate to serve the utility needs of the business as currently conducted. All of said utilities are installed and operating, and all installation and connection charges have been paid in full.
- (i) Seller has not received any written or, to the Knowledge of Seller, oral, notice of the institution or threat of any Action to change the existing zoning classification as to all or any portion of the Facility, and there are no zoning violation, change or variance Actions, previously decided, pending or to the Knowledge of Seller, threatened, which would adversely affect Purchaser's continued use of the Facility as currently operated.
- (j) There are no structural or latent defects in any of the buildings or other improvements which are a part of the Facility, including the heating, ventilation, air conditioning, electrical, mechanical and plumbing systems, and the drainage at or servicing the Facility and the Facility and all such improvements are in good working order and adequate in quantity and quality for the normal operation thereof.
- (k) The Facility is not subject to any service contracts which would be binding on the facility or the Purchaser other than the Assumed Contracts and Permitted Encumbrances.
- (l) Seller has made available true, correct and complete copies of all material documents in its possession related to the Facility, including regarding title, survey, zoning, environmental conditions as set forth in Phase I or Phase II environmental site assessment reports prepared within the last five years and the Recorded Environmental Covenant (which are the sole documents required to be provided hereunder for environmental matters), physical condition, condition of all FF&E and any appraisals.
- (m) Neither Seller, nor any record or beneficial owner of Seller, is (i) currently listed on the Specially Designated Nationals List ("SDN List") or any similar list maintained by the Office of Foreign Assets Control ("OFAC") at the United States Department of the Treasury; (ii) owned or controlled, directly or indirectly, by a Person who is listed on the SDN List or any similar list maintained by OFAC; (iii) a Person with whom a citizen of the United States is prohibited from engaging in transactions by any trade embargo, economic sanction, or other prohibition of U.S. Law; or (iv) incorporated in any country subject to U.S. country-based economic sanctions whereby conducting transactions with that Person would be in violation of any applicable Law.

5.20 Data Privacy and Security.

- (a) Seller complies and since January 1, 2021, has complied in all material respects with, all: (i) applicable Privacy Laws; (ii) Seller's published policies and contractual obligations; and (iii) all required industry standards including, to the extent applicable, the Payment Card Industry Data Security Standard and all other applicable requirements of the payment card brands, in each case as related to (A) the privacy of all individuals including all users of any web properties, applications, products and/or services of Seller, all Seller employees and all other individuals about whom Seller collects or processes Personal Information, (B) the collection, use, storage, retention, disclosure, transfer, disposal, or any other processing of any Personal Information collected or used by Seller; and (C) the recording or any interception of any communications (collectively, the "Privacy Requirements").
- (b) Seller displays a privacy policy on each website owned, controlled or operated by Seller to the extent required by Privacy Laws, and each such privacy policy incorporates all disclosures to data subjects required by the Privacy Laws. None of the disclosures made or contained in any such privacy policy has been inaccurate, misleading or deceptive, or in violation of the Privacy Laws.
- (c) Seller regularly conducts vulnerability testing or audits of its systems and products, and uses commercially reasonable efforts to remediate or document exceptions for any material vulnerabilities identified in such tests and audits. Seller uses commercially reasonable efforts to timely install software security patches and other fixes to identified technical information security vulnerabilities.
- (d) In connection with each third-party processing Personal Information on behalf of Seller, Seller has entered into written data processing agreements with terms as required by applicable Privacy Laws.
- (e) Since January 1, 2021, there have not been any Actions against Seller related to any data security incidents, ransomware incidents, or any violations of any Privacy Requirements, and there are no facts or circumstances which would reasonably be expected to serve as the basis for any such allegations or claims. Since January 1, 2021, Seller has not received any written or, to the Knowledge of Seller, oral, correspondence relating to, or notice of any Actions or alleged violations of the Privacy Requirements from any person or Governmental Authority, and there is no such ongoing Action or allegation.
- **5.21 U.S. Foreign Corrupt Practices Act**. Neither Seller nor any of its directors, officers, employees or, to the Knowledge of Seller, agents have, in connection with the Facility Operations or the Purchased Assets, directly or indirectly, made, offered, promised or authorized any payment or gift of any money or anything of value to or for the benefit of any "foreign official" (as such term is defined in the U.S. Foreign Corrupt Practices Act of 1977 (the "FCPA")), foreign political party or official thereof or candidate for foreign political office for the purpose of (a) influencing any official act or decision of such official, party or candidate, (b) inducing such official, party or candidate to use his, her or its influence to affect any act or decision of a foreign governmental authority, or (c) securing any improper advantage, in the case of (a), (b) and (c) above in order to assist Seller in obtaining or retaining business for or with, or directing business to, any Person. Neither Seller nor any of their respective directors, officers, employees or, to the Knowledge of Seller, agents have, in connection with the Facility Operations or the Purchased Assets, made or authorized any bribe, rebate, payoff, influence payment, kickback or other unlawful payment of funds or received or retained any funds in violation of any applicable Law. Neither Seller nor any of its officers, directors or employees are the subject of any allegation, voluntary disclosure, investigation, prosecution or other enforcement action related to the FCPA or any other anti-corruption Law.

5.22 Trade Control Laws.

- (a) Since April 24, 2019, Seller (i) has been in compliance in all material respects with all applicable import, export control, and economic and trade sanctions Laws, including the Export Administration Regulations, the International Traffic in Arms Regulations, and the regulations administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury (the "<u>Trade Laws</u>") and (ii) has obtained, or is otherwise qualified to rely upon, all necessary import and export Governmental Authorizations from, and made any filings with, any Governmental Authority required for (A) the import, export, and reexport of products, services, software and technologies and (B) releases of technologies and software to foreign nationals (the "<u>Trade Approvals</u>"), except as would not, individually or in the aggregate, reasonably be expected to be material to the Facility Operations or the Purchased Assets.
- (b) All items developed, produced, or exported by the Seller in connection with the Facility Operations or the Purchased Assets in the past five years are designated EAR99 under the Export Administration Regulations.
- (c) There are no pending or, to the Knowledge of Seller, threatened claims against the Seller nor any actions, conditions, facts, or circumstances, in connection with the Facility Operations or the Purchased Assets, that would reasonably be expected to give rise to any material future claims with respect to the Trade Laws or Trade Approvals.
- **5.23 Brokers**. Except for UBS Securities LLC, the fees and expenses of which are a Seller Transaction Expense, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Transactions based upon arrangements made by or on behalf of Seller or any of its Affiliates.

6. Representations and Warranties of Purchaser

Purchaser represents and warrants to Seller as follows:

6.1 Organization. Purchaser is a corporation duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation.

6.2 Authority; Binding Effect.

- (a) Purchaser has all requisite corporate power and authority to (i) own and operate its properties and assets, (ii) carry on its business as it is now being conducted and (iii) execute and deliver this Agreement and each of the Ancillary Agreements and to perform its obligations hereunder and thereunder. The execution and delivery by Purchaser of this Agreement and the Ancillary Agreements, and the performance by Purchaser of its obligations hereunder and thereunder, have been duly authorized by all requisite corporate action. Immediately after giving effect to the sale, conveyance, assignment and transfer of the Purchased Assets to Purchaser at the Closing in accordance with Section 2.1 (and assuming (x) the continued use by Purchaser of any Nonassigned Asset following the Closing in accordance with Section 2.2(b) and (y) the receipt by Purchaser of services to be provided to Purchaser or any of its Affiliates pursuant to any of the Ancillary Agreements), effective as of and following the Closing, Purchaser shall be able to conduct the Facility Operations in substantially the same manner as conducted by Seller immediately prior to the Closing.
- (b) This Agreement has been duly executed and delivered by Purchaser and constitutes a valid and binding obligation of Purchaser, and each of the Ancillary Agreements will be, prior to the Closing, duly executed and delivered by Purchaser and will, after the Closing, constitute a valid and binding obligation of Purchaser, in each case enforceable against Purchaser in accordance with its terms, except as enforcement may be limited by the Enforceability Exceptions.

6.3 Non-Contravention.

- (a) The execution and delivery of this Agreement by Purchaser does not, and the execution and delivery by Purchaser of the Ancillary Agreements will not, and the consummation of the Transactions and compliance with the terms hereof and thereof will not, conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancelation or acceleration of any obligation or to loss of a material benefit under, or result in the creation of any Liens upon any of the properties or assets of Purchaser under, any provision of (i) the certificate of incorporation or bylaws (or equivalent organizational documents) of Purchaser, (ii) any agreement, contract or other arrangement to which Purchaser is a party or by which any of its properties or assets are bound or (iii) any judgment, order, or decree, or, subject to the matters referred to in paragraph (b) below, Laws applicable to Purchaser or its properties or assets, other than, in the case of clauses (ii) and (iii) above, any such items that would not, individually or in the aggregate, have a Purchaser Material Adverse Effect.
- (b) No consent, approval, license, permit, order or authorization of, or registration, declaration or filing with, any Governmental Authority is required to be obtained or made by or with respect to Purchaser in connection with the execution, delivery and performance of this Agreement, the Ancillary Agreements or the consummation of the Transactions, other than (i) compliance with and filings under applicable Law in connection with the Specified Governmental Authorizations and (ii) such consents, approvals, licenses, permits, orders, authorizations, registrations, declarations and filings the absence of which, or the failure to make or obtain which, would not, individually or in the aggregate have a Purchaser Material Adverse Effect.
- **6.4 No Litigation**. As of the date hereof, no Action by or before any Governmental Authority is pending or, to the knowledge of Purchaser, threatened in writing against Purchaser, in each case, that would, individually or in the aggregate, have a Purchaser Material Adverse Effect. To the knowledge of Purchaser, there exist no facts or circumstances that would materially impair or materially delay the ability of Purchaser to consummate the Transactions or to perform its obligations under this Agreement and the Ancillary Agreements.
- 6.5 Financial Capability. Purchaser has received the Equity Commitment Letter from the Sponsor, pursuant to which the Sponsor has committed, on the terms and subject to the conditions set forth therein, to provide to Purchaser the Equity Financing. The obligation to fund the full commitment under the Equity Commitment Letter is not subject to any conditions precedent, other than the conditions expressly set forth in the Equity Commitment Letter. The Equity Commitment Letter is in full force and effect as of the date hereof and constitutes the valid and binding obligation of Purchaser and each other Person party thereto, enforceable in accordance with its terms against Purchaser and each other Person party thereto. The Equity Commitment Letter provides, and will continue to provide, that Seller is an express third-party beneficiary of the Equity Commitment Letter and Seller is entitled to enforce, directly or indirectly, the Equity Commitment Letter in accordance with its terms and, in connection therewith, Seller has the right to seek an injunction or other form of specific performance or other equitable relief, to cause Purchaser to cause, or to directly cause, the Sponsor to fund the Equity Financing in accordance with the terms thereof. The Equity Commitment Letter has not been amended or modified as of the date hereof, and the respective commitment contained in the Equity Commitment Letter has not been withdrawn or rescinded in any respect as of the date hereof. No event has occurred, or is reasonably expected to occur, which, with or without notice, lapse of time, or both, would constitute a breach or default under the Equity Commitment Letter. The net cash proceeds of the Equity Financing, when funded in accordance with the Equity Commitment Letter, will be sufficient for the satisfaction of all of Purchaser's payment obligations under this Agreement. As of the date hereof, Purchaser has delivered to Seller a copy of the Equity Commitment Letter.

- **6.6 Guarantee**. Concurrently with the execution and delivery of this Agreement, the Sponsor has delivered to Seller the Limited Guarantee, dated as of the date of this Agreement. The Limited Guarantee is in full force and effect and has not been withdrawn or terminated or otherwise amended, supplemented or modified in any respect. The Limited Guarantee is a legal and valid and binding obligation of the Sponsor, enforceable against the Guarantor (as defined in the Limited Guarantee) in accordance with its terms, subject to the Enforceability Exceptions. No event has occurred which, with or without notice, lapse of time or both, could constitute a default or breach on the part of the Sponsor under the Limited Guarantee.
- **6.7 Solvency**. Immediately after giving effect to the Transactions, Purchaser shall be able to pay its debts as they become due and shall own property which has a fair saleable value greater than the amounts required to pay its debts (including a reasonable estimate of the amount of all contingent liabilities). Immediately after giving effect to the Transactions, Purchaser shall have adequate capital to carry on its businesses, including the Facility Operations. No transfer of property is being made and no obligation is being incurred in connection with the Transactions with the intent to hinder, delay or defraud either present or future creditors of the Facility Operations.
- **6.8 Brokers**. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the Transactions based upon arrangements made by or on behalf of Purchaser or any of its Affiliates.
- **6.9 Board Approval.** The board of directors of Purchaser, or a duly authorized committee thereof, by resolutions duly adopted and not subsequently rescinded or modified in any way, has duly (i) resolved that this Agreement and the Transactions are in the best interests of Purchaser and its shareholders and (ii) approved the execution, delivery and performance of this Agreement and of the Ancillary Agreements by Purchaser. No approval of Purchaser's shareholders is necessary for Purchaser to execute and deliver this Agreement or any Ancillary Agreement or consummate the Transactions or perform any of its obligations hereunder or thereunder.

7. COVENANTS

7.1 Information and Documents; Notifications.

- (a) From time to time prior to the Closing, upon reasonable advance notice and to the extent permitted by applicable Law, Seller shall, and shall cause its Affiliates to, permit Purchaser and its representatives to have reasonable access, during normal business hours, to properties, assets, books, records, agreements, documents, data and files of Seller to the extent relating to the Facility Operations or the Purchased Assets and as may be reasonably requested by Purchaser for the sole purpose of consummating the Transactions; provided, however, that such access shall not unreasonably interfere with Seller's or such Affiliates' operation of their respective businesses, including the Facility Operations; and provided further that Seller may restrict the foregoing access (i) to the extent that, in the reasonable judgment of Seller, such access or provision of information would result in a violation of confidentiality obligations to a third party, (ii) to the extent that disclosure of any such information would result in disclosure of any proprietary information or trade secrets of Seller or any other Person (other than with respect to the Facility Operations), (iii) to the extent that disclosure of any such information would result in the loss or waiver of the attorney-client privilege or (iv) in connection with any Action between Seller and its Affiliates, on the one hand, and Purchaser and its Affiliates, on the other hand. In no event shall Purchaser or its representatives be entitled to conduct any invasive or intrusive sampling or testing of any environmental media, including any air, soil, subsurface strata, sediment, surface water or groundwater or any other materials at, on or under the Facility prior to the Closing.
- (b) All information received by Purchaser and given by or on behalf of Seller or any of its Affiliates in connection with this Agreement and the Transactions will be held by Purchaser and its Affiliates, agents and representatives as "Evaluation Material" as defined in, and pursuant to the terms of, the Confidentiality Agreement.
- (c) Each of Purchaser and Seller shall promptly notify the other Party if it becomes aware of any Action instituted or threatened against such Party to restrain, prohibit or otherwise challenge the legality of any of the Transactions. If a Party fails to notify the other Party under this Section 7.1(c), a failure to comply with this Section 7.1(c) shall not constitute a breach of covenant for purposes of Article 4, Article 7.19 or Article 9.

7.2 Conduct of Facility Operations.

- (a) From and after the date hereof until the earlier of the Closing and the termination of this Agreement pursuant to Article 9 (the "Interim Period"), except (i) as set forth on Schedule 7.2 or as otherwise expressly contemplated by this Agreement, (ii) as Purchaser shall otherwise consent in writing (which consent shall not be unreasonably withheld, conditioned or delayed solely with respect to any matter that is unrelated to the Purchased Assets, Assumed Liabilities, Facility Employees or Facility Contractors or Purchaser's interest therein pursuant to this Agreement) or (iii) as is reasonably required to comply with any applicable Law, Seller shall: (A) operate and carry on the business conducted in connection with the Facility Operations in the ordinary course of business in all material respects as conducted immediately prior to the date hereof; (B) preserve, in all material respects, the Purchased Assets in good operating condition and (C) use its commercially reasonable efforts to maintain, in all material respects, its existing relationships with Facility Employees, Facility Contractors and suppliers with respect to the Facility Operations (except for terminations by Seller or its Affiliates for cause or terminations by the Facility Employees, Facility Contractors or suppliers). During the Interim Period, except (w) as set forth on Schedule 7.2 or as otherwise expressly contemplated by this Agreement, (x) as Purchaser shall otherwise consent in writing (which consent shall not be unreasonably withheld, conditioned or delayed solely with respect to any matter that is unrelated to the Purchased Assets, Assumed Liabilities, Facility Employees or Facility Contractors or Purchaser's interest therein pursuant to this Agreement) or (y) as is reasonably required to comply with any applicable Law, Seller shall not, in each case with respect to the business conducted in connection with the Facility Operations:
 - (i) incur, create or assume any Lien, other than Permitted Encumbrances, with respect to any material Purchased Asset;

- (ii) sell, transfer, lease, sublease, license or sublicense or otherwise dispose of any Purchased Assets, other than any Inventory and obsolete or excess equipment sold or otherwise disposed of in the ordinary course of business consistent with past practice;
- (iii) make any changes in the compensation or benefits payable, whether conditionally or otherwise, to any current or former Facility Employee, other than (A) as required under any Benefit Plan, (B) other than in connection with benefit renewals in the ordinary course of business which do not result in any material increase in cost, expense or Liability or (C) pursuant to contractual commitments or arrangements set forth on Schedule 7.2(a)(iii) and which will not result in any Liability under this Agreement or otherwise to Purchaser or its Affiliates (including any retention or similar arrangements that will be paid solely by Seller or its Affiliates);
- (iv) with respect to any Facility Employee, (A) transfer such Facility Employee to another business unit of Seller, (B) offer such Facility Employee employment by another business unit of Seller or any of its Affiliates after the Closing Date or (C) otherwise attempt to persuade any such employee to terminate his or her relationship with Seller or not to commence employment with Purchaser following the Closing;
- (v) grant any severance, change of control, retention, termination pay or similar compensation or benefits to any current or former Facility Employee or Facility Contractor;
- (vi) grant or promise to grant any equity or equity-based incentive award to any current or former Facility Employee or Facility Contractor, including for purposes hereof equity interests of Seller;
- (vii) hire or engage employee, officers, directors, managers, independent contractors, or other individual service providers to be a Facility Employee or Facility Contractor, or terminate the employment or engagement of any Facility Employee or Facility Contractor (except for any terminations for "cause" upon prior written notice to Purchaser);

- (viii) enter into, adversely modify, amend, terminate or otherwise waive, release or assign any material rights, claims or benefits under any Specified Assumed Contract or would be a Specified Assumed Contract if entered;
 - (ix) enter into or negotiate any collective bargaining agreement or similar Contract;
- (x) violate, withdraw, materially amend, allow to lapse or otherwise take any action that would result in Seller or any of its Affiliates being in default (with or without notice or lapse of time or both) under any material Governmental Authorizations, registrations, qualifications or approvals relating to the Facility, or take any other action or cause any other event that would result in the suspension, modification, revocation or nonrenewal thereof, or giving to any other person any right of termination, amendment or cancellation thereof;
- (xi) settle, or offer or propose to settle, (A) any Action or other proceeding involving the Purchased Assets or the Facility or (B) any Action that relates to the Transactions, in each case, that would impose any ongoing obligations on the Purchased Assets or the Facility Operations;
- (xii) adopt or effect any complete or partial liquidation or authorize or undertake a dissolution, consolidation, restructuring or other reorganization of or file a petition in bankruptcy under any provisions of federal or state bankruptcy Law or consent to the filing of any bankruptcy petition against Seller; or
 - (xiii) agree to take any of the foregoing actions.
- (b) Notwithstanding the foregoing, nothing herein will prevent Seller or any of its Affiliates from taking actions prior to the Closing (including (i) contributions, transfers, assignments and acceptances of any assets and liabilities (other than the Purchased Assets), (ii) the repayment of Indebtedness and the extinguishment of Liens and (iii) the cancellation of any Intercompany Contracts and other Contracts that would not constitute Assumed Contracts if not terminated) in order to facilitate the consummation of the Transactions.
- (c) During the Interim Period, except as otherwise permitted by this Agreement or consented to by Seller, Purchaser shall not interfere with the Purchased Assets or the Facility Operations or contact or otherwise communicate with any Facility Employee or Facility Contractor regarding the Transactions.
- **7.3** Consents. During the Interim Period, Purchaser and Seller shall each use their respective reasonable best efforts in attempting to secure any consents, approvals, authorizations and confirmations of non-objection of any third party (including any Governmental Authority) and taking other actions as is necessary to consummate the Transactions as soon as reasonably practical following the date hereof; provided, however, that such actions shall not include any requirement of any Party or any of its Affiliates to pay any out-of-pocket expenses, commence or participate in any Action, offer or grant any accommodation or undertake any obligation or liability (in each case financial or otherwise) to any third party (including any Governmental Authority); provided, further, that neither Party nor its respective officers, employees or authorized representatives may contact any such third party without the other Party's prior written consent, and such consent shall also provide the other Party a reasonable opportunity to participate in any such contact.

7.4 Employees and Employee Benefits.

- (a) Offers of Employment. Prior to the Closing, Purchaser shall, or shall cause one or more of its Affiliates to, offer employment to all Facility Employees (including all Facility Employees on vacation, leave or other authorized absence) to be effective on the Closing Date. Each such offer of employment contemplated pursuant to the immediately preceding sentence shall (i) be on terms and conditions consistent with applicable Law and this Section 7.4, (ii) require the execution by offered Facility Employees of Purchaser's form of offer letter and restrictive covenant agreement consistent with applicable Law and (iii) be made not later than five Business Days prior to the Closing Date (or with respect to any individual who first becomes a Facility Employee fifteen (15) or fewer Business Days prior to the Closing Date; provided that Seller has promptly informed Purchaser of such individual's hire with Seller, not later than the day immediately prior to the Closing Date) for a position having a title and duties that are comparable to those of the applicable Facility Employee as of immediately prior to the Closing Date. Each Facility Employee who accepts such an offer of employment is referred to herein as a "Transferred Employee."
- (b) <u>Comparability of Compensation and Benefits</u>. For a period of at least 12 months following the Closing Date, Purchaser shall, or shall cause an Affiliate of Purchaser to, provide each Transferred Employee with: (i) a base salary or wage rate that is substantially comparable to that provided to such Transferred Employee by Seller immediately prior to the Closing Date, (ii) short-term target cash incentive compensation opportunities that are comparable to those provided to such Transferred Employee by Seller or its Affiliates immediately prior to the Closing Date and (iii) employee benefits (excluding any equity or equity-based compensation, severance, defined benefit, deferred compensation, retiree health or welfare, long-term incentive, change in control, retention or other similar compensation or benefits) that are comparable to the employee benefits provided to such Transferred Employee immediately prior to the Closing Date.
- (c) <u>Defined Contribution Plans</u>. Purchaser shall use commercially reasonable efforts to, or shall use commercially reasonable efforts to cause its Affiliates to, permit all Transferred Employees to be eligible to participate, as of the Closing Date, in one or more qualified defined contribution plans sponsored by Purchaser or any of its Affiliates ("<u>Purchaser DC Plans</u>"), which shall contain all provisions necessary for the acceptance of direct rollovers of "eligible rollover distributions," as defined in the Code, from accounts of Transferred Employees under Seller's or its applicable Affiliate's defined contribution plans ("<u>Seller DC Plans</u>"). The Purchaser DC Plans shall contain provisions to permit any such direct rollover to include the promissory note or notes representing any plan loans outstanding to Transferred Employees under Seller DC Plans on the date of the direct rollover, and Purchaser and Seller shall cooperate with each other to enable such direct rollovers to occur before any such loans become defaulted.
- (d) <u>Termination of Participation in Benefit Plans</u>. Effective as of the Closing Date, except as otherwise specifically provided by the terms of a Benefit Plan or as required by applicable Law, all Transferred Employees will cease any participation in, and any benefit accrual under, all Benefit Plans.

- (e) <u>Credit under Purchaser Plans</u>. To the extent that service is relevant for purposes of eligibility, entitlement to benefits and vesting (excluding for purposes of defined benefit pension benefit accruals) under any employee benefit plan, program or arrangement established or maintained by Purchaser or any of its Affiliates for the benefit of the Transferred Employees following the Closing Date (the "<u>Purchaser Benefit Plans</u>"), Purchaser shall use commercially reasonable efforts to provide that such plan, program or arrangement shall credit such Transferred Employees for service earned prior to the Closing Date with Seller, any of its Affiliates or any of their respective predecessors in addition to service earned with Purchaser or any of its Affiliates on and after the Closing Date; <u>provided</u>, <u>however</u>, such credit for service shall only be provided to the extent allowable under such Purchaser Benefit Plans.
- (f) <u>Purchaser Welfare Plans</u>. Purchaser shall use commercially reasonable efforts, or shall use commercially reasonable efforts to cause its Affiliates, to permit all Transferred Employees and their eligible dependents to participate, in welfare benefit plans sponsored or designated by Purchaser ("<u>Purchaser Welfare Plans</u>") or any of its Affiliates that provide benefits in accordance with the comparability standard set forth in <u>Section 7.4(b)</u>, and shall use commercially reasonable efforts to waive any waiting periods and actively-at-work requirements and any limitations on eligibility, enrollment and benefits relating to any preexisting medical conditions of Transferred Employees and their eligible dependents, in each case, to the extent allowed by the Purchaser Welfare Plans and in accordance with the terms of such plans. Following the Closing Date, Purchaser shall use commercially reasonable efforts to recognize, or shall use commercially reasonable efforts to cause its Affiliates to recognize, for purposes of annual deductible and out-of-pocket limits under Purchaser Benefit Plans, expenses paid toward deductibles and out-of-pocket limits by Transferred Employees and their respective dependents under analogous Benefit Plans in the calendar year in which the Closing Date occurs
- (g) <u>Vacation and Paid Time Off</u>. With respect to each Transferred Employee, Seller shall, or shall cause its applicable Affiliate to, pay each such Transferred Employee for any accrued but unused vacation or paid time off through the Closing to the extent required by applicable policies or Law, and Purchaser shall not assume such vacation or paid time off.
- (h) Flexible Spending Account Plans. Purchaser shall use commercially reasonable efforts, or shall use commercially reasonable efforts to cause its Affiliates, to sponsor or designate a healthcare flexible spending account plan and a dependent care flexible spending account plan (collectively, "Purchaser FSA Plans") for each Transferred Employee who, in the portion of the calendar year on or prior to the Closing Date, contributed to a healthcare flexible spending account plan or dependent care flexible spending account plan, as applicable, maintained by Seller or its Affiliates (collectively, "Seller FSA Plans"). The beginning balance as of the Closing Date in each Purchaser FSA Plan shall be the unused portion of the balance in the analogous Seller FSA Plan. Purchaser shall assume and be solely responsible for all claims for reimbursement by Transferred Employees following the Closing Date, whether incurred prior to, on or after the Closing Date, that have not been paid in full as of the Closing Date, which claims shall be paid pursuant to and under the terms of the Purchaser FSA Plans.

- (i) <u>COBRA</u>. Following the Closing Date, Purchaser shall, or shall cause an Affiliate to, provide continuation health care coverage to all Transferred Employees and their qualified beneficiaries, for any "qualifying event" that occurs following the Closing Date, in accordance with the continuation health care coverage requirements of Section 4980B of the Code and Title I, Subtitle B, Part 6 of ERISA with respect to claims incurred at any time on or after the Closing Date.
- (j) <u>Workers' Compensation Liabilities</u>. As of the Closing Date, Purchaser shall assume, cause its Affiliate to assume, or reimburse Seller for, all Liabilities relating to compensation and benefits under any state workers' compensation or similar Law in connection with claims of Transferred Employees based upon injuries, illnesses or other incidents arising following the Closing Date.
- (k) Work Authorization. If any Facility Employee requires a work permit, employment pass, visa or other legal or regulatory approval for his, her or their employment with Purchaser or its Affiliates, Purchaser shall, and shall cause its Affiliates to, use commercially reasonable efforts to cause any such permit, pass, visa or other approval to be obtained and in effect prior to the Closing Date. Purchaser agrees to serve as the successor-in-interest with respect to Seller's U.S. immigration-related filings submitted on behalf of applicable Facility Employees, with no material changes in the positions, geographic work locations, and base salaries reflected in the immigrant petitions, nonimmigrant filings, and labor certification applications.
- (1) <u>Inducement</u>. Purchaser acknowledges and agrees that it shall not, and shall cause its Affiliates not to, direct or request any Transferred Employee to breach any contractual non-use, confidentiality or nondisclosure obligation for the benefit of Seller and with respect to Seller's or any of its Affiliates' confidential or proprietary information (except to the extent such confidential or proprietary information relates solely to the Purchased Assets or the Assumed Liabilities).
- (m) No Amendment; No Third Party Beneficiaries. The provisions of this Section 7.4 are for the sole benefit of the Parties, and nothing in this Section 7.4, expressed or implied, is intended or shall be construed to confer upon or give to any Person (including, for the avoidance of doubt, any Transferred Employee, Facility Contractor or other current or former employee or independent contractor of Seller or any of its Affiliates), other than the Parties and their respective permitted successors and assigns, any legal or equitable or other rights or remedies (including any third-party beneficiary rights). Nothing contained herein shall (i) constitute or be deemed to constitute the establishment, adoption, amendment, modification or termination of any Benefit Plan or any other benefit or compensation plan, program, policy, agreement or other arrangement or (ii) obligate Seller, Purchaser or any of their respective Affiliates to (A) maintain any particular Benefit Plan or any other benefit or compensation plan, program, policy, agreement or other arrangement or (B) retain the employment of any particular employee or the services of any particular independent contractor. Except as otherwise provided herein and provided that Purchaser complies with the terms of Section 7.4(a) and Section 7.4(a). Seller shall (i) retain and be solely responsible for all employee Liabilities relating to Facility Employees who do not become Transferred Employees, including Facility Employees who do not accept an offer of employment by Purchaser, and all employee Liabilities relating to Transferred Employees that arise from or relate to any event, circumstance or condition occurring or existing during the time period in which they were employed by Seller or its Affiliates, including any such Liabilities that arise as the result of termination of employment or services with Seller and (ii) comply with all applicable Laws relating to the termination of service of any Facility Employees.

7.5 Bulk Transfer Laws. Purchaser acknowledges that Seller has not taken, and does not intend to take, any action required to comply with any applicable bulk sale or bulk transfer Laws or similar Laws of any jurisdiction. Purchaser hereby waives compliance by Seller with the provisions of any bulk sale or bulk transfer Laws or similar Laws of any jurisdiction in connection with the Transactions; provided, that, notwithstanding the foregoing, any Liability resulting from the failure to comply with any applicable bulk sale or bulk transfer Laws or similar Laws shall be treated as an Excluded Liability.

7.6 Litigation Support; Election to Control Non-Indemnified Claims.

- (a) <u>Litigation Support</u>. Following the Closing, Purchaser and its Affiliates, on the one hand, and Seller and its Affiliates, on the other hand, shall reasonably cooperate with each other in the defense or settlement of any Liabilities involving the Purchased Assets or Facility Operations for which the other Party has responsibility under this Agreement by providing the other Party and such other Party's legal counsel reasonable access to employees, records, documents, data, equipment, facilities, products, parts and other information regarding the Purchased Assets and the Facility Operations as such other Party may reasonably request, to the extent maintained or under the possession or control of the requested Party; <u>provided</u>, <u>however</u>, that such access shall not unreasonably interfere with Purchaser's or its Affiliates', or Seller's or its Affiliates', as the case may be, respective businesses; and <u>provided further</u> that any Party may restrict the foregoing access to the extent that (i) such restriction is required by applicable Law, (ii) such access or provision of information would result in a violation of confidentiality obligations to a third party or (iii) disclosure of any such information would result in the loss or waiver of the attorney-client privilege. The requesting Party shall reimburse the other Party for its reasonable out-of-pocket expenses paid to third parties in performing its obligations under this Section 7.4(m).
- (b) Election to Control Non-Indemnified Claims. From and after the Closing, Purchaser and its Affiliates shall use commercially reasonable efforts to promptly notify Seller of any Action instituted by a third party with respect to the Purchased Assets or the Facility Operations with respect to which Purchaser is not entitled to indemnification under Section 8.1 (a "Non-Indemnified Claim"). With respect to any such Non-Indemnified Claim, if such Non-Indemnified Claim could reasonably be expected to materially adversely affect the business of Seller or its Affiliates and is reasonably expected not to materially adversely affect the business of Purchaser and its Affiliates, Seller shall have the right (i) to participate in the defense of such Non-Indemnified Claim and, at its own expense, to employ counsel of its own choosing for such purpose and (ii) in the event Seller agrees to indemnify Purchaser for all judgments, awards, damages and settlement costs and obligations arising from such Non-Indemnified Claim, to conduct and control, at its expense and through counsel of its choosing, the defense, compromise and settlement of such Non-Indemnified Claim; provided, however, that (A) Seller shall give Purchaser advance notice of any proposed compromise or settlement and shall not, without the prior written consent of Purchaser, consent to or enter into any compromise or settlement that commits Purchaser to take, or to forbear to take, any action or does not provide for a full and complete written release by the applicable third party of Purchaser, (B) Seller may not consent to the entry of any judgment that does not relate solely to monetary damages arising from any such Non-Indemnified Claim (which Seller has agreed to pay in full) without the prior written consent of Purchaser and (C) Purchaser shall have the right to participate in the defense of such Non-Indemnified Claim and, at its own expense, to employ counsel of its own choosing for such purpose. The provisions of Article 8 shall apply to any Third Party Cl

7.7 Trade Notification. Purchaser and Seller shall agree on the method and content of the notifications to vendors, suppliers and other service providers of the sale of the Purchased Assets to Purchaser. Purchaser and Seller agree that such notifications are intended to provide sufficient advance notice of the sale and the plans associated therewith, with the objective of minimizing any disruption of the Facility Operations.

7.8 Further Assurances.

- (a) Subject to Section 2.2, for a period of 12 months after the Closing Date, if either Purchaser or Seller becomes aware that any of the Purchased Assets has not been transferred to Purchaser or that any of the Excluded Assets has been transferred to Purchaser, it shall promptly notify the other, and the Parties shall, as soon as reasonably practicable, ensure that such property is transferred, at Purchaser's expense and with any necessary prior third-party approval, consent or waiver, which Seller agrees it shall use commercially reasonable efforts to obtain, to (i) Purchaser, in the case of any Purchased Asset which was not transferred at the Closing, or (ii) Seller, in the case of any Excluded Asset which was transferred at the Closing, or, at Seller's election, such Excluded Asset may be deleted or destroyed as set forth in Section 7.8(b). For a period of 12 months following the Closing, Purchaser shall execute and deliver, or cause to be executed and delivered, to Seller, at Purchaser's expense, such other instruments of assumption as Seller may reasonably request or as may be otherwise necessary to evidence the assumption by Purchaser of the Purchaser, at Purchaser's expense, such other instruments of assumption as Purchaser may reasonably request or as may be otherwise necessary to evidence the assumption by Purchaser of the Purchased Assets.
- (b) If, following the Closing Date, Purchaser becomes aware of any documents, standard operating procedures, materials or other information in its possession that is not a Purchased Asset, Purchaser shall, and shall cause its Affiliates to, subject to Purchaser's obligation to transfer such Excluded Assets to Seller as set forth in Section 7.8(a), destroy or delete any copies of such documents, standard operating procedures, materials or other information as soon as practicable.
- (c) If, following the Closing Date, Purchaser becomes aware of any instance of any Software in its possession that is owned by or licensed to Seller or any of its Affiliates and that is not licensed to Purchaser or any of its Affiliates, Purchaser shall, and shall cause its Affiliates to, at Purchaser's election (i) immediately cease any use of such Software and delete such instances of Software or (ii) obtain a license to such Software, in each case ((i)-(ii)) as soon as practicable.

- **7.9 Payments To or From Third Parties**. If, after the Closing, either Party shall receive any payments or other funds due to the other Party (other than any payments or other funds expressly addressed in the Mutual Transition Services Agreement) pursuant to the terms of this Agreement, then the Party receiving such funds shall promptly forward such funds to the proper Party. The Parties acknowledge and agree there is no right of offset regarding such payments and a Party may not withhold funds received from third parties for the account of the other Party in the event there is a dispute regarding any other issue under this Agreement or any of the Ancillary Agreements.
- **7.10 Excluded Records**. Purchaser acknowledges and agrees that (a) it is not acquiring any right, title or interest (including any license rights or rights of use) in, to or under any Excluded Records to the extent inadvertently included or embedded in the Facility Books, Documents and Records or other Purchased Assets and (b) it shall not, and shall cause its Affiliates not to, attempt to reconstruct, assemble, recompile, reverse engineer or otherwise access copies of any such Excluded Records.
- **7.11 Insurance**. As of the Closing Date, the coverage under all insurance policies related to the Purchased Assets, the Facility Operations and the Facility Employees shall continue in force only for the benefit of Seller and its Affiliates and not for the benefit of Purchaser. Purchaser agrees to arrange for its own insurance policies (which may include self-insurance) with respect to the Purchased Assets, the Facility Operations and the Facility Employees covering all periods and agrees not to seek, through any means, to benefit from any insurance policies of Seller or any of its Affiliates that may provide coverage for claims relating in any way to the Purchased Assets, the Facility Operations or the Facility Employees prior to the Closing.

7.12 Tax Matters.

- (a) Cooperation. After the Closing, each of Purchaser and Seller shall (and shall cause their respective Affiliates to):
- (i) assist the other Party in preparing any Tax Returns which such other Party is responsible for preparing and filing, and which the preparing Party has a commercially reasonable reason for requesting the other Party's assistance;
- (ii) cooperate fully in preparing for any audits of, or disputes with Taxing Authorities regarding, any Tax Returns that relate to the Purchased Assets or the Facility Operations;
- (iii) make available to the other and to any Taxing Authority as reasonably requested all information, records and documents relating to Taxes imposed with respect to the Purchased Assets or the Facility Operations;
- (iv) furnish the other Party with copies of all correspondence received from any Taxing Authority in connection with any Tax audit or information request with respect to any Tax for which such other Party may be liable; and

- (v) timely sign and deliver such certificates or forms as may be necessary or appropriate to establish an exemption from (or otherwise reduce), or file Tax Returns or other reports with respect to, Transfer Taxes.
- (b) Notwithstanding the foregoing Section 7.12(a), (or any other provision in this Agreement), in no event shall Seller or Purchaser (or any of their respective Affiliates) be required to provide any consolidated, combined, unitary or other similar Tax Return to the other Party.
- (c) Withholding Taxes. Each Party shall be permitted to withhold amounts from all amounts payable hereunder as required under applicable Laws, including Alabama Code Section 40-18-86. However, as soon as reasonably practicable prior to the expected Closing Date, Purchaser shall deliver to Seller a schedule of any expected withholding amounts with written explanations for each. Furthermore, each Party shall reasonably cooperate to reduce the amount of withholding Taxes imposed on amounts payable hereunder, including by executing and filing any forms or certificates reasonably required to claim an available reduced rate of, or exemption from, withholding Taxes. To the extent that amounts payable to a recipient are so withheld, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the recipient.
- (d) <u>Straddle Period Taxes</u>. For purposes of this Agreement, personal and real property Taxes and other Taxes on the Purchased Assets or the Facility Operations that are calculated on an annual or periodic basis for any Straddle Period shall be prorated on a per diem basis between Purchaser and Seller as of the Closing Date. With respect to any other Taxes imposed on the Purchased Assets or the Facility Operations with respect to any Straddle Period, such Taxes shall be allocated on a "closing of the books" basis as two partial periods, one ending at the time of the Closing and the other beginning immediately after the Closing.
- **7.13** Assistance with Financial Reporting. For a period of two (2) annual accounting periods after the Closing, Purchaser shall use its commercially reasonable efforts to provide reasonable assistance to Seller, during normal business hours and upon reasonable advance notice, in connection with any financial reporting obligations of Seller and at Seller's sole cost and expense, in each case, related to the Purchased Assets or the Facility Operations during any period prior to the Closing.
- **7.14 Shared Contracts**. With respect to each of the Shared Contracts set forth on <u>Schedule 2.3(f)</u>, prior to the Closing and for a period of 12 months following the Closing, Seller shall cooperate with Purchaser and use its commercially reasonable efforts to cause the counterparty to each such Shared Contract to enter into an agreement with Purchaser or one or more of its Affiliates, to be effective as of the Closing Date (or as soon as reasonably possible following the Closing Date, if the counterparty to such Shared Contract has not entered into such agreement as of the Closing Date), providing for the assets, services, rights or benefits provided by such counterparty with respect to the Purchased Assets or the Facility Operations pursuant to such Shared Contract; <u>provided</u>, <u>however</u>, nothing in this <u>Section 7.14</u> shall be deemed to require Seller or any of its Affiliates to make any payments to such counterparty or to assign any Shared Contract or portion of such Shared Contract.

7.15 Use of Trademarks.

- (a) Neither Seller nor any of its Affiliates is conveying ownership rights with respect to any Trademarks of Seller or any of its Affiliates (collectively, the "Retained Names and Marks").
- (b) Promptly following the Closing Date (but in any event prior to use with or on behalf of a third party), except as expressly authorized in the Manufacturing and Supply Agreement, Purchaser shall, and shall cause its Affiliates to, (i) cease and discontinue any and all uses of the Retained Names and Marks, (ii) remove, destroy or irrevocably strike over the labeling, stationery, forms, supplies, displays, advertising and promotional materials, manuals, and other materials existing as of the Closing Date that bear any Retained Names and Marks and (iii) remove all Retained Names and Marks from all assets, packaging materials, websites, email and other online materials and from all signage and other displays.
- (c) In the event Purchaser or any of its Affiliates violates any of its obligations under this <u>Section 7.15</u>, Seller and its Affiliates may proceed against it in law or in equity for such damages or other relief as a court may deem appropriate. Purchaser acknowledges that a violation of this <u>Section 7.15</u> may cause Seller and its Affiliates irreparable harm which may not be adequately compensated for by money damages. Purchaser therefore agrees that in the event of any actual or threatened violation of this <u>Section 7.15</u>, Seller and its Affiliates shall be entitled, in addition to other remedies that it or any of them may have, to seek a temporary restraining order or preliminary or final injunctive relief against Purchaser or such Affiliate of Purchaser to prevent any violations of this <u>Section 7.15</u>, without the necessity of posting a bond.
- 7.16 Representation and Warranty Insurance. Prior to the Closing, Purchaser will (a) pay or cause to be paid all premiums, underwriting costs, due diligence fees, brokerage or placement fees, premium Taxes and any other fees, costs or expenses incurred, paid or payable in connection with obtaining the RWI Policy and (b) take all steps and other actions required to bind the coverage under the RWI Policy as of the Closing. Purchaser shall not, nor shall it permit any insured policy under the RWI Policy to, amend or waive the RWI Policy in any manner that would allow the insurer thereunder to subrogate or otherwise make or bring any Action against Seller, any of its Affiliates or any of their respective directors, officers, employees, Affiliates, controlling Persons and advisors arising out of or related to this Agreement or any Ancillary Agreement without the prior written consent of Seller (in its sole discretion), except with respect to claims for Fraud. Purchaser shall take all actions reasonably necessary to cause all Closing and post-Closing deliverables under the RWI Policy to be delivered as promptly as possible at or after the Closing and shall deliver a copy of the final, issued RWI Policy to Seller promptly after issuance.
- **7.17 Exclusive Dealings**. During the Interim Period, Seller shall not, and Seller shall not authorize or permit any of their respective or their Affiliates' respective representatives to, directly or indirectly: (a) discuss, encourage, solicit, initiate, knowingly facilitate or continue inquiries regarding an Acquisition Proposal; (b) enter into discussions or negotiations with, or provide any information to, any Person concerning a possible Acquisition Proposal; or (c) enter into any Contracts regarding an Acquisition Proposal. Seller shall promptly (and in any event within 24 hours) advise Purchaser in writing if Seller or any of its Affiliates, or any of their respective directors, officers, employees or other representatives receives any proposal for an Acquisition Proposal, any request for information with respect to any such Acquisition Proposal or any inquiry with respect to or which could reasonably be expected to result in an Acquisition Proposal (including advising Purchaser of the material terms and conditions of such request, proposal or inquiry, and the identity of the Person making the same).

- **7.18 Title Insurance; Due Diligence**. Purchaser acknowledges and agrees that, on or before the date hereof: (a) Seller previously afforded Purchaser all reasonable opportunity to assess, examine, inspect, investigate, review and study any and all aspects of the Facility and confirm such matters as Purchaser deems appropriate in order to determine that the Facility is acceptable to Purchaser and Purchaser has satisfied itself in respect of all aspects of the Facility and (b) Purchaser has obtained a pro forma Title Policy from the Title Company in form and substance acceptable to Purchaser. Seller shall reasonably cooperate with Purchaser in connection with causing the Title Company to issue the Title Policy in a form approved by Purchaser at Closing.
- **7.19** Non-Competition and Non-Solicitation of Business Contacts. In order to protect the value and goodwill of the Purchased Assets, the Facility and the Facility Operations and as a material inducement to Purchaser to consummate the Transactions and purchase the Purchased Assets and assume the Assumed Liabilities, and in order for Purchaser to have and enjoy the full benefit of the Purchased Assets as so acquired, Seller agrees that for a period of [* * *] after the Closing Date (such [* * *] period plus, and as extended by, any period in which Seller is not in compliance with the restrictions set forth herein, the "Restriction Period"), Seller shall not, directly or indirectly:
- (a) (i) engage in, undertake, or carry on a Competitive Business or (ii) have any financial or other interest in, or in any other manner assist any other Person to engage in, undertake, or carry on a Competitive Business, in each case, anywhere in the United States of America (the "Business Area"); provided, that nothing in this Section 7.19(a) shall prohibit Seller from being a purely passive owner of not more than [* * *]% in the aggregate of the outstanding stock of any class of a Person which is a publicly traded entity that is engaged in a Competitive Business; or
- (b) solicit, encourage, induce, entice, or recruit, or attempt to solicit, encourage, induce, entice, or recruit, any customer, client, investor, vendor, supplier, distributor, licenser, business partner, or other business relation (including any Person who was or has been a customer, client, investor, vendor, supplier, distributor, business partner, or other business relation in connection with the Facility or the Facility Operations at any time during the [* * *] period preceding the Closing Date) (i) to alter, reduce, terminate, or refrain from entering, performing, or continuing its contractual or other business relationship concerning the Facility or the Facility Operations, or (ii) to purchase, license, or enter into any contractual or other business relationship with respect to any products, services, systems, solutions, goods, or technologies that are competitive with any products, services, systems, solutions, goods or technology of the Facility, from any Person other than Purchaser or its Affiliates from and after the Closing.
- (c) For clarity, the foregoing covenants in this Section 7.19 do not restrict or limit (i) any of Seller's activities that relate: (A) generally to the operation of other equipment not located on or at the Facility or the business of Seller not conducted at the Facility, in each case, as of the Closing, (B) to warehousing for Seller or any of its Affiliates, (C) to the conduct of quality testing services for Seller or any of its Affiliates or (D) Seller's rights under Section 2.5 of the Manufacturing and Supply Agreement; or (ii) any acquisition of Seller or substantially all of its assets by any other Person that was involved in a Competitive Business as of the consummation of such acquisition. The Parties agree that the foregoing covenants in this Section 7.19 impose a reasonable restraint on Seller in light of the activities and business of Purchaser on the date of the execution of this Agreement and the current plans of Purchaser. The covenants in this Section 7.19 are severable and separate, and the unenforceability of any specific covenant shall not affect the provisions of any other covenant. Moreover, in the event any court of competent jurisdiction shall determine that the scope, time or territorial restrictions set forth are unreasonable, then it is the intention of the Parties that such restrictions be enforced to the fullest extent which the court deems reasonable, and the Agreement shall thereby be reformed. All of the covenants in this Section 7.19 shall be construed as an agreement independent of any other provision in this Agreement, and the existence of any claim or cause of action of Seller against Purchaser, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by Purchaser of such covenants. It is specifically agreed that the periods set forth in this Section 7.19 during which the agreements and covenants made in this Section 7.19 shall be effective, shall be computed by excluding from such computation any time during which the Person bound by such agreement or covenant is found by a court of competent jurisdiction to have been in violation of any provision of this Section 7.19. From and after the Closing Date, the covenants contained in this Section 7.19 shall not be affected by any breach of any other provision hereof by any Party. Each of the Parties hereby agrees that the covenants set forth in this Section 7.19 are a material and substantial part of the Transactions and are supported by adequate consideration.

8. SURVIVAL AND INDEMNIFICATION

- **8.1 Survival.** The Parties, intending to modify the applicable statute of limitations, acknowledge and agree as follows:
- (a) the representations and warranties of the Parties contained in this Agreement or any certificate or other document delivered pursuant hereto shall terminate effective as of the Closing and shall not survive the Closing for any purposes and thereafter, there shall be no liability, nor shall any claim be made by, any Party in respect thereof, except in the case of Fraud, which shall survive the Closing until the applicable statute of limitations with respect to any such Fraud;
- (b) subject to Section 8.1(c), all covenants and other agreements contained in this Agreement or any certificate or other document delivered pursuant hereto that contemplate performance: (i) at or prior to the Closing, shall terminate effective as of the Closing and shall not survive the Closing and (ii) following the Closing, shall survive the Closing for the period contemplated by its terms. If any such post-Closing covenant or agreement also contains pre-Closing or Closing covenants or agreements, then only the portion (if any) of such covenant or other agreement for the period after the Closing shall survive and any portion of any such covenant or other agreement contemplating performance at or prior to the Closing shall terminate effective as of the Closing and shall not survive the Closing; and
- (c) the covenants and other agreements of Seller contained in $\underline{Section~8.2(c)}$ shall survive the Closing until 11:59 p.m. (Pacific time) on the date that is [* * *] years following the Closing Date; $\underline{provided}$, $\underline{however}$, if, at any time on or prior to such date, Purchaser delivers to Seller a written notice asserting in good faith a claim for recovery under $\underline{Section~8.2(c)}$, then the claim asserted in such notice shall survive such date until such time as such claim is fully and finally resolved.

- **8.2** Indemnification by Seller. Subject to the provisions of this Article 8, from and after the Closing, Seller agrees to defend, indemnify and hold harmless Purchaser and its Affiliates and, if applicable, their respective directors, officers, agents, employees, successors, direct and indirect equity holders, members, agents, representatives, successors and assigns, from and against any and all Losses suffered or incurred by such Persons to the extent arising out of or resulting from:
 - (a) any Excluded Liability;
 - (b) any breach by Seller of any of its covenants or agreements contained in this Agreement or the Bill of Sale that are required to be performed after the Closing; and
 - (c) any Third Party Claim arising out of or in connection with any Environmental Liabilities arising out of, or relating to, actions occurring or conditions existing at the Facility or with respect to the Purchased Assets or Facility Operations prior to the Closing (regardless of when discovered or asserted).
- **8.3 Indemnification by Purchaser**. Subject to the provisions of this <u>Article 7.19</u>, from and after the Closing, Purchaser agrees to defend, indemnify and hold harmless Seller and its Affiliates and, if applicable, their respective directors, officers, agents, employees, successors and assigns, from and against any and all Losses suffered or incurred by such Persons to the extent arising out of or resulting from:
 - (a) any Assumed Liabilities (including, for the avoidance of doubt, any Losses resulting from Purchaser's non-performance or breach of, or default under, an Assumed Contract); and
 - (b) any breach by Purchaser of any of its covenants or agreements contained in this Agreement or the Bill of Sale that are required to be performed after the Closing.

8.4 Claim Procedures.

(a) If any of the Persons to be indemnified under this Article 8 (the "Indemnified Party") has suffered or incurred any Loss, the Indemnified Party shall so notify the party from whom indemnification is sought (the "Indemnifying Party") promptly (but in any event, no more than 30 days following the first date following the Closing Date on which such Indemnified Party has knowledge of the facts, matters or circumstances from which it is reasonably apparent that such an occurrence is likely to have occurred in writing, describing such Loss), the amount or estimated amount thereof, if known or reasonably capable of estimation, and the method of computation of such Loss, all with reasonable particularity and containing a reference to the provisions of this Agreement, the Bill of Sale, or any other instrument or certificate delivered pursuant hereto in respect of which such Loss shall have occurred (a "Claim Notice"); provided, that no failure to give notice of such claim prior to the expiration of such 30-day period and no defect in the information contained in any such Claim Notice shall relieve the Indemnifying Party from any obligation under this Article 8 except to the extent that such failure or defect actually prejudices such Indemnifying Party. The Indemnified Party shall cooperate with and provide to the Indemnifying Party such information under the Indemnified Party's control as the Indemnified Party may reasonably request for the purposes of determining the validity of the allegations made in the Claim Notice. The Indemnifying Party and the Indemnified Party shall use reasonable best efforts to avoid production of confidential information (consistent with applicable Law) to third parties and to cause all communications among employees, counsel and others representing any party to a Third Party Claim to be made so as to preserve any applicable attorney-client or work product privileges.

- (b) In the event the Claim Notice results from any Action asserted or threatened against, the Indemnified Party by a third party (a "Third Party Claim"):
 - (i) The Indemnified Party shall promptly (but in any event, no more than 10 Business Days following the Indemnified Party's receipt of the Third Party Claim and in any event no later than the 20th Business Day preceding the date by which an appearance is required to be made before a court, arbitrator or other tribunal or an answer or similar pleading is required to be filed in an Action) notify (the "Third Party Claim Notice") the Indemnifying Party of such Third Party Claim; provided, that no failure by the Indemnified Party to deliver any such Third Party Claim notice shall relieve the Indemnifying Party of any obligation under this Article 8, except to the extent such failure actually prejudices such Indemnifying Party.
 - (ii) The Indemnifying Party under this Article 8 shall have the right, but not the obligation, exercisable by written notice to the Indemnified Party within 30 days of receipt of a Third Party Claim Notice (or the fifth day preceding the date on which an appearance is required to be made before a court, arbitrator or other tribunal or an answer or other similar pleading is required to be filed in an Action) from the Indemnified Party with respect thereto, to assume conduct and control, at the expense of the Indemnifying Party and through counsel of its choosing that is reasonably acceptable to the Indemnified Party, the Third Party Claim identified in such Third Party Claim Notice. If the Indemnifying Party shall control the defense of any such Third Party Claim, the Indemnifying Party shall indemnify the Indemnified Party for all judgments, awards, damages and settlement costs and obligations arising from such Third Party Claim and shall not consent to the entry of any judgment or enter into any compromise or settlement with respect to such claims without the prior written consent of the Indemnified Party, unless such judgment, settlement, compromise or offer to settle, compromise or cease to defend such Third Party Claim (A) provides for the payment by the Indemnifying Party of money as sole relief for the claimant, (B) results in the full and general release of all Indemnified Parties from all liabilities arising or relating to, or in connection with, the Third Party Claim, and (C) involves no finding or admission of any violation of Law or the rights of any Person and no effect on any other claims that may be made against the Indemnified Party or any of its Affiliates. The Indemnifying Party shall permit the Indemnified Party; provided that the fees and expenses of such counsel shall be borne by the Indemnified Party.
 - (iii) If the Indemnifying Party does not duly and timely make such election, the Indemnified Party shall have the right to assume conduct and control of the Third Party Claim; provided, that the Indemnified Party may not compromise or settle any such Third Party Claim for which it is seeking indemnification hereunder without the consent of the Indemnifying Party. If the Indemnifying Party elects not to control or conduct the defense of a Third Party Claim, the Indemnifying Party nevertheless shall have the right to participate in the defense of any Third Party Claim and, at its own expense, to employ counsel of its own choosing for such purpose.

(c) The Indemnified Party and the Indemnifying Party shall cooperate in order to ensure the proper and adequate investigation and defense of all Third Party Claims, including by providing reasonable access to each other's relevant business records, documents and employees, for purposes of investigation, document production, testimony and otherwise. The Indemnified Party and the Indemnifying Party shall keep each other fully and promptly informed with respect to the status of all Third Party Claims and shall deliver to each other copies of all material written notices and documents (including court papers) received by the other that relate to any Third Party Claims. The Person controlling the defense of a Third Party Claim shall in good faith allow the Indemnifying Party or Indemnified Party, as the case may be, to make comments to the materials filed or submitted in such defense, and shall consider such comments in good faith.

8.5 Certain Limitations.

- (a) Notwithstanding the other provisions of this Article 8, (i) the aggregate amount required to be paid by Seller pursuant to Section 8.2 shall not exceed an amount equal to the Purchase Price, except in case of Fraud, and (ii) the aggregate amount required to be paid by Seller pursuant to Section 8.2(c) shall not exceed an amount equal to \$[***].
- (b) In no event shall any Indemnifying Party be obligated to indemnify any Indemnified Party or any other Person with respect to any Losses to the extent that such Losses are expressly included in the calculation of the Purchase Price (including any adjustment thereto, if any, pursuant to Section 2.7).
 - (c) No Indemnified Party shall be entitled to be compensated more than once for the same Loss.
- (d) In the event that there is any claim arising out of this Agreement that would reasonably be expected to be characterized as arising out of an inaccuracy in or breach of any representations or warranties contained in this Agreement (as compared to a Loss arising from an Excluded Liability or a breach of covenant), Purchaser shall, to the extent commercially reasonable, first assert such claim under the RWI Policy before asserting such claim under <u>Section 8.2</u>.

8.6 Losses Net of Insurance, Etc.

(a) Each Indemnified Party shall use its commercially reasonable efforts, and shall cause its Affiliates to use their respective commercially reasonable efforts, to mitigate any indemnifiable Loss upon becoming aware of any event that would reasonably be expected to, or does, give rise thereto; <u>provided</u> that the foregoing shall not be deemed to limit the ability of such Indemnified Party and its Affiliates to incur reasonable costs and expenses in connection therewith.

- (b) The amount of any Loss for which indemnification is provided under Section 8.1 or 8.3 shall be net of (i) any amounts actually recovered by the Indemnified Party pursuant to any indemnification by or indemnification agreement with any third party and (ii) any insurance proceeds actually received as an offset against such Loss, in each case, net of costs of recovery and increased premium or similar amounts, if any. An Indemnified Party shall use commercially reasonable efforts to make such recoveries. If the amount to be netted hereunder from any payment required under Section 8.1 or 8.3 is determined after payment by the Indemnifying Party of any amount otherwise required to be paid to an Indemnified Party pursuant to this Article 8, the Indemnified Party shall repay to the Indemnifying Party, promptly after such determination, any amount that the Indemnifying Party would not have had to pay pursuant to this Article 8 had such determination been made at the time of such payment.
- 8.7 Sole Remedy/Waiver. Except as set forth in Section 8.8 and for injunctive and provisional relief (including specific performance), the Parties acknowledge and agree that, if the Closing occurs, the remedies provided for in this Agreement shall be the Parties' sole and exclusive remedy with respect to the breaches of representations and warranties and covenants and agreements contained herein. Notwithstanding the foregoing, nothing in this Agreement shall affect the rights and remedies available to the parties thereto under, or in connection with, the Ancillary Agreements (other than the Bill of Sale). In furtherance of the foregoing, each of the Parties hereby waives, to the fullest extent permitted by applicable Law, any and all other rights, claims and causes of action (including rights of contributions, if any) known or unknown, foreseen or unforeseen, which exist or may arise in the future, that it may have against Seller or any of its Affiliates, or Purchaser or any of its Affiliates, as the case may be, arising under or based upon any federal, state, local or foreign Law, including any Environmental Law. Notwithstanding the foregoing, nothing in this Agreement shall affect the rights and remedies available to the parties thereto under, or in connection with, the Ancillary Agreements (other than the Bill of Sale).
- **8.8 Fraud.** Nothing in this Article 8 including, for the avoidance of doubt, Section 8.5(a), shall operate to limit the liability in any respect of any Party to the other Party in the event such Party committed Fraud against Purchaser or any of its Affiliates, on the one hand, or Seller or any of its Affiliates, on the other hand.
- **8.9** No Special Damages. WITH RESPECT TO THE PARTIES' RESPECTIVE OBLIGATIONS PURSUANT TO THIS <u>ARTICLE 8</u> AND WITHOUT LIMITING ANY RIGHTS UNDER THE ANCILLARY AGREEMENTS (OTHER THAN THE BILL OF SALE), IN NO EVENT SHALL ANY PARTY BE LIABLE FOR AND EACH PARTY WAIVES ANY CLAIM FROM THE OTHER TO ANY SPECIAL, EXEMPLARY OR PUNITIVE DAMAGES, WHETHER ARISING UNDER ANY LEGAL OR EQUITABLE THEORY OR ARISING UNDER OR IN CONNECTION WITH THIS AGREEMENT, ALL OF WHICH ARE HEREBY EXCLUDED BY AGREEMENT OF THE PARTIES UNLESS ACTUALLY AWARDED TO A THIRD PARTY BY A COURT OF COMPETENT JURISDICTION IN A FINAL, NON-APPEALABLE DETERMINATION.

9. TERMINATION

- **9.1 Termination**. This Agreement may be terminated at any time prior to the Closing:
 - (a) by written agreement of Purchaser and Seller;
- (b) by Purchaser, by giving written notice of such termination to Seller, if (i) any one or more of the representations or warranties of Seller contained in Article 5 shall prove to have been inaccurate when made or (ii) Seller shall be in material breach of its agreements and obligations contained in this Agreement and, in the case of each of clause (i) and (ii), such inaccuracy or breach (A) would give rise, if continuing on the Closing Date, to the failure of any of the conditions set forth in Section 4.2 and (B) cannot be or has not been cured by the earlier of: (x) the date that is 30 days after written notice thereof is given by Purchaser to Seller and (y) the Termination Date;
- (c) by Seller, by giving written notice of such termination to Purchaser, if (i) any one or more of the representations or warranties of Purchaser contained in Article 6 shall prove to have been inaccurate when made or (ii) Purchaser shall be in material breach of its agreements and obligations contained in this Agreement and, in the case of each of clause (i) and (ii), such inaccuracy or breach (A) would give rise, if continuing on the Closing Date, to the failure of any of the conditions set forth in Section 4.3, and (B) cannot be or has not been cured by the earlier of: (x) the date that is 30 days after written notice thereof is given by Seller to Purchaser and (y) the Termination Date;
- (d) by either Purchaser or Seller, by giving written notice of such termination to the other Party, if the Closing shall not have occurred on or prior to [* * *] (unless otherwise extended by written agreement of Purchaser and Seller prior to such date, the "<u>Termination Date</u>"); <u>provided</u> that neither Party may terminate this Agreement pursuant to this <u>Section 9.1(d)</u> if the failure to consummate the Closing by the Termination Date shall be due to the failure of the Party seeking to terminate this Agreement to have fulfilled any of its obligations under this Agreement; or
- (e) by either Purchaser or Seller, by giving written notice of such termination to the other Party, if any court of competent jurisdiction or other Governmental Authority with competent jurisdiction shall have issued a Law or Governmental Order permanently making illegal, enjoining or prohibiting the consummation of the Transactions and such Law or Governmental Order shall have become final and nonappealable.

9.2 Effect of Termination.

(a) In the event of the termination of this Agreement in accordance with Section 9.1, this Agreement shall thereafter become void and have no effect, and no Party shall have any Liability to the other Party or their respective Affiliates, directors, officers or employees; provided, however, that (i) subject to Section 9.2(d), no such termination shall relieve any Party of any Liability to the other Party resulting from any material breach of this Agreement; and (ii) the obligations of the Parties contained in this Section 9.2 and in Sections 7.1(b) ("Information and Documents; Notifications"), 10.1 ("Notices"), 10.9 ("Public Disclosure"), 10.10 ("Expenses") and 10.12 ("Governing Law; Jurisdiction; No Jury Trial") hereof shall survive the termination of this Agreement.

- (b) In the event of the termination of this Agreement in accordance with <u>Section 9.1</u> and at such time any Party is in material breach of or default under any term or provision hereof, such termination shall be without prejudice to, and shall not affect, any and all rights to damages that any other party may have hereunder or otherwise under applicable Law.
- (c) In the event of the termination of this Agreement in accordance with Section 9.1: (i) Purchaser shall return to Seller all documents and other material received by Purchaser, its Affiliates and their respective representatives from Seller, any of its Affiliates or any of their respective representatives relating to the Transactions, whether so obtained before or after the execution hereof, to Seller and (ii) all Confidential Information received by Purchaser, any of its Affiliates and their respective representatives with respect to Seller, any of its Affiliates and the Purchased Assets shall be treated in accordance with the Confidentiality Agreement, which shall remain in full force and effect notwithstanding the termination of this Agreement.
- (d) Except for the Company's rights under Section 10.14 to require Purchaser to specifically perform its obligations under this Agreement, prior to the valid termination of this Agreement, the sole and exclusive remedy (whether at law, in equity, in contract, in tort or otherwise) of Seller against Purchaser or any of its Affiliates in respect of this Agreement and the transactions contemplated hereby shall be the Seller's right to terminate this Agreement in accordance with Section 9.1(c) and seek monetary damages from Purchaser up to an amount not to exceed \$[* * *] (the "Purchaser Liability Cap") for its material breach of this Agreement (including seeking payment of any such damages from Sponsor pursuant to and subject to the terms of the Limited Guarantee solely to the extent provided therein), and upon payment of such amounts by Purchaser (or by the Sponsor pursuant to the Limited Guarantee) pursuant to a final, non-appealable judgment by a court of competent jurisdiction, Purchaser shall have no further liability or obligation relating to or arising out of this Agreement, any Ancillary Agreement (including the Equity Commitment Letter) or any of the transactions contemplated hereby or thereby. Except with respect to Seller's rights under this Section 9.2(d) to pursue a remedy at law for monetary damages not to exceed the Purchaser Liability Cap for material breach of this Agreement by Purchaser and as expressly set forth in the Limited Guarantee to seek payment of any such damages award from the Sponsor, upon the termination of this Agreement for any reason, none of Seller nor any of its Affiliates shall thereafter be entitled to bring and shall in no event support, facilitate, initiate or participate in (or permit any of the foregoing parties to support, facilitate, initiate or participate in) any Action or encourage or take any action or bring any claim (in each case under any legal theory, whether sounding in law (whether for breach of contract, in tort or otherwise) or in equity), other than opposing the bringing of any such Action against Purchaser, Sponsor or any of their respective Affiliates with respect to, arising out of, relating to or in connection with the failure of the Closing to occur or for a breach or failure to perform hereunder, under the Equity Commitment Letter or otherwise (in any case, whether willfully, intentionally, unintentionally or by willful and/or material breach, or otherwise), and Seller shall cause any such Action pending as of any termination of this Agreement to be dismissed with prejudice as promptly as practicable after such termination. All former, current or future direct or indirect stockholders, equity holders, controlling persons, portfolio companies, directors, officers, employees, general or limited partners, members, managers, trustees, attorneys, agents, representatives or Affiliates of Sponsor and the Purchaser, or any former, current or future direct or indirect stockholder, equity holder, controlling person, portfolio company, director, officer, employee, general or limited partner, member, manager, trustee, attorney, agent, representative or Affiliate of any of the foregoing are express third party beneficiaries of this Section 9.2(d) and may enforce the provisions of this Section 9.2(d) against the Parties to this Agreement. Seller will be entitled to seek specific performance to cause the Closing to occur solely to the extent permitted by Section 10.14 prior to terminating this Agreement in accordance with its terms or seek a monetary damages remedy against Purchaser (and Sponsor solely to the extent provided in the Limited Guarantee) solely to the extent provided in this Section 9.2(d) and Section 10.15, but shall not be entitled to both specific performance to cause the Closing to occur pursuant to Section 10.14 and payment of any such monetary damages (in no event to exceed the Purchaser Liability Cap).

10. MISCELLANEOUS

10.1 Notices. Any notice or other communication required or permitted to be delivered to any Party under this Agreement shall be in writing and shall be deemed properly delivered, given and received: (a) if delivered by hand, when delivered; (b) if sent on a Business Day by email and receipt is confirmed or no failure message is generated, when transmitted; (c) if sent by email on a day other than a Business Day and receipt is confirmed or no failure message is generated, on the Business Day following the date sent; (d) if sent by registered, certified or first class mail, the third Business Day after being sent; and (e) if sent by overnight delivery via a national courier service, one Business Day after being sent, in each case to the address set forth beneath the name of such Party below (or to such other address as such Party shall have specified in a written notice given to the other Parties):

to Seller:

Nektar Therapeutics 455 Mission Bay Boulevard South San Francisco, CA 94158 Attention: Mark Wilson, Chief Legal Officer Email: [* * *]

with required copies (which shall not constitute notice) to:

Sidley Austin LLP 1999 Avenue of the Stars, 17th Floor Los Angeles, CA 90067 Attn: Joshua Hofheimer and Daniel Belke

Email: [* * *]

to Purchaser:

[* * *] c/o Ampersand Capital Partners **Ampersand Capital Partners** One Post Office Square, Suite 2900 Boston, MA 02109 Attention: Dana L. Niles; David Q. Anderson Email: [* * *]

with a copy to:

Goodwin Procter LLP 100 Northern Avenue Boston, MA 02210 Attn: James T. Barrett; Eric Carlson Telephone: [* * *]

Email: [* * *]

10.2 Amendment; Waiver. Any provision of this Agreement may be amended or waived if, and only if, such amendment or waiver is in writing and signed (i) in the case of an amendment, by Purchaser and Seller, and (ii) in the case of a waiver, by the Party against whom the waiver is to be effective. No failure or delay by any Party in exercising any right, power or privilege hereunder shall operate as a waiver thereof, nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege.

10.3 Assignment. No Party may assign any of its rights or obligations under this Agreement, including by sale of stock, operation of Law in connection with a merger or sale of substantially all the assets, without the prior written consent of the other Parties, except that (i) Purchaser may, without such consent, assign its rights to acquire the Purchased Assets hereunder, in whole or in part, to one or more of its Affiliates or to any lender in connection with collateral assignment and (ii) Seller may, without such consent, assign its rights and obligations hereunder, in whole or in part, to one or more of its Affiliates; provided, however, that no such assignment by Purchaser or Seller shall relieve Purchaser or Seller, as applicable, of any of its respective obligations hereunder.

10.4 Access to Records After Closing.

(a) For a period of three years after the Closing Date and to the extent not prohibited or restricted by applicable Law, Seller and its representatives shall have reasonable access to (and rights to make copies of) the Facility Books, Documents and Records, any other books and records relating to the Facility Operations or the Purchased Assets, and any personnel of Purchaser or an Affiliate of Purchaser with knowledge thereof, in each case, to the extent that such access may reasonably be required by Seller in connection with matters relating to or affected by the Facility Operations or ownership or control of the Purchased Assets, in each case prior to the Closing Date, or for any other reasonable purpose of Seller. Such access shall be afforded by Purchaser and its Affiliates upon receipt of reasonable advance notice and during normal business hours. Seller shall be solely responsible for any costs or expenses incurred by it pursuant to this Section 10.4(a). If Purchaser or any Affiliate of Purchaser shall desire or be required by applicable Law to dispose of any of the books and records described in the first sentence of this Section 10.4(a) (including any Facility Books, Documents and Records) prior to the expiration of such three-year period, Purchaser shall use reasonable efforts to, prior to such disposition, give Seller a reasonable opportunity, at Seller's expense, to segregate and remove such books and records as Sellers may select.

(b) For a period of three years after the Closing Date and to the extent not prohibited or restricted by applicable Law, Purchaser and its representatives shall have reasonable access to all of the books and records relating to the Facility Operations or the Purchased Assets which Seller or any Affiliate of any Seller may retain after the Closing Date, and any personnel of Seller or an Affiliate of Seller with knowledge thereof. Such access shall be afforded by Seller and its Affiliates upon receipt of reasonable advance notice and during normal business hours. Purchaser shall be solely responsible for any costs and expenses incurred by it pursuant to this Section 10.4(b). If Seller or any Affiliate of any Seller shall desire or be required by applicable Law to dispose of any of the books and records described in the first sentence of this Section 10.4(b) prior to the expiration of such three-year period, Seller shall use reasonable efforts to, prior to such disposition, give Purchaser a reasonable opportunity, at Purchaser's expense, to segregate and remove such books and records as Purchaser may select.

10.5 Confidential Nature of Information. Each Party agrees that all Confidential Information which it shall have obtained regarding the other Parties during the course of the negotiations leading to the consummation of the Transactions (whether obtained before or after the date hereof), the investigation provided for herein and the preparation of this Agreement, the Ancillary Agreements and other related documents shall be held in confidence pursuant to the Confidentiality Agreement, which Confidentiality Agreement shall terminate at Closing. From and thereafter, Seller shall, and shall take all reasonable measures to cause their respective agents, representatives, Affiliates, employees, officers and directors to, as applicable, treat and hold as confidential all Confidential Information relating to the Facility, the Facility Operations or the Purchased Assets. Seller shall, and shall take all reasonable measures to cause their respective agents, representatives, Affiliates, employees, officers and directors to, refrain from directly or indirectly disclosing, revealing, divulging or otherwise communicating any such Confidential Information to any Person other than authorized Affiliates, officers, managers and employees of Purchaser or using or otherwise exploiting any such Confidential Information for such Person's own benefit or for the benefit of anyone other than Purchaser and its Affiliates. Notwithstanding the foregoing, Confidential Information shall not include: (a) information that, at the time of disclosure, is available publicly and was not disclosed in breach of any obligation (contractual or legal) to keep such information confidential; (b) information obtained after the date of this Agreement, on a non-confidential basis from a third party and who (to the recipient's knowledge after reasonable inquiry) was not bound by any obligation (contractual or legal) of confidentiality to a Party or any of their respective Affiliates with respect to such information; or (c) information shown by contemporaneous record to have been independently developed without use of or reference to other Confidential Information. In the event that a Seller or any of their respective agents, representatives, Affiliates, employees, officers or directors becomes legally compelled to disclose any such Confidential Information, Seller shall provide Purchaser with prompt written notice of such requirement so that Purchaser may seek a protective order or other remedy or waive compliance with the provisions of this Section 10.5. In the event that a protective order or other remedy is not obtained or if Purchaser waives compliance with this Section 10.5, the Party required to disclose such Confidential Information shall (or the applicable Party shall cause its agents, representatives, Affiliates, employees, officers and directors to, as applicable) furnish only that portion of such Confidential Information as is legally required to be provided and exercise its reasonable best efforts to obtain assurances that confidential treatment will be accorded to such Confidential Information.

- 10.6 Entire Agreement. This Agreement, the Disclosure Letter, the exhibits and schedules referred to herein, the documents delivered pursuant hereto, including the Ancillary Agreements, and the Confidentiality Agreement contain the entire understanding of the Parties with regard to the subject matter contained herein or therein, and supersede all other prior representations, warranties, agreements, understandings or letters of intent between or among any of the Parties.
- **10.7 Fulfillment of Obligations**. Any obligation of any Party to any other Party under this Agreement, which obligation is performed, satisfied or fulfilled by an Affiliate of such Party, shall be deemed to have been performed, satisfied or fulfilled by such Party.
- 10.8 Parties in Interest. This Agreement shall inure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns. Nothing in this Agreement, express or implied, is intended to confer upon any Person other than Purchaser, Seller or their respective successors or permitted assigns, any rights or remedies under or by reason of this Agreement.
- 10.9 Public Disclosure. No Party shall originate any publicity, news release or other similar public announcement, written or oral, whether relating to this Agreement or any of the Ancillary Agreements or the existence of any arrangement between the Parties, without the prior written consent of the other Party whether or not named in such publicity, news release or other similar public announcement, except (a) Seller and its Affiliates and Purchaser and its Affiliates shall issue a joint press release in connection with the execution and delivery of this Agreement, (b) either Party may originate any such publicity, news release or other similar public announcement as may be necessary to comply with any applicable Law or the rules and regulations of any stock exchange upon which the securities of such Party or its Affiliates are listed, if any, or (c) as may otherwise be required by any other applicable Law; provided that in such event under clauses (b) and (c), the Party issuing such publicity, news release or other similar public announcement shall still be required to consult with the other Party, whether or not named in such publicity, news release or other similar public announcement, a reasonable time prior to its release to allow the other Party to comment thereon and, after its release, shall provide the other Party with a copy thereof; provided, further, the requirements of this Section 10.9 shall not apply to any disclosure by Seller, Purchaser or any of their respective Affiliates, of any information concerning this Agreement or the Transactions for (w) the bona fide marketing and publicity of the Facility by Purchaser and its Affiliates following the Closing to actual and potential customers and other business relations; provided such communication does not reference the terms of this Agreement or other information subject to an obligation of confidentiality, (x) in connection with any dispute between the Parties or their respective Affiliates; (y) to such Party's (or any of its Affiliates') legal, accounting or financial advisors to the extent reasonably necessary for such adviser to perform its legal, accounting or financial services, respectively, for such Party (or its Affiliates); or (z) to any of such Person's Affiliates and their respective direct and indirect equity holders and limited partners, any bona fide actual or potential licensee, sublicensee or existing and potential investors, lenders and acquirers and the accountants, advisors and other professional representatives of any of the foregoing.

10.10 Expenses. Except as otherwise expressly provided in this Agreement (including, for the avoidance of doubt, <u>Section 10.10</u>), whether or not the Transactions are consummated, all costs and expenses incurred in connection with this Agreement and the Transactions shall be borne by the Party incurring such expenses.

10.11 Interpretation; Disclosure Letter. Articles, titles and headings to sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement. The schedules and exhibits referred to herein shall be construed with and as an integral part of this Agreement to the same extent as if they were set forth verbatim herein. Any fact or item disclosed in any section of the Disclosure Letter shall be deemed to have been disclosed with respect to every other section in this Agreement to which such disclosure is reasonably apparent on its face. Neither the specification of any dollar amount in any representation or warranty contained in this Agreement nor the inclusion of any specific item in any schedule hereto is intended to imply that such amount, or higher or lower amounts, or the item so included or other items, are or are not material, and no Party shall use the fact of the setting forth of any such amount or the inclusion of any such item in any dispute or controversy between the Parties as to whether any obligation, item or matter not described herein or included in any schedule is or is not material for purposes of this Agreement. Unless this Agreement specifically provides otherwise, neither the specification of any item or matter in any representation or warranty contained in this Agreement nor the inclusion of any specific item in any section of the Disclosure Letter is intended to imply that such item or matter, or other items or matters, are or are not in the ordinary course of business, and no Party shall use the fact of the setting forth or the inclusion of any such item or matter in any dispute or controversy between the Parties as to whether any obligation, item or matter not described herein or included in any section of the Disclosure Letter is or is not in the ordinary course of business for purposes of this Agreement.

10.12 Governing Law; Jurisdiction; No Jury Trial.

- (a) This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to principles of conflicts of laws.
- (b) With respect to any Action relating to this Agreement, each Party expressly and irrevocably (i) expressly and irrevocably consents and submits to the exclusive jurisdiction of each state and federal court located in the County of New Castle, State of Delaware (and each appellate court located in the County of New Castle, State of Delaware) in connection with any such Action and (ii) waives any objection which it may have at any time to the laying of venue of any Action brought in any such court, waives any claim that such Action has been brought in an inconvenient forum and further waives the right to object, with respect to such Action, that such court does not have any jurisdiction over such Party.
- (c) THE PARTIES HEREBY IRREVOCABLY WAIVE, AND AGREE TO CAUSE THEIR RESPECTIVE SUBSIDIARIES TO WAIVE, THE RIGHT TO TRIAL BY JURY IN ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT, THE ANCILLARY AGREEMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY.

10.13 Disclaimer of Warranties. Except as expressly set forth in this Agreement (as modified by the Disclosure Letter) and the Bill of Sale, Seller makes no representation or warranty, inducement, promise, agreement assurance or statement to Purchaser upon which Purchaser is relying, or in connection with which Purchaser has made or will make any decisions concerning the Purchased Assets, including with respect to (i) any projections, forecasts, estimates or budgets provided or made available to Purchaser of future revenues, future results of operations (or any component thereof), future cash flows or future financial condition (or any component thereof) or any forward-looking information provided to Purchaser or (ii) any other information or documents provided or made available to Purchaser or its counsel, accountants or advisors. Purchaser and its representatives and agents have made all inspections and investigations of the Facility Operations and the Purchased Assets deemed necessary or desirable by Purchaser. Purchaser is purchasing the Purchased Assets based on the results of its inspections and investigations, and not on any representation or warranty of Seller or any of its Affiliates not expressly set forth in this Agreement (as modified by the Disclosure Letter). There is no assurance that any projected or forecasted results will be achieved. EXCEPT AS TO THOSE MATTERS EXPRESSLY COVERED BY THE REPRESENTATIONS AND WARRANTIES SET FORTH IN ARTICLE 5 (as modified by the Disclosure Letter), SELLER IS SELLING THE PURCHASED ASSETS ON AN "AS IS, WHERE IS, WITH ALL FAULTS" BASIS IN THEIR EXISTING CONDITION (WHETHER PHYSICAL, FINANCIAL OR OTHERWISE) AND SELLER DISCLAIMS ALL OTHER WARRANTIES, REPRESENTATIONS AND GUARANTEES WHETHER EXPRESS OR IMPLIED. SELLER MAKES NO REPRESENTATION OR WARRANTY AS TO FITNESS, NONINFRINGEMENT, MERCHANTABILITY, HABITABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE AND NO IMPLIED WARRANTIES WHATSOEVER. Purchaser acknowledges that none of Seller, its Affiliates nor their respective directors, officers, employees, Affiliates, controlling Persons, agents, advisors and representatives nor any other Person has made any representation or warranty, express or implied, as to the accuracy or completeness of any memoranda, charts or summaries heretofore made available by Seller or its representatives to Purchaser (including any of the information set forth in management presentations made available to Purchaser, its Affiliates or its representatives, in materials made available in any "data room," financial projections or other projections, in "break out" discussions, in responses to questions submitted by or on behalf of Purchaser, its Affiliates or its representatives, whether orally or in writing, in materials prepared by or on behalf of Seller or its Affiliates in any other form) or any other information which is not expressly set forth this Agreement (including the Disclosure Letter and the schedules and exhibits hereto) (including any delivered or made available pursuant to Section 7.1(a)), and none of Seller, its Affiliates or their respective directors, officers, employees, Affiliates, controlling Persons, agents, advisors and representatives or any other Person will have or be subject to any liability to Purchaser or any other Person resulting from (i) the distribution of any such information to, or use of any such information by, Purchaser or any of its Affiliates, agents, consultants, accountants, counsel or other representatives or (ii) any errors in or omissions in any such information. Purchaser acknowledges and agrees that it is not relying, nor is it entitled to rely, upon any representations or warranties or other statements of fact or opinion, other than the representations and warranties expressly set forth in this Agreement (as modified by the Disclosure Letter) and any Ancillary Agreement.

10.14 Specific Performance. The Parties agree that irreparable damage would occur and that the Parties would not have any adequate remedy at law in the event that the obligations of the Parties to effect, on the terms and conditions set forth herein, the Closing hereunder (including the obligations set forth in Section 7.3) were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that, notwithstanding anything contained to the contrary in this Agreement, including, if applicable, under Section 9.2, the Parties shall be entitled to an injunction or injunctions to prevent such (and only such) breaches of this Agreement and to enforce specifically (without proof of actual damages or harm, and not subject to any requirement for the securing or posting of any bond in connection therewith) such terms and provisions of this Agreement in the United States District Court for the District of Delaware or any Delaware State court sitting in the County of New Castle, this being in addition to any other remedy to which they are entitled at law or in equity, including money damages. If on the Termination Date, there is a pending action that has been brought by a party hereto before a court of competent jurisdiction seeking the remedies provided for in this Section 10.14, then, without further action, the Termination Date shall be automatically extended until the date that is five Business Days after the dismissal, settlement or entry of a final and non-appealable order with respect to such action.

10.15 Non-Recourse. No past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney or representative of any Party or its Affiliates shall have any liability for any obligations or liabilities of such Party under this Agreement or of such Party its applicable Affiliates under any Ancillary Agreements of or for any claim based on, in respect of, or by reason of, any of the Transactions.

10.16 Release. Effective from and after the Closing, each Party, on their own behalf and on behalf of their respective Affiliates, and each of the foregoing's successors, assigns, heirs and beneficiaries, irrevocably and unconditionally waives, releases and promises never to assert any claims, causes of action or similar rights of any type (however described and however arising) that any Party or any of their respective successors or assigns may currently have, whether or not now known, against the other Party, their respective Affiliates or their respective current or former managers, equity holders, officers, agents or employees (the "Released Parties") which are on account of any matter whatsoever attributable to the period, or arising during the period, from the beginning of time through and including the Closing Date, with the exception of the Specified Obligations (the "Released Claims"). Each Party acknowledges and agrees that (a) the release contained in this Section 10.16 (this "Claim Release") shall be effective as a bar to all Released Claims, (b) this Claim Release shall be given full force and effect according to each and all of its express terms and provisions and (c) if any Party or any of their respective Affiliates assert any Released Claim against any other Released Parties, then this Claim Release shall serve as a complete defense to such Released Claim and the Party attempting to assert such Released Claim shall, jointly and severally, hold harmless, indemnify, pay and reimburse Released Parties to the extent of any Losses arising or resulting from the assertion of an Released Claim. As used herein, "Specified Obligations" means the rights of any Party: (i) set forth in or arising under any provisions of this Agreement or any of the Ancillary Agreements or with respect to the subject matter hereof, including any claim indemnifiable hereunder, (ii) with respect to indemnification or expense reimbursement from Seller pursuant to such Seller's governing documents or mandated under applicable Law (other than such rights with respect to matters indemnifiable by Seller), or (iii) under a Benefit Plan of Seller. As to the Released Parties, each Party acknowledges and agrees that it and he is aware of, has had the opportunity to seek legal counsel and is familiar with the provisions of California Civil Code Section 1542, which provides as follows:

"A general release does not extend to claims that the creditor or releasing party does not know or suspect to exist in his or her favor at the time of executing the release and that, if known by him or her, would have materially affected his or her settlement with the debtor or released party."

With full awareness and understanding of this provision, each Party hereby waives all rights that this provision or any comparable provision under any state, federal or non-U.S. law may give to any other Party as well as under any other statute or common law principles of similar effect. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY, FOR EACH SUCH PARTY AND THEIR RESPECTIVE SUCCESSORS AND ASSIGNS, WAIVES THE BENEFIT OF ANY PROVISION OF APPLICABLE LAW TO THE EFFECT THAT A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS OR MATTERS WHICH THE RELEASING PARTY DID NOT KNOW OR SUSPECT TO EXIST IN THE RELEASING PARTY'S FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY THE RELEASING PARTY MAY HAVE AFFECTED ITS SETTLEMENT WITH THE RELEASED PARTY.

10.17 Counterparts; Exchange by Electronic Transmission. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, and all of which shall constitute one and the same agreement. This Agreement may be executed by PDF or by other means of electronic signature (including, e.g., DocuSign), and the exchange of a fully executed Agreement (in counterparts or otherwise) by electronic transmission in PDF format, shall in each case be sufficient to bind the Parties to the terms and conditions hereof.

10.18 Severability. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any term or other provision of this Agreement, or the application thereof to any person or entity or any circumstance, is invalid, illegal or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other persons, entities or circumstances shall not be affected by such invalidity, illegality or unenforceability, nor shall such invalidity, illegality or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

[Remainder of page intentionally left blank]

FINAL CONFIDENTIAL PRESS RELEASE FOR ISSUANCE

MONDAY NOVEMBER 4, 2024 AT 5:30AM PT, 7:30 AM CT



Nektar Announces Definitive Agreement with Ampersand Capital Partners to Sell Its Commercial PEGylation Reagent Manufacturing Business in Alabama

Huntsville-based facility to be spun out as standalone Ampersand portfolio company.

Nektar to receive \$90 million in total consideration for the business, comprised of \$70 million in cash and \$20 million equity ownership in new portfolio company.

Strategic divestiture allows Nektar to streamline its operations and continue its strategic focus on the development of core R&D programs in immunology.

SAN FRANCISCO -- Nov. 4, 2024 -- Nektar Therapeutics (Nasdaq: NKTR), a global biotechnology company focused on the discovery and development of novel therapies to treat autoimmune disorders, today announced that it has entered into a definitive agreement to sell its Huntsville, Alabama manufacturing facility and reagent supply business to Ampersand Capital Partners, a Boston-based private equity firm with a decades-long track record of investing in life sciences and healthcare companies, including contract manufacturing and pharma services businesses.

Ampersand has agreed to acquire Nektar's commercial-scale manufacturing facility and PEGylation reagent supply business for a total consideration of \$90 million, comprised of \$70 million in cash proceeds and \$20 million in a retained equity position for Nektar in a newly-created Ampersand portfolio company. Ampersand has also committed to invest additional growth equity capital into the new portfolio company. Following the closing of the transaction, Nektar will be entitled to appoint a representative to the board of the new Ampersand portfolio company.

The Huntsville site is a 124,000 square foot, commercial-scale specialized manufacturing facility with a strong history of supporting commercial supply chains for PEGylated therapeutics across global markets. The facility has several commercial-scale supply chain contracts with leading pharmaceutical companies. All of Nektar's employees at the Huntsville facility will be offered employment at the new portfolio company, ensuring continuity in the high-quality manufacturing and PEGylation expertise that longstanding customers trust and rely on.

"This sale streamlines Nektar's operations as we continue to focus on the future success and clinical advancement of rezpegaldesleukin and our other antibody-based immunology pipeline assets, including our TNFR2 antibody and bispecific programs," said Howard W. Robin, President and CEO of Nektar Therapeutics. "We believe Ampersand is an optimal partner to lead the manufacturing activities at the Huntsville facility. Importantly, Ampersand's commitment to investing in the plant's business will help ensure that Nektar's existing commercial customers of PEGylation reagents will continue to be well served and will also provide uninterrupted access to a reliable supply of PEGylation reagents for Nektar's needs. The sale also further extends Nektar's cash runway into the fourth quarter of 2026."

Nektar and the new Ampersand portfolio company will be entering into manufacturing supply agreements to meet Nektar's PEG reagent needs for rezpegaldesleukin and certain pipeline programs.

"We were immediately impressed with the world-class PEGylation reagent manufacturing capabilities at this facility," said David Anderson, General Partner, Ampersand Capital Partners. "The Huntsville site and its employees have played an important role in the development of significant FDA-approved PEGylated therapeutic medicines. We look forward to investing in and growing the site as a stand-alone manufacturing business dedicated to serving existing and new customers."

The sale is not subject to financing contingencies. The transaction will be subject to customary closing conditions and costs and is expected to close by December 2, 2024. Following the closing, Nektar will retain all rights to current and future royalty streams and milestones related to existing PEGylated product license agreements.

UBS Investment Bank acted as exclusive financial advisor and Sidley Austin LLP served as legal advisor to Nektar Therapeutics. Goodwin Procter LLP acted as legal advisor to Ampersand Capital Partners.

About Nektar Therapeutics

Nektar Therapeutics is a clinical-stage biotechnology company focused on developing treatments that address the underlying immunological dysfunction in autoimmune and chronic inflammatory diseases. Nektar's lead product candidate, rezpegaldesleukin (REZPEG, or NKTR-358), is a novel, first-in-class regulatory T cell stimulator being evaluated in two Phase 2b clinical trials, one in atopic dermatitis and one in alopecia areata. Our pipeline also includes a preclinical candidate NKTR-0165, which is a bivalent tumor necrosis factor receptor type II agonist antibody. Nektar, together with various partners, is also evaluating NKTR-255, an investigational IL-15 receptor agonist designed to boost the immune system's natural ability to fight cancer, in several ongoing clinical trials. Nektar is headquartered in San Francisco, California. For further information, visit www.nektar.com and follow us on LinkedIn.

About Ampersand Capital Partners

Ampersand Capital Partners, founded in 1988, is a middle-market private equity firm with \$3 billion of assets under management, dedicated to growth-oriented investments in the healthcare sector. With offices in Boston, MA, and Amsterdam, Netherlands, Ampersand leverages a unique blend of private equity and operating experience to build value and drive long-term performance alongside its portfolio company management teams. Ampersand has helped build numerous market-leading companies across each of the firm's core healthcare sectors. For additional information, visit ampersandcapital.com or follow us on LinkedIn.

Cautionary Note Regarding Forward-Looking Statements

This press release contains forward-looking statements which can be identified by words such as: "will," "expect," "develop," "extend," "advance," "anticipate," "can," and similar references to future periods. Examples of forward-looking statements include, among others, statements regarding the therapeutic potential of, future development plans for rezpegaldesleukin, NKTR-0165 and our other drug candidates and whether all closing conditions of the announced definitive agreement will be met. Forward-looking statements are neither historical facts nor assurances of future performance. Instead, they are based only on our current beliefs, expectations and assumptions regarding the future of our business, future plans and strategies, anticipated events and trends, the economy and other future conditions. Because forward-looking statements relate to the future, they are subject to inherent uncertainties, risks and changes in circumstances that are difficult to predict and many of which are outside of our control. Our actual results may differ materially from those indicated in the forward-looking statements. Therefore, you should not rely on any of these forward looking statements. Important factors that could cause our actual results to differ materially from those indicated in the forward-looking statements include, among others: (i) our statements regarding the therapeutic potential of rezpegaldesleukin, NKTR-0165 and our other drug candidates are based on preclinical and clinical findings and observations and are subject to change as research and development continue; (ii) rezpegaldesleukin, NKTR-0165 and our other drug candidates are investigational agents and continued research and development for these drug candidates is subject to substantial risks, including negative safety and efficacy findings in future clinical studies (notwithstanding positive findings in earlier preclinical and clinical studies); (iii) rezpegaldesleukin, NKTR-0165 and our other drug candidates are in preclinical and clinical development, and the risk of failure is high and can unexpectedly occur at any stage prior to regulatory approval; (iv) the timing of the commencement or end of clinical trials and the availability of clinical data may be delayed or unsuccessful due to regulatory delays, slower than anticipated patient enrollment, manufacturing challenges, changing standards of care, evolving regulatory requirements, clinical trial design, clinical outcomes, competitive factors, or delay or failure in ultimately obtaining regulatory approval in one or more important markets; (v) patents may not issue from our patent applications for our drug candidates, patents that have issued may not be enforceable, or additional intellectual property licenses from third parties may be required; and (vi) certain other important risks and uncertainties set forth in our Quarterly Report on Form 10-Q filed with the Securities and Exchange Commission on August 9, 2024. Any forward-looking statement made by us in this press release is based only on information currently available to us and speaks only as of the date on which it is made. We undertake no obligation to update any forward-looking statement, whether written or oral, that may be made from time to time, whether as a result of new information, future developments or otherwise.

Contact:

For Investors:

Vivian Wu Nektar Therapeutics 628-895-0661

For Media:

Madelin Hawtin LifeSci Communications 603-714-2638 mhawtin@lifescicomms.com